KWAME NKRUMAH UNIVERSITY OF SCIENCE AND TECHNOLOGY, KUMASI, GHANA

Effect of Climate Change and Variability on Pearl Millet (Pennisetum glaucum (L.) R. Br.) Production in the Sudanian and Sahelian Agro-Ecological Zones in Mali

By

Halimatou Aboubacar Touré

(MSc in Planning Geography)

A Thesis submitted to the Department of Civil Engineering, College of Engineering

in partial fulfilment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

Mag. SANE CLIMATE CHANGE AND LAND USE

May, 2016

NO

DECLARATION

I hereby declare that this submission is my own work towards the PhD and that, to the best of my knowledge, it contains no material previously published by another person, nor material which has been accepted for the award of any other degree of the University, except where due acknowledgment has been made in the text.

Halimatou Aboubacar Touré (PG6679511)	Signature	Date
Certified by:		
Prof N Kyei- Baffour		
(Supervisor 1)	Signature	Date
Dr Kalifa Traore (Supervisor 2)	Signature	Date
Prof. Y. A. Tuffour		
(Head of Department)	Signature	Date

ABSTRACT

Climate change is expected to increase vulnerability in all agro-ecological zones of Mali through rising temperature and more erratic rainfalls, which will have drastic consequences on food security and economic growth. This study aimed at assessing the effect of Climate Change and Variability on Pearl Millet (Pennisetum glaucum (L.) R. Br.) Production in the Sudanian and Sahelian Agro-Ecological Zones in Mali. First, the perception of farmers on climate change variability and their adaptation practices to overcome or reduce the negative impacts of climate change on their farming system as well as their livelihoods were investigated in the Sahelian zone. Therefore, 119 farmers' household including women and youth were randomly selected and interviewed using structured questionnaire. Next, the changes in trends of daily temperature and precipitation extremes in Sotuba and Cinzana during the period 1961 - 2014 were assessed. In order to investigate extreme precipitation and temperature, daily minimum and maximum records for two stations (Ségou and Bamako) were collected at the Aghrymet Meteorological Centre of Niamey for the period 1961-2014. Finally, the performance of two millet varieties in two agroecological zones of Mali was assessed using the DSSAT (The decision support system for agrotechnology transfer) model under different climate conditions. Two years' experiments were therefore conducted at Agricultural Research Station of Sotuba and Cinzana in the randomised complete block design with 4 replications. The fertiliser treatments included Control, MANURE, NPK and NPK+MANURE. DSSAT model were used to simulate crop grain yields under 2 different weather conditions (historical and future). The results showed that farmers perceived a decrease in annual rainfall as the main factors of climate change and variability. Several strategies such as selling of animals, planting improved crop varieties, engaging in new activities (outside agriculture) and credit were the commonly preferred adaptation strategies to deal with climate change and variability. A significant decrease of warming trends in cool days, cool nights, whereas warm extreme nights, day times and warm spells on the contrary showed positive significant increasing warming throughout the Ségou Region. The results of precipitation extremes for Ségou showed significant decrease in consecutive wet and extremely wet day. The study provided evidence that during the last 53 years, Ségou was particularly affected by warm extremes based on night time indices rather than cold extremes based on day time indices. At Sotuba, the average grain yield was 1293 kg ha⁻¹ and 1503 kg ha⁻¹ for the years 2013 and 2014, respectively. In Cinzana, the observed average grain yields in 2013 and 2014 were 1390 and 1530 kg ha⁻¹, respectively. The analyses did show significant differences between the varieties and among the fertilizer treatments. Moreover, millet yield responded to the different historical climatic conditions under all treatments. In line with the experimental data, the highest historical grain yields were achieved when the combination NPK+MANURE was applied. Millet grain yields were higher under the simulations with historical weather data than the simulations with climate change scenarios. Simulation of climate change effects on millet grain yield showed that all scenarios underestimated crop yield compared to the Baseline for all treatments and both varieties. Significant differences (P < 0.05) were revealed among the scenario outputs. All the varieties showed lowest grain yields under the four treatments for ACCESS1-0 (Hot-Wet) among the

scenarios. Based on the findings of this study, policy could be developed to enhance farmers' adaptation strategies in the Sudanian and Sahelian zones of Mali.



DEDICATION

To my father (Aboubacar Boncano Touré) and my mother (Mme Touré Aissatou Traoré), for making me what I am today. They bore me, raised me and supported me in the hardest of times. Also to you, my husband, Idriss Sermé and my twin daughters Habi Amira Idriss and Aissatou Charifa Idriss. To my beloved siblings, Arahamatou, Youssouf and Mariam Aboubacar Touré for prayers and encouragements.



ACKNOWLEDGEMENTS

At this moment, I would like first and foremost to give my sincere gratitude to the Almighty Allah for His grace, mercy, blessings, protection and strength throughout this study. The one who was, who is and is to come, the author and source of all wisdom and knowledge, the architect of our survival, the giver of insight, and the father of all discoveries.

The successful completion of this PhD thesis would not have been possible without the contribution either in knowledge, time and efforts of many people and institutions. All of them deserve my appreciation and acknowledgements, and I am pleased to take this opportunity to thank them. I cannot cite all the names, but I would like to mention a few in particular.

I am grateful to the German Federal Ministry of Education and Research (BMBF) which fully funded my PhD studies and this research through the West African Science Service Centre on Climate Change and Adapted Land use (WASCAL).

I am grateful to main my supervisor, Prof. N. Kyei-Baffour of the department of Agricultural Engineering at the Kwame Nkrumah University of Science and Technology, Kumasi, Ghana, for his invaluable guidance throughout this study. He accepted to supervise me on my request and provided me with valuable feedback on my work. It has always been a pleasure meeting him.

I also express gratitude to my co-supervisor, Dr Kalifa Traore. His valuable comments and his constant encouragement are always greatly appreciated. He also helped me lifewise. I really appreciate and will forever be grateful. God bless you.

I am also greatly indebted to Dr. Aboubacar Touré, Dr Abdoulaye Diallo, Abdoulaye Yorote, Sidi Toure, David Togo of the Institut d'Economie Rural (IER, Sotuba) of Mali for their moral and social support and also for their valuable comments and suggestions throughout this study. My warmest gratitude to Dr Seydou Traore, Dr Agaly Alhassane, Dr Mohamed Ly, Dr Lawali, Baba Sidiki, both from AGRHYMET who inspired and encouraged me. Thanks for your assistance during my stay there.

My colleagues of the first and second batches are thanked for their numerous stimulating discussions and constructive criticisms and suggestions during both the internal PhD Colloquiums at Steven Paris Hostel and at Wascal building.

I am wholly indebted to my parents, Aboubacar Touré and Mme Touré Aissatou Traoré for making me what I am today. Their prayers, continuous support and encouragements throughout my PhD study, also to my brother Youssouf Touré and sisters Arahamatou and Mariam A. Touré for their encouragements. I express my everlasting sense of gratitude to you, your endless efforts which sustained me at all stages of my life. I am very thankful to my in-laws specially my father and mother in-laws Ganadolo Sermé and Habi Coulibaby for their blessings through this research.

Lastly, but certainly not the least, my beloved husband, Idriss Sermé for his support, encouragements and for supporting my extended absences either in Mali, Niger and Ghana and my twin daughters Habi Amira Idriss and Aissatou Charifa Idriss Sermé for the love they showed to me and their patience and endurance during my absence for the last three to four years.

BADW

Halimatou Aboubacar Touré

May 2016

vi

WJSANE

TABLE OF CONTENTS

DECLARATION
ABSTRACT
ii
DEDICATION
iii
ACKNOWLEDGEMENTS
1V
LIST OF FIGURES
LIST OF TABLES
X1
LIST OF ABBREVIATIONS
AII CHADTER 1. CENERAL INTRODUCTION
CHAPTER I: GENERAL INTRODUCTION
1.1. Background
1.2. Problem Statement
1.3. Objectives of the Study
1.3.1 Main objective
1.3.2 Specifics Objectives
1.4. Hypothesis and Research Questions
1.4.1 Hypotheses
1.4.2 Research Questions
1.5. Organisation of the Research
CHAPTER 2: LITERATURE REVIEW
2.1. Introduction
9 SANE NO
2.2. Climate Change in Perspective
2.2.1. Definitions of Terms and their Concepts
9

	2.2.2 Climate Change Science 1	2
	2.2.3 Observation and Prediction of Climate Change 1	5
	2.2.4 Impact of Climate Change 1	6
2.3.0	Generality on Pearl Millet	:5
	2.3.1 Origin, Taxonomy and Geographical Distribution	:5
	2.3.2. Morphological Characters	•••
	2.3.3. Biological Characteristics27	•••
	2.3.4 Ecology	•••
	2.3.5 Mode of Reproduction	9
	2.3.6 Production and Economic Importance29	•••
	2.3.7 Adaptation and Agro-Climatological Requirements	1
2.4	Potential Yields	•••
37		
2.5.	Other Background Information	8
	2.5.1 Diseases, Pests and Weeds38	•••
	2.5.2. Insect and Pest	9
	 2.5.3 Drought Tolerance	•••
2.6.0	Gaps in Knowledge	-1
CHAP	TER 3: MATERIALS AND METHODS 4	-2
3.1. 42	Introduction	•••
3.2. 42	Study Area	•••
3.3 44	Cinzana Commune	•••
	3.3.1 Physical Characteristics	

3.4. 45	Sotuba Commune
	3.4.1. Physical Characteristics
3.5.	Perception of Famers on Climate Change/ Variability and their Adaptation 47
	3.5.1. Target Population and Sampling47
	3.5.2. Data Analyses
3.6. and 49	Assessment of the Trends in Extreme Precipitation and Temperature Indices for Sotuba Cinzana
	3.6.1. Tu Tiempo and NASA POWER data 49
	3.6.2 Brief Description of RClimDex
3.7. Clin	Assessment of The Changes on Yields of Millet Crop under Different Scenarios of nate Change Using DSSAT Model
	3.7.1. Experimental Design and Treatments
	3.7.2. Brief Description of Crop Model 'DSSAT'
CHAP ADAP	TE <mark>R 4: FARMERS' PERCEPTIONS ON CLIMATE VARIABILITY AND</mark> TATION STRATEGIES TO CLIMATE CHANGE IN CINZANA, MALI
4.1. 54	Introduction
4.21	Material and Methods
	4.2.1. Overview of the Study Area 57 4.2.2. Data Analyses 60
4.3. 61	Results and Discussion
	4.3.1 Socio-Economic Characteristics of Respondents
	4.3.2. Farmers' Perceptions of Climate Change and Variability and its Impacts on TheirFarming System64
	4.3.3. Climatological Evidence Compared to Farmers' Perceptions
	4.3.4. Coping and Adaptation Mechanisms to Climate Change and Variability
	4.3.5 Correlation Estimation of Some Factors Influencing the Farmers' Adaptation Strategies to Climate Change Effects

4.4 74	Conclusion	••
CHAP TEMP	TER 5: ASSESSMENT OF CHANGING TRENDS OF DAILY PRECIPITATION AND PERATURE EXTREMES IN SOTUBAAND CINZANA IN MALI	5
5.1. 75	Introduction	, .
5.2. 79	Materials and Methods	·•
	5.2.1. Study Area	9
	5.2.2 Analysis of Extreme Events80	••
5.3. 84	Results and Discussion	••
	5.3.1. Temperature and Rainfall Indices	4
5.4 94	Conclusion	••
CHAP	TER 6: ASSESSING THE CHANGES IN YIELDS OF MILLET CROP UNDER	
DIFFE 95	ERENT SCENARIOS OF CLIMATE CHANGE USING DSSAT MODEL	•
6.1 95	Introduction	••
6.2 97	Materials and Methods	••
	6.2.1. Experimental Design and Crop Management	7
	6.2.2 Data Collected	••
	6.2.3 Crop Model	••
	6.2.4 Selection of Pearl Millet Varieties	0
	6.2.5 Soil and crop management input	0
	6.2.6 Experiment for model parameterization for crop yield simulation	1
	6.2.7 Modelling of Effect of Climate change and variability on millet grain yield 102	2
	6.2.8 Statistical Analyses	••

6.3. Results and discussion
6.3.1. Inter-Annual Rainfall Variability, Rainfall Distribution and its Relation with Crop Yields for Sotuba and Cinzana
6.3.2. Analysis of Variance of Agronomic Parameters Studied 107
6.3.3. Validation and Evaluation of the CSM-CERES Millet Model for Sotuba and Cinzana 113
6.3.4 Simulated millet grain yield under climate variability over 1983 – 2012 in Sotuba and Cinzana 117
6.3.5. Effect of climate change scenarios on millet grain yield for Cinzana and Sotuba 120
6.4 Conclusion
CHAPTER 7 CONCLUSIONS AND RECOMMENDATIONS 129
7.1 Introduction
7.2 conclusions
7.3 Recommendations
APPENDIX
134
REFERENCES
SANE NO

LIST OF FIGURES

Figure 3. 1 Study Area showing Sotuba (left) and Cinzana(right)
Figure 3. 2 Different components and software applications of DSSAT (Source: Jones et al., 2003)
53 Figure 4. 1: The Location of the Study Area
57 Figure 4. 2: Main Crops Cultivated by Farmers at Cinzana in 2013
61
Figure 4. 3: Cinzana Farmer's Reasons for Crop Choices
Figure 4. 4: Farmers' Perceptions of Climate Change and Variability in Cinzana Community in
2013
64
Figure 4. 6: Effects of climate change as perceived by farmers in Cinzana
Figure 4. 5: Annual Rainfall Indexes (mm) from 2000 to 2013 in Cinzana (Data source: Cinzana
Meteorological Station)
68
Figure 4. 6: Annual Temperature Means from 2000 to 2013 (Data source: Cinzana
Meteorological Station)
68
Figure 4. 7: Famers' Adaptation Strategies to Climate Change and Variability Stressor in
Cinzana community in 2013
70
Figure 5. 1 Evolution of the percentage of cold nights (Tn10p) and that of cold days (Tx10p) in
Sotuba (a and c) and Cinzana (b and d) from 1960 to 2014 period
Figure 5. 2 Evolution of the percentage of warm nights (Tn90p) and that of warm days (Tx90p)
and warm spells (wsid) in Sotuba (a and c) Ségou (b and d) from 1960 to 2014 period 88
Figure 5.3: Evolution of the percentage of warm spells (wsid) in Sotuba (a) Cinzana (b) from
1960 to 2014 period
89
Figure 5. 3 Evolution of the maximum 5 day, precipitations (RX5 Day) and that of consecutive
wet days (CWD) in Sotuba (a, b and d) Cinzana (a and c) from 1960 to 2014 period
Figure 5. 4 Evolution of extremely wet days (R95p) and that of total annual precipitation
(PRCPIOIE) in Sotuba (a, b and d) Cinzana (a and c) from 1960 to 2014 period
Figure 6. 1 Rainfail distribution in 2013 and 2014 in Sotuba (A and B) and in Cinzana (C and D)
105 Figure 6, 2 Varieties for plant height yields in 2013 and 2014
107
Figure 6. 3 Varieties 1000 grain weight in 2013 and 2014 in Sotuba
Figure 6. 4 Straw yields in 2013 and 2014 in Sotuba
Figure 6. 6a and b. Year x varieties and year x amendments for grain yields in 2013 and 2014 in

Cinzana					•
112					
Figure 6. 7 Com	parison between simulate	ed and observe	ed grain yields for	Sotuba (a and b) and	
Cinzana (ca	and d) location with differ	rent varieties			ý
Figure 6.8. Simu	ılated millet grain yield u	nder historica	l weather data for	Cinzana (a and b) and	
Sotuba (c	and d)				•
118					



LIST OF TABLES

Table 4. 1 Coping Strategies to Climate Change and Variability of Farmers in Cinzana
Commune in 2013
Table 4. 2 Correlation of factors studied on adaptation mechanisms in Cinzana commune 72
Table 5. 1: Meteorological Sations Used in the Study
Table 5. 2 Summary of rainfall and temperature indices used in the study
Table 5. 3 Trend per percentage and day with positive and negative slope of temperature indices
extremes in Sotuba and Cinzana for 1960-2014 periods
Table 5. 4 Trend per percentage, days with positive and negative slopes of rainfall indices
extremes in Sotuba and Cinzana for 1961-2014 period91
Table 6. 1 Experiment layout for randomised complete bock design with four replicates
Table 6. 2 Chemical properties at Cinzana site for model evaluation experiment
Table 6. 3 Chemical properties at Sotuba site for model evaluation experiment 101
Table 6. 4 Summary of analysis of variance on Yields and Yield components of two (2) varieties
evaluated in Sotuba in the Sudanian agro-ecological zone 107
Table 6. 5 Summary of analysis of variance on Yields and Yield components of two (2) varieties
evaluated in Cinzana a Sahelian agro-ecological zone109
Table 6. 6 Statistical parameters of simulation performance for the Cinzana location
Table 6. 7 Statistical parameters of simulation performance for Sotuba location
Table 6.8. Historical average crop yields and coefficient of variation in Cinzana 120
Table 6.9. Historical average crop yields and coefficient of variation in Sotuba
Table 6.10 Changes in grain yields for different climate scenarios and treatments in Cinzana. 126
Table 6.10 Changes in grain yields for different climate scenarios and treatments in Sotuba 126



LIST OF ABBREVIATIONS

AGRHYMET	:	Centre for Agriculture Hydrology and Meteorology	
ANOVA	:	Analysis of Variance	
BMBF	:	German Federal Ministry of Education and Research	
CCI	:	Canadian Conservation Institute	
CERES	:	Crop Environment Resource Synthesis	
CH4	:	Methane	
CILSS	:	Permanent Interstate Committee for Drought Control in the Sahel	
CLIVAR	:	Climate Variability and Predictability	
CO ₂	:	Carbon dioxide	
CRRA	:	Regional Agriculture Research Centre	
CGCM	:	Couplet General Circulation Model	
CGIAR		Consultation Group for International Agricultural Research	
DAS		Days After Sowing	
DSSAT	G	Decision Support System for Agro-Technological Transfer	
ET	1:	Expert Team	
ETCCDI	:	Expert Team on Climate Change Detection and Indices	
FAO	:-	Food and Agriculture Organization	
GCC	:	Global Climate Change	
GCM		General Circulation Model	
GHG	:	Greenhouse Gases	
HADCM	:	Hadley Centre Couplet Climate Model	
IAPPS	10	International Association for Plant Protection Sciences	
ICASA	2	Independent Communications Authority of South Africa	
ICRISAT	: <	International Crops Research Institute for the Semi-Arid Tropics	
IER	:	Institute for Rural Economy	
IFAD	:	International Fund for Agricultural Development	
IPCC	:	Intergovernmental Panel on Climate Change	

IITA	:	International Institute of Tropical Agriculture
JCOMM	:	Joint Technical Commission for Oceanography and Marine
		Meteorology
KNUST	:	Kwame Nkrumah University of Science and Technology
NASA	:	National Aeronautics and Space Administration
NCDC	:	National Climatic Data Centre
NOAA	:	National Oceanic and Atmospheric Administration
N2O	:	Nitrous oxide
NSIDC	:	National Snow and Ice Data Centre
O 3	:	Tropospheric ozone
OCDE	:	Organisation for Economic Co-operation and Development
RMSE	:	Root Means Square Error
SRES	:	Special Report on Emissions Scenarios
SSWA	:	Sudanian Sahelian Zone West Africa
Tmax	:	Maximum Temperature
Tmean		Mean Temperature
Tmin		Minimum Temperature
UN	1	United Nations
UNCCD	/ :/	United Nations Convention on Climate and Desertification
UNFCCC	·	United Nations Framework for the Convention of Climate Change
UNEP		United Nations Environment Programme
UNFCCC		United Nations Framework for the Convention of Climate Change
USAID		United States Agency for International Development
WASCAL		West African Science Service Centre on Climate Change Adapted
12	kr	Land Use
WISARD	40	Web Based Information Services for Agricultural Research for
	~	Development
WBGU	. ~	German Advisory Council on Global Change
WMO	:	World Meteorological Organisation
WRI	:	World Resources Institute



CHAPTER 1:

GENERAL INTRODUCTION

1.1. Background

Broadly scientists are certain that global climate change (GCC) is mainly due to increasing concentrations of greenhouse gases (GHG) from anthropogenic activities such as deforestation and burning of fossil fuels causing global climate change (GCC) (Mendelsohn and Dinah, 2005; IPCC, 2007; National Research council, 2010). They are also certain that ineluctable global warming will have important impact on the worldwide climate (Intergovernmental Panel for Climate Change [IPCC], 2007). Greenhouse gases greatly affect the earth's temperature, which has resulted in the present earth temperature of 14 °C (le Treut et al., 2007; Karl and Trenberth, 2003). According to Lu et al. (2007), the effect of an increase in global temperature causes a rise in a change in the pattern and amount of precipitation, a more frequent occurrence of extreme weather event including heat waves, heavy rainfall and droughts, species extinction and ocean acidification caused by shifting temperature regimes. However, the climate change system which encloses an important deal of climate fluctuations and natural variability have already contributed to the further warming of the globe of around 0.1 °C per decades which for several decades is high (Solomon et al., 2007; Mendelson and Dinah, 2005). Global warming has been projected to accelerate within the 21^{st} century (IPCC, 2007).

According to Clark *et al.* (2002), climate change is like an intricate and interdependent environmental challenge facing the entire world. The two consequences expected of climate are:

- Bio-physical and
- socio-economic

Both are related to this present study. Bio-physical impact includes more frequent and intense storms, rising sea water level, extinction of species, crop failure and worsening drought. Furthermore, it has been observed that changes in precipitation and cloud cover, glaciers, reducing snow cover and melting of polar caps are among the bio-physical impact (United Nations Framework for the Convention of Climate Change [UNFCCC], 2007; Mendelson and Dinah, 2005; UNDP, 2004). The impact of socio-economic development is characterised with biophysical impact including environmental degradation. A typical example is the linkage of environmental degradation to poverty and food security reduction (Koch *et al.*, 2006; Clark *et al.*, 2002). Those linkages stemmed out of the expectations that climate change will affect water resources and food that are critical for survival across developing countries especially in African countries. The populace of these countries heavily rely on supply system that are dependent on climate variation (Nhemachena and Hassan, 2007; Chipo, 2010).

Global warming is responsible for the rise in the earths atmospheric and ocean temperatures. Increased Earth mean surface temperature has been recorded of 0.8 °C since the early 20th century, in 1980 about two-thirds of the increase occurred (National Research Council, 2010). In addition, global average temperature showed a warming of 0.78 °C (0.72 to 0.85) °C. Over the period from 1850 to 2012 and recent predictions for the end of the 21st century indicate that global average temperature increase will be between 1.5 °C and 2 °C (IPCC, 2013).

Climate change is expected to increase vulnerability in all agro-ecological zones of Mali through rising temperature and more erratic rainfalls, which will have drastic consequences on food security and economic growth (Butt and McCarl, 2006; Dell *et al.*, 2012). Climate change projections from the Hadley Centre Coupled Model (HADCM) and Coupled General Climate Model (CGCM), suggest that by the year 2030, average temperature in Mali may increase by 1 -

2.75 °C and 2 - 4 °C before 2060 (Butt *et al.*, 2005; 2006; Konaté, 2010). The impact of temperature on agriculture is of major significance: A 1 °C rise is linked to a 2.7 % reduction in growth in agricultural outputs (Dell *et al.*, 2012). The biophysical basis is fairly clear-cut. Temperature affects water availability, evapotranspiration and plant physiology directly. Projections for these agro-ecological zones have indicated that rainfall may increase or decrease (Butt *et al.*, 2005; Kanno *et al.*, 2008). Increases in rainfall may reach 12 to 16 % (Butt *et al.*, 2005). Rainfall characteristics are specially affected by these changes in rainfall, the onset and ending of rainy season and the distribution and intensity of rainfall events (Traoré *et al.*, 2000; Konaté, 2010). The impact of these changes on rainfall characteristic is 20 - 60 % yield losses for agricultural productions by 2025 (Ben *et al.*, 2002; Butt *et al.*, 2005). Here too, the biophysical basis is fairly clear-cut that rainfall characteristics affect water balance, more specifically water availability, evapotranspiration and plant physiology directly (Dell *et al.*, 2012).

Agriculture remains the main source of employment (60% of the population) and contribute significantly to gross domestic product (GDP) (30%) for countries in the Sudano-Sahelian zone of West Africa (SSWA) (FAO, 2012). Smallholder agricultural production is dominated by rain-fed production of sorghum, maize and millet for food consumption and cotton for the market. Farmers experience low yields resulting in increasing uncertainty about being able to produce the food needed for their families (Breman and Sissoko, 1998; Drechsel *et al.*, 2001). Major factors contributing to such uncertainty and low productivity are climate variability, poor soil fertility, poor agricultural management and climate change. In West Africa, soil fertility is inherently low (Bationo and Buerkert, 2001; Giller *et al.*, 2011; Piéri, 1989; Vanlauwe *et al.*, 2010) and represents the main constraint for agricultural development. This situation is heightened by the reduction of cultivation of fragile lands, reduction in fallow lengths, limited use of inorganic fertilizers due to high world market fertilizer prices and limited access to credit (De Graaff *et al.*, 2011; Ehui and

Pender, 2005). Also, low availability of organic fertilizers contributes to the decline in soil fertility. Land degradation including both water and wind erosion further impoverishes the soils in this region (Cleaver and Schreiber, 1994).

Significantly, rainfall increases have been observed in northern Europe, the eastern part of South and North America and in northern and central Asia, while a decrease and drying has been observed in the Sahel (IPCC, 2007a), against the background of strong multi-decadal variability in rainfall (Dai and Trenberth, 2004; Le Barbé *et al.*, 2002). In the Sahel region, wet conditions in the 1960s alternated with drier conditions in the 1970s and 1980s. Because of large regional differences, gaps and lack of long term data and spatial coverage, the evidence of changes in rainfall at global scales is complex. IPCC (2013), climate predictions indicate that the contrast in rainfall between dry and wet seasons between wet and dry regions will increase even the projections of rainfall are uncertain for the West African region because of uncertainty in the quantification of potential vegetation-climate links. Different global circulation models do not agree in their predictions of climate change for the region, although results of a number of the models indicate that on average the amount of rainfall in Mali will not change, but that the interannual variability of the amount of rainfall will increase. Also the frequency of extreme drought is expected to increase (Traore *et al.*, 2007; Washington and Harrison, 2004).

1.2. Problem Statement

Rainfall is highly variable spatially and in total amount. In Mali, climate change poses its worst impact through interference with food security to the growing population. Climate change impact can clearly be seen in recent climate. Despite the fact that both temperature and rainfall indicates an overall increase over time, it is clear that less rainfall is recorded in warmer years. It can therefore be concluded that temperature increase is likely to result in lower rainfall. Several GCM

projections have indicated that it will be a decrease in rainfall as well as an increase in temperature between the year 2040 and 2069. The effects of such changes on agriculture are uncertain. This therefore presents the need for an assessment of crop production responses to climate change to support adaptation strategies and policy formulation development to offset any adverse impacts. Multidisciplinary approaches including the use of models are more suitable since agricultural production involves a number of factors.

Change in rainfall patterns, rainfall variability and temperature are likely to increase the severity of periodic drought and extend dry seasons and thus decrease crop yield in several regions. Agricultural production in the Sahel region is heavily dependent on rainfall for the production of sorghum, millet, and cotton and maize. Farmers experience low yields resulting in increasing uncertainty about being able to produce the food needed for their families (Breman and Sissoko, 1998; Drechsel *et al.*, 2001).

Soil fertility is inherently low in Africa (Giller *et al.*, 2011; Vanlauwe *et al.*, 2010; Bationo and Buerkert, 2001) and represents the main constraint for agricultural development. Moreover, low availability of organic fertilizers contributes to the decline in soil fertility. Land degradation including both water and wind erosion further impoverishes the soils in this region (Cleaver and Schreiber, 1994).

The West African semi-arid tropics are those areas where rainfall intensity limits potential evapotranspiration for two to seven months annually. According to Matlon (1990), the potential increases in sorghum and millet yields exist only in the Sudano-Guinean zone and to an extent in the Sudanian zone. These regions are the most important taking about 70 % of total cereal production (Dendy, 1995). The yield of sorghum and millet may be low under the area's

environmental conditions but they are relatively the most dependable. Millet is the main cereal crop in Mali with a production of 1,157,810 t and a production area of about 1,484,190 ha (2005/2006) (Giller *et al.*, 2011). It is grown in all agricultural areas of Mali mainly in Sahelian, Sudanian and northern Guinean zones with planting areas estimated respectively at 38.6 %, 21.1 % and 12 % (Giller *et al.*, 2011; Traoré *et al.*, 2000) of the total cultivated area. The choice of crop to maturity is influenced by the onset of rain and agricultural calendar of the farmer. In case of early onset of the rainy season, the relatively long cycle varieties are sown first. In case of late onset of the rainy season, the emphasis is shifted to early maturing varieties. Some of the constraints to millet production include; low yield of available cultivars, inappropriate cycle of the varieties in the erratic climatic conditions, downy mildew, etc.

1.3. Objectives of the Study

1.3.1 Main objective

The main objective of this study was to determine the effect of climate change and variability on Pearl Millet yields in the Sudanian and Sahelian agro-ecological zones of Mali.

1.3.2 Specifics Objectives

1. To evaluate farmers' perceptions on climate variability and adaptation strategies to climate change in Cinzana Commune

2. To assess the trends in extreme precipitation and temperature indices for the Sudanian zone of Sotuba and Sahelian zone of Cinzana and

3. To Model the performance of two millet varieties in two agro-ecological zones of Mali using the DSSAT model under different climate conditions

1.4. Hypothesis and Research Questions

1.4.1 Hypotheses

- Farmers were aware of climate change and developed strategies for adaptation
- Rainfall and temperature extremes study to climate-is related to decisions and adaptation strategies of stakeholders and
- DSSAT model is a performance tool that can simulate millet grain yield in changing historical and future climate conditions for Sahelian and Sudanian zones of Mali.

1.4.2 Research Questions

To able to test the hypotheses, the research answered the following questions:

- What are farmers' perceptions about climate change and how have they been adapting to it?
- How has climate change affected rainfall and temperature extremes in the study areas?
- How will climate change and variability affect pearl millet production under different climatic conditions in Mali?

1.5. Organisation of the Research

This thesis comprises 7 chapters in total. After the introductory chapter (chapter 1), Chapter 2 presents literature review and describes climate change and variability and its impact on agriculture and economic, it also describes the generality of pearl millet and its potential yield in Mali.

Chapter 3 give the general methodology used for carrying out the research. It contains an overview of the study region, a brief description of RClimdex model and DSSAT model. Chapter 4 covers objective 1 by analysing farmer's perception and their adaptation strategies on climate change in Cinzana commune. Chapter 5 explores changing trend in the daily temperature and rainfall extremes. It aims at achieving objective 2. It established by analysing 5 indices for temperature and 4 indices for rainfall extremes.

Chapter 6 addresses objective 3, is the central of the thesis. It describes the field data, which formed the bases for DSSAT model modification and evaluation using independent data sets from 2 agroecological zones.

Chapter 7 presents general conclusion comprising concluding remarks and recommendations.



CHAPTER 2: LITERATURE REVIEW

2.1. Introduction

This section presents the climate change and variability context in two components. The first one focuses on the context of climate change in four sub-sections. It also gives some important definitions and clarification of basic terminologies often used for such assessment. According to Braimoh (2004), the concepts and definitions are essential since they facilitate comprehension and exchange among researchers and also improve understanding of research by broad readership. The selected terms are: climate variability and change, vulnerability to climate change adaptation; capacity for adaption to climate change and its effects. The second component trace the emergence of climate change sciences for understanding the context of climate change, since it is important to trace the trajectories of climate to the present time. The third section deals with the debates about the drivers of climate change. Panoplies of factors are cited as driving climate changes, which are crucial to understand the debates about change at a global scale as well as in Africa. This section also clarifies the difference between anthropogenic and natural causes of climate change. Observed and predicted climate changes scenarios are discussed in the fourth section in Africa and in Mali. The chapter presents also the climate change effects on agriculture and the economy at the international level, in Africa and Mali.

2.2. Climate Change in Perspective

2.2.1. Definitions of Terms and their Concepts

a) Climate Change and Variability

According to NSIDC (2010), climate refers to statistical weather information that describes the variation of weather at a given place for 30 years' interval. Weather is also defined as the state of

the atmosphere at a given time and places with regard to variables such as moisture, wind velocity, temperature and barometric pressure. Even though the difference between climate variability and climate change has been widely debated, their difference is often based on time and scale. Climate variability is defined as the fluctuations in the climate system over short time scales such as (months, years or decades) whereas climate change refers to changes in climate system over longer periods (3 decades or longer). This distinction was noticed in the definitions given by the IPCC (Waston, 2001), UNFCCC recommended other parameters of distinction between the two concepts (Pielke, 1998; Waston, 2001). UNFCCC (2002), however, presented a different distinction between climate change and climate variability. Climate change in this context is defined as change in climate as a result of direct or indirect human activity which alters global atmospheric composition. Therefore, climate change is in addition to natural climate variability observed over comparable time periods. The IPCC (2001) defines climate change as any change in climate over time, whether due to natural variability or as a result of human activity.

One of the most challenging problems currently facing the world is climate change (O'Brien and Leichenko, 2000; Houghton *et al.*, 1990). In the same respect, climate variability refers to the seasonal and yearly fluctuations in the rainfall and temperature patterns between and within regions or countries (O'Brien and Leichenko, 2000). Variability may be caused by natural internal processes within the climate system or to fluctuations in natural external forces and is driven by physical processes of the climate system (Waiswa, 2003). The difference between climate variability and climate change is that climate variability is related to seasonal variations of the climate while climate change operates over 3 decades or longer (Kristie *et al.*, 2005).

b) Vulnerability to Climate Change

Vulnerability to climate change has been defined as the degree to which a system is susceptible to, or unable to cope with, adverse effects of climate change, including climate variability and extremes. Vulnerability has been indicated to be a function of the magnitude, character and rate of climate variation to which a system is exposed, its adaptive capacity and sensitivity (IPCC 2007; Ekwe *et al.*, 2011). Here the system related to communities, admitting that communities are not equaled, so particular individuals or households within communities may have different degrees of vulnerability. Exposure to climate variation is firstly a function of geography. For instance, coastal communities will be affected highly by cyclones and sea level rise, while communities in semi-arid areas may be most affected by droughts. Sensitivity is the degree to which the communities are affected by climate stresses. For example, a community which depends on rainfed agriculture is much sensitive than where the main livelihood strategy is labour in a mining facility (Ekwe *et al.*, 2011).

c) Adaptive Capacity

IPCC (2001) defined adaptive capacity as the ability of the system to adjust to climate change including extremes to moderate potential damages and climate variability, to take advantage of opportunities or to cope with the consequences. Important climate adaptive resource may include firstly, local understanding of climate risk and conservation skills. Secondly, seed and grain storage facilities, physical irrigation infrastructure and farmer-based organisations and finally, diversified income sources, natural water sources and productive land (Ekwe *et al.*, 2011).

Adaptive capacity can vary over time depending on the variation of the conditions, and may change in relation to specific hazards (Ekwe *et al.*, 2011). It is noticed that people in the poorest countries are also less resilient to the effects of climate change due to the fact that they have limited access to those resources that would facilitate adaptation (Ekwe *et al.*, 2011). For example, women are often most vulnerable to the effect of climate change because of their responsibilities in the household and lack of information, resources and services.

d) Adaptation to Climate Change

This is defined as adaptation, and adjustment in natural or human system in response to actual or expected climatic stimuli or their impact which moderates harm or exploits beneficial opportunities (IPCC, 2007a). To reduce climate vulnerability, it is important to focus on building adaptive capacity, mostly for the less resilient people, and in some cases on reducing exposure or sensitivity to climate effects (Ekwe *et al.*, 2011). It must ensure that development initiative do not unintentionally make local population more vulnerable.

2.2.2 Climate Change Science

a) Climate Change Science Historically

The changes in climate was notified by scientists as early as the 1800s (Chipo, 2010). For more than a century, scientists recognized the sensitivity of the world climate to greenhouse gases in the atmosphere (Fleming, 1998; Weart, 2003). Furthermore, some scientists reported that greenhouse gases such as CO₂ and methane (CH₄) play significant roles in increasing the world's temperature (Fleming, 1998). Other scientists indicated that the use of fossil fuels were likely to cause global warming. Arrhenius in (1896) predicted that an increase or decrease in 40% of CO₂ concentration in the atmosphere might trigger glacial advances. This assertion was confirmed by IPCC (2007). A remarkable increase in CO₂ concentration and temperature in the atmosphere was observed in the 1900s (Callender, 1938), however, variation in temperature in the past has shown abnormal temperature increase in the past 50 years than the natural temperature variation in 1000 years (Callender, 1938; Chiotti and Johnson, 1995; Ohshita, 2007).

IPCC over the last decades has investigated in concluding the fact that climate change and global warming are most often caused by anthropogenic factors. This study was done through climate review involving national government and climate experts (Ohshita, 2007). IPCC concluded in their second and third report in 1995 and 2001 respectively that evidence so far indicate human activity have a great impact on climate change. IPCC (2007) in their fourth report again, stressed that greenhouse gases emitted through human activities is responsible for global warming. Furthermore, scope and magnitude of present climate risks are well known and actions urgently needed (Adger *et al.*, 2003).

b) Causes of Climate Change Debates

There is a significant debate concerning the cause of climate change, whether it is induced by human activities or simply with the range of natural variability. Therefore, there is agreement on the causes of climate change and a general opinion about future climate and the potential implications for agriculture (Chiotti and Johnson, 1995). Nonetheless, IPCC's usage of climate change refers to any variations in climate over time, whether caused by variability of nature or as a result of anthropogenic activities. This usage is different from that of UNFCCC where climate change addresses a change of climate that is attributed indirectly or directly to anthropogenic activity that alters the composition of the global atmospheric and that is in addition to natural climate variability observed over comparable time period (Boko *et al.*, 2007). Therefore, what is of importance is that in spite of the factors driving climate, both UNFCCC and IPCC concur there have been noticeable changes in the global climate over the last five decades (Chipo, 2010). *c) Natural Causes*

Heat storage in the atmosphere that causes climate change have been found out to be the natural cause of climate change. There has been natural accumulation of greenhouse gases in the

atmosphere throughout the earth's history which is also responsible for climate change. Some of the natural activities that have caused greenhouse gas emission are stratospheric aerosols and volcanic eruptions most of which occurred from 1880 to 1991 (Chipo, 2010). The amount of heat changes from the sun and how heat is stored in the oceans also contribute to climate changes.

d) Human Induced Causes

Many studies agree that climate change is slowly occurs the atmosphere which caused indirectly and directly by various human activities with natural climate variability over time. Nonetheless many researchers have come to the conclusion that climate change is mostly as a result of human activities (IPCC, 2000; 2007; Sutton, 2007). The contribution of greenhouse gases in the atmosphere by human activities is through removal of natural sinks such as forests or introduction of harmful gases into the atmosphere, through burning of wood and the use of fossil fuels and other activities.

The key greenhouse gases generated as a result of human activities include CO_2 , CH₄ nitrous oxide (N₂O) and tropospheric ozone (O₃). Carbon dioxide has been identified as the major greenhouse gas produced from human activities and it has the highest impact on climate change (Chipo, 2010). The largest growth in greenhouse emissions has been recorded between 1970 and 2014 which is caused by various human activities (Chipo, 2010). 270 billion of CO₂ has been released into the atmosphere since the start of the industrial revolution around 1750 (Chipo, 2010; IPCC, 2007; Marland *et al.*, 2000).

2.2.3 Observation and Prediction of Climate Change

a) Climate Change in Africa

In Africa, the warming of temperature is estimated approximately at 0.7 °C in the 20th century (Chipo, 2010). For example, in Africa in the tropical forests over ten year periods, there has been warming rate of 0.29 °C (Malhi and Wright, 2004) and in South Africa, 0.1 °C to 0.3 °C (Kruger and Shongwe, 2004). Africa is expected to rise in mean temperature ranging from 0.2 to 0.5 °C and inconsistent rainfall patterns (Hulme, 1996). This warming is more important in the interior of semi-arid margins of the Sahara and central southern Africa (IPCC, 2001). However, that a decrease in temperature was observed from weather station located near to the coast or to major inland lakes in Eastern Africa was observed (King'uyu et al., 2000). This implies that climate changes vary by location rather than ranking blanketing changes as being the same over the continent (Chipo, 2010). For example, there may be a global increase in average temperature, but these global warming impacts vary at the lower scale, by region and country (Chipo, 2010). Observations in Western Africa on the one hand, indicate there has been climate change in the form of a decrease in mean annual precipitation since the end of the 1960s (Chipo, 2010). For example, between 1931 and 1960 and 1968 and 1990, a decrease of 30 to 40 % was observed (Nicholson *et al.*, 2000; Dai and Trenberth, 2004). In the same way, in West Africa, there was a decrease in annual rainfall mean in the tropical rain forest zone of 4 %, 3 % in North Congo and 2 % in South Congo for the period from 1960 to 1999 (Malhi and Wright, 2004). On in the other hand, there has been an important increase in rainfall along Guinean Coast in the last 30 years (Chipo, 2010). Climate change impact with respect to temperature variation and inconsistent rainfall patterns just like most parts of Africa is similar to the situation in Mali. The impact of these challenges on farming activities is expected to be felt hard in these parts of the region owing to its geographical location.

2.2.4 Impact of Climate Change

Many studies have been carried out to give a better understanding of interwoven and complex sphere of climate change, and there is lack of information on the understanding of effect of climate change (Wheaton, 1994). In addition, there are knowledge gaps with regards to impact analysis despite a growing interest in the community of scholars (Tol *et al.*, 2004). At the local scale, impact knowledge is considered to be incomplete and uneven.

Analyses of climate change effects in the 1970s tended to be upon the impact approach, in a rather simplistic non-interactive model and one-way which attributed the impacts upon an exposure unit directly and only to variation in climatic conditions (Chiotti and Johnson, 1995). There was a shift in focus in the 1980s as some authors began to agree that climate change effects could be measured and occur throughout society and at various scales (Warrick and Bowden, 1981) and that climate change impact were also controlled by interactions with other factors (Garcia and Escudero, 1981). Although a number of studies have been carried out to determine the impact of climate change on global agricultural trade (Reilly *et al.*, 1994; Fischer *et al.*, 1994), neither of these studies considered agricultural interaction with other structural economic change or economic sectors that might influence agricultural production, even with climate change (O'Brien and Leichenko, 2000).

The impact of climate change may be experienced at different levels both negative and positive (Boko *et al.*, 2007) due to two reasons. Firstly, global circulation model project spatial differences in the direction of climate change and magnitude and, secondly, even in a given region experiencing the same characteristics of climate change, the impacts are likely to be different because some sectors, social groups or ecosystems do not have the same resilience to climate change than others (O'Brien and Leichenko, 2000). According to Rosenzweig and Hillel (1995) in the middle and higher latitudes, global warming will increase the length of the potential growing

season, leading to earlier planting of crops in the spring, earlier maturation and harvesting, and the possibility of completing two or more cropping cycles during the same season.

A study carried out by Seo and Mendelson (2006) revealed that the increase in temperatures is favourable to small farms that keep sheep and goats because of the facility to substitute animals that are heat-tolerant. Inversely, farms are more dependent on species such as cattle, which are very vulnerable to heat. Furthermore, effects or benefits can be identified for some social groups and some regions, but they are expected to decrease as the magnitude of climate change increases. Others identified adverse effects and expected increases in both extent and severity with the degree of climate change (Chipo, 2010)

In addition, in climate change discussion, policy and scientists are reluctant to recognize, address and discuss the existence of both negative and positive impacts, particularly the positive ones, for discussions are seen to be diverse and counter the efforts to gain a global consensus on climate change (Glantz, 1995; Schneider, 1989). For example, climatically the gradual change view of the future assumes that agriculture will keep on thriving and growing seasons will lengthen. North America and Northern Russia and Europe will prosper agriculturally while southern Africa, Europe, and Central and South American will be more affected by the increased heat, dryness, water shortage and reduced production. Global food productions under many typical climate scenarios increase (Schwartz and Randall, 2003). Nevertheless, it is important to engage in a negative and positive discussion on the effect from climate change, based on the assumption that farmers may benefit from the positive aspects and advantages from climate to improve their livelihoods. Moreover, assessment of climate effects inevitably points to winners and losers and the perception alone of the winning or losing groups can significantly influence climate negotiations (UNEP, 1993; Rosenzweig and Hillel, 1995).

a) Impact of Climate Change on Agriculture Case in Africa

The agricultural sector faces the challenge of increasing production to ensure food security for the projected human population of 2 billion by mid-century without harming the environment and the functioning of its ecosystems (Naab *et al.*, 2012). In the same respect, climate change constitutes an obstacle for stability of production and productivity since Africa is found to be the most vulnerable region to the effects of climate change and variability. Subsistence of smallholder farmers in Africa relying on rainfed farming systems are considered the most sensitive to the impact of climate change (Jarvis *et al.*, 2011).

Agricultural production is affected directly by climate variability since agriculture is linked to climatic parameters and is one of the most sensitive sectors to the risks and climate change impact. At the global scale, any significant change would impact agriculture at the rural scale most (Rosenzweig and Hillel, 1995; Parry *et al.*, 1999). In the past, it may be argued that African farmer have adapted to climate variations and future climate change may force wide region of marginal agriculture to stop producing (Mendelsohn et al., 2000). Climate change impacts on agricultural activities both on economic and physical sectors have been shown to be significant for low input farming systems, such as subsistence farming in poor countries of sub-Saharan Africa that are located in marginal areas and have the lowest capacity to adapt to climate change effects (Rosenzweig and Parry, 1994; Reilly and Schimmelpfennig, 1999; Kates, 2000; McGuigan et al., 2002). Already, thus, Africa is a continent under pressure from climate stresses and is highly sensitive to their impact (Chipo, 2010). Therefore, estimations show that on third of African people already live in drought prone areas and 220 million are exposed to drought yearly (UNFCCC, 2007). IPCC projects that yield from rainfed agriculture in some countries could decrease by the year 2020 and agricultural production and access to food is also projected to be strongly
compromised. About 250 million Africans could also face water shortages (Vermuelen *et al.*, 2008).

In Africa, agriculture is an important source of food and a prevailing way of life. WRI (1998), stated that an average of 70 % of the population lives by farming and 40 % of all exports are earned from agricultural products. Owing to this fact, economic growth and agriculture must rise in order to realise basic development goals (Chipo, 2010). The most challenge facing Africa is to increase agricultural production and achieve sustainable economic growth as both are important in improving food security. But in Africa, food security is already affected by climate extreme events, especially floods and droughts (Scoones, 1996; Kadomura, 1994).

Africa is projected to be the continent that will be more affected in spite of its least greenhouse gases (Naab *et al.*, 2012). Moreover, in Africa, several studies show a negative impact of climate change on crop productivity (Naab *et al.*, 2012). Future scenarios suggest that among the main world crops, African sorghum and millet production will witness, by far, the highest growth rate for the 2000 to 2050 period (Nelson and Stathers, 2009). A review of predicted effect on sorghum and pearl millet indicates 10 - 15 % losses that would be statistically significant by 2050 according to Knox *et al.* (2011). In addition, Roudier *et al.* (2011) used a meta database of future crop yields from 16 current studies, to evaluate the potential effect of climate change on yields in West Africa and reported yield changes ranging from -50 % to +90 % with a median yields loss of -11 %. The projected effects were wider in the Sudano-Sahelian countries (-1 18 %) than in the Guinean countries (-13 %) because of the drier and warmer conditions in the former countries (Naab *et al.*, 2012). The scientific evidence leaves little gap for doubt that climate is changing and it will have a significant impact on agriculture. This makes adapting agriculture to climate change and

variability an important element of agricultural research and development programmes (Naab *et al.*, 2012).

The IPCC reveals that areas of the Sahara are likely to emerge as the most sensitive to climate change by 2100 with likely agriculture losses of between 2 and 7 % of GDP. Central and Western Africa are projected to have losses ranging from 2 to 4 % and Southern and Northern Africa are expected to lose 0.4 to 1.3 % (Mendelsohn *et al.*, 2000). A study done in South Africa carried out by the University of Pretoria and focusing at the provincial level, found a significant correlation between greater historical temperatures and decreased dryland staple production, and forecast decrease in net crop revenues by as much as 90 % by 2100 and the study found small-scale farmers to be the most affected by the decrease.

b) Impact of Climate Change on Agriculture Case in Mali

Agricultural growth is affected by irregular and low rainfall, fragile and poor soils, as well as generally low productivity caused by the common use of traditional farming techniques (World Bank, 2007). The climate in Mali differs largely at the regional scale. The major productive region is Sikasso in Southern Mali with rainfall ranging from 800 – 1100 mm. Therefore, Malian agriculture receives around 400 – 600 mm of rainfall. Rainfall decreased significantly since 1950 (Olof, 2008). According to Ministère de l'Environnement et de l'Assainissement, (2006), rainfall and agricultural production are strongly related since up to 29 % of the value of GDP is added by agriculture to the economy.

A study on climate change impact on agriculture, temperature was projected to raise 1.7 °C with slightly decreased levels of rainfall by 2030 (World Bank, 2006). According to this same study, the main impacts include food insecurity and malnutrition and lower agricultural yields. Impacts

will be mostly felt by poor subsistence farmers. Other scholars reveal lower yield of millet, cotton, sorghum and rice (Butt and McCarl, 2006).

Except rice cultivation, agricultural productivity per hectare has been low or even decreasing for main crops such as sorghum, cotton and millet (World Bank, 2006). Agricultural growth outcome has mainly been achieved by increase of agricultural land, including marginal lands, reaching close to 7 % per year in southern parts of the country and thus leading to deterioration of ecosystems (Ministère de l'Environnement et de l'Assainissement, 2006). In southern Mali, the annual nutrient losses of soils have been rated to be equal to 40 % of incomes of farmers, (Van der Pol and Traore, 1993). Agricultural land has been lost being as much as 7 - 15 % to land degradation (PNUD, 2006).

In addition, predicted change in temperature and rainfall are likely to impact negatively on crop yield in both wetter and drier areas, since food production in wetter areas will primarily be impacted by the rising temperatures while changes in rainfall have wider impact in arid areas (World Bank, 2013a). An exhaustive literature reviews reveals that the median decline in crop yield in the Sahel caused by climate change is -18 % although predictions include wide differences (Roudier *et al.*, 2011). In the same respect, a study on Mali found that without adaptation the percentage of people found to be at risk of hunger rises from 34 to 64 - 72 % and the economic losses are estimated to be in the order of 1 % of GDP (Butt and McCarl, 2006).

c) Impact of Climate Change on Economies Case in Africa

Agriculture is crucial in supporting rural livelihoods and economic growth over most of Africa. Agriculture is the backbone of major African economies employing between 60- 90 % of the total labour force and accounting for as much as 35 - 34 % of the total export earnings (UNEP, 2006).

Regarding developing states, Africa counts the least and poorest developed nations of the world. But in Africa, even though some countries have been making good socio-economic progress but 41.1 % of the population earn less than \$1 per day, as compared to the situation in Southern Asia and respectively with 29.5 %, and 9.9 % in Eastern Asia (UN, 2007). Furthermore, Africa is the one continent in the world where per capita food production has been decreasing or stagnant at a level that is less than adequate (Scholes and Biggs, 2004). In the past 2 - 3 decades in sub-Saharan Africa land, degradation has relentlessly continued with 10 % of the major productive regions visibly affected as noticed from space (Vlek *et al.*, 2010). Nevertheless, Africa loses the equivalent of \$ 4billion per year because of nutrient mining (Naab *et al.*, 2012), and another \$42billion is lost in income and 6 million hectares of productive land is threatened each year caused by land degradation (Naab *et al.*, 2012).

Therefore, climate change impacts on the economy are far reaching and studies focusing on economic impacts of climate change in Africa are generally correlated to the performance of agriculture (De Wit, 2006; Calzadilla *et al.*, 2009). For most sub-Saharan African economies, agriculture is crucial in supporting between 70 % and 80 % of employment accounting for 30 % of GDP and at least 40 % of exports (Chipo, 2010). Whereas Asia experienced a rapid increase in food production and yield during the green revolution in the late 1970s and early 1980s, in subSaharan Africa per capita food production and yield have stalled (Calzadilla *et al.*, 2009).

According to many authors, in Sub-Saharan Africa this represent a cause for concern because 62 % of the population live in rural communities and depend mainly on agriculture. Local poverty accounts for 90 % of the total poverty in the region and approximately 80 % of the poor still rely on agricultural of farm labour for their livelihoods (FAO, 2005). National governments often struggle to furnish information on food security during times of crisis (Ayalew, 1997). For instance, Dilley and Heyman (1995) stated that, for international and national agencies, the cost of

climatic hazards may result in a shift in expenditure from dealing sensitively to simply dealing with immediate threats. However, irrigation development and improvements in agricultural productivity are major variables not only for future economic development, food security and poverty reduction in Sub-Saharan Africa, but also for climate change adaptation (FAO, 2008).

d) Impact of Climate Change on the Economy Case in Mali

Renewable natural resources and non-renewable natural resources of Mali play a crucial role for development as fisheries; livestock and agriculture which employ 80 % of the population (République du Mali, 2009). However, lower agricultural and forest productivity is caused by forest and land degradation and health cost linked with polluted water and to a lesser extent air pollution (République du Mali, 2009). These costs are mostly borne by sensitive groups that depend on rainfed agriculture and lack access to clean sanitation and water. Local agricultural growth is recognised as important for the decrease of poverty and food security (République du Mali, 2009). Growing rural agriculture is critical for poverty reduction and food security. In the medium term, loss of vegetation covers and land degradation is a main obstacle for Mali's ambition to be an agro-industrial power (République du Mali, 2009). The economy has been growing at around 5 % between 1996 and 2012 (Staatz et al., 2011). It is first and foremost caused by growth in the primary sector, food crops other than rice and to a lesser extent livestock that have been the motor of the Mali's economy in recent years (Olof and Cesar, 2013). The economy is projected to return toward a growth rate of around 5 % by 2013 but the uncertain economic crisis and political situation constitutes a significant risk for economic and social development (World Bank, 2013b).

Agriculture including fisheries, cropping, forestry and livestock was found to account for around 37 % of GDP in 2012, while services and industry contributed 23 % and 40 % respectively (Olof and Cesar, 2013). With the exception of 2012, the long term trend is a decrease in agricultural

share of economy caused by a rise in gold production, demographic shift toward urban areas and the normal structural transformation of the economy toward services and manufacturing (Olof and César, 2013). Although, investments increased in irrigation has somewhat decreased the sensitivity to rainfall changes as the link between agricultural production and economic development and rainfall remain very strong (IMF, 2006).

The growth in agricultural, forestry and livestock system has important potential in Mali. Local land is estimated at around 47 million hectares including 30 million hectares of grazing land, 12 million hectares of arable land, 3.3 million hectares of wildlife reserves and 1.1 million hectares of forestry reserves. For irrigation, 2.2 million hectares are favourable and the country has no negligible water resources (Djiré *et al.*, 2013), while only 12 % of the countries large irrigation capacity has been currently used (IFAD, 2011).

It has been observed that the rise in yields per hectare in millet and sorghum remain very low; the average yearly growth rate of yield per hectare of main food crops is 2.5 % which can be compared to a population growth of 3.6 % per year (Olof and Cesar, 2013). Also, it should be noted that this masks the wide gaps between low potential and high potential farming land and various yields for different farm sizes (Olof and Cesar, 2013). For instance, production of cereal per capita is often three times higher for those with bigger farms compared to those with smaller farms (Staatz *et al*, 2011). About 1 % of GDP represent cotton and 15% for export incomes. Cotton yields per hectare failed to increase between 2001/2002 and 2008/2009 (Olof and Cesar, 2013).

According to the Office du Niger, investment growths have been substantial which contributed to significantly higher yields per hectare. A sharp increase of land and water disputes has also emerged at the Office du Niger (Olof and Cesar, 2013). The investments have occurred in the context of an unclear land tenure system creating wide governance challenges and where

legislation to manage environmental and social impacts face significant challenges for implementation (Djiré *et al.*, 2013).

The creation of job through labour intensive public works can be an important way for reversing environmental deterioration and increasing investments in agriculture through plantation of forests, irrigation works and soil and water conservation (Bourdet, 2011).

2.3. Generality on Pearl Millet

2.3.1 Origin, Taxonomy and Geographical Distribution

Pearl millet is scientifically known as *Pennisetum glaucum* (L.) R. Br. It is a yearly, allogamous cross-pollinated and diploidia cereal, belonging to the *Poaceae* family, subfamily *Panicoideae*, tribe *Paniceae*, subtribe *Panicinae*, section *Pennicillaria* and genus *Pennisetum* (Rai *et al.*, 1997). The essential wild relatives of cultivars of pearl millet include the progenitor, *Pennisetum glaucum subsp. Monodii Maire*, *P. purpureum K. Schumach*, *P. pendicellatum Trim.*, *P, oriental Rich*, *P. mezianum Leeke*, and *P.squamulatun Fresen*. Previous were *P. Typhoideum L.C. Rich* and *P. americanum* (L.) *Leeke*. The four cultivated variety types/ species pearl millet are *typhoides* (generally found in India and Africa), *Nigritarum* (dominant in the eastern Sahel), *globosum* (in the western Sahel) and *leonis* (dominant in the West African coast) (Syngeta Foundation for Sustainable Agriculture, 2006).

Pearl millet was domesticated at the southern some edge of the Sahara 3000-5000 years ago from *Pennisetum violaceum* (Lam.) rich in the Sahel (Andrews and Kumar, 2006). It was used in East Africa and from there to southern Africa and the Indian subcontinent, about 3000 years. It reached the tropical Americas in the eighteenth century, and the United States of America in the nineteenth century. Millet is a common cereal in the semi-arid areas of West Africa and the drier parts of

eastern and southern Africa and the Indian subcontinent (Andrews and Kumar, 2006; Oumar *et al.*, 2008).

2.3.2. Morphological Characters

Millet (*Pennisetum glaucum (L.) R. BR*)) is a grass upright to which the stem, spinal cord without shortcomings, reaches a height which varies from 50 to 400 or even 500 cm, depending on the variety (Rachie and Majmoudar, 1980; Diop, 1999). However, in the semi-arid zone, the stem height does not exceed 300 to 380 cm and has a diameter of 1 to 2 cm (Zongo *et al.*, 1988; Amadou, 1994). The root system consists of shallow roots and others that can penetrate up to 200 cm in the soil, which explains the adaptation of this plant to dune soils (Boubacar, 1985; Sivakumar and Salaam, 1994; Alhassane, 2009).

The stem is rigid and has full internodes including those of the base which are the shortest. The nodes bases are able to give primary, secondary and tertiary tillers. Tillers are not all fertile: 1-7 tillers per plant usually fail to produce heads. Leaves alternate sheathing and parallel ridges are inserted on the stem at the nodes. They have a sheath enclosing the shaft and blade lanceolate. The ridges are well developed and prevent the blade to bend. The blade is a holder of stomata on both sides (Denis, 1984). The inflorescence is constituted of an apical panicle very dense and cylindrical in shape. Its length and diameter depend on the variety (15 to 140 cm in length and 0.5 to 4 cm diameter). The panicle (fake head) is formed of a rigid spine bearing spikelet stalked and grouped in clusters. Each spikelet has two flowers: the upper flower is hermaphrodite or female, generally fertile, while the lower flower is sterile or male. In general, there are few days' gap between male and female blooms which promotes cross-fertilization.

The fruit (sub-form of grain) is a caryopsis (globular elliptical) of about 4 mm length and of varying colour (whitish, yellowish or grayish). Seeds of millet are densely distributed on the head (Denis, 1984).

2.3.3. Biological Characteristics

Millet is an annual plant whose growing cycle can be divided into three phases: the vegetative and reproductive and the grain filling stages.

□ *Vegetative Phase*

The vegetative phase, which can range from 0 to more than 50 days after sowing (DAS), starts at the seedling emergence and continues until panicle initiation. Germination is hypogenous in millet and occurs about 48 hours after planting if conditions are favourable. The rise vegetative phase take place with the appearance of the first leaf, 4-5 DAS. As the raised ends, the buds of all the leaves have emerged and in early varieties, 6-7 leaves are already developed (Maiti, and Bidinger 1981). The seedling develops its primary root system and forms numerous adventitious roots. Tillering begins around 15 days and continues for 10 to 20 days for early varieties. It is generally longer for semi-late and late varieties, while remaining linked to the sowing date in sensitive varieties. Tillers belatedly produced generally do not form head; even if they give, they rarely reach maturity. During the active phase of tillering, the size of the plant remains small (rosette plant), since the elongation of inter-nodes is not yet started. During the vegetative phase, biomass accumulation mainly concerns the leaves and roots. It may also concern stems especially in case of early sowing of a photosensitive variety. The initiation of panicle is marked by the elongation of the apical dome and allows entry into the next phase.

□ *Reproductive Phase*

The reproductive phase is often observed around the 50th to 75th DAS. It starts with panicle initiation and often marks the beginning of start of bolting, which means the lengthening of the internodes of the stems from the base. Tillers begin the start of bolting in the same way as the main stem, but with a certain time gap. During the reproductive phase, the accumulation of biomass relates stems and panicles, in addition to the roots and leaves. Panicle spikelet develops on which emerging male and female flowers and seeds appear after pollination. Flowering occurs two to three days after the actual appearance of the panicle. Female flowers bloom earlier than males and their appearance is gradually from the top of panicle to the base. Five to six days later, flowering and pollination of panicle are complete (Maiti and Bidinger, 1981).

Grain Filling Phase

This phase begins after pollination of the flowers of main inflorescence, from the 75th DAS in early and semi-late varieties. It can peak before that date in ultra-early varieties and continues until the total maturity of the head (head of the main stem and tillers). During this phase, biomass accumulation mainly concerns caryopses and the leaves and stems of tillers may produce spikes. The biomass accumulation of grains (and grain filling) is often at the expense of older leaves and young non-productive tillers with dry senescence. Leaf senescence continues until the last 2 or 3 leaves to the top of the stem. The cowpea passes through a milky stage, a waxy or glass phase before reaching physiological maturity, about 20 to 25 days after flowering depending on the variety (Maiti and Bidinger, 1981).

2.3.4 Ecology

Millet is an annual herbaceous plant of the semi-arid areas. It is usually grown in areas with rainfall ranging from 200 to 1000 mm spread over 3-5 months, according to agro-ecological zones of the

WJ SANE NO

cultivable strip of Mali (Traore et al., 2000). Millet is adapted to the constraints of the Sahel. It has a strong ability to develop physiological mechanisms of tolerance to drought: slow water loss in the upper leaves, maintaining a favourable water level at the right grain filling stage (Winkel and Do, 1992). Less demanding as sorghum, millet is a cereal which is generally suited to light sandy soils and sandy clay. It is drought tolerant and more successful on low fertility level and at relatively high temperature of soils.

2.3.5 Mode of Reproduction

Millet is an annual species, diploid (x = 7, 2n = 2x = 14 chromosomes), sexed, a hermaphrodite, preferably cross-pollinated with a strongly influenced protogenos. Its pollination is an emophilous (Diouf, 2000). The length of the crop cycle from sowing to harvest can vary from 60 days to the earliest varieties to 180 days for later. The photoperiodic varieties behaviour determines the choice of location. The semi-late and late varieties are the most numerous in the Sudano-Sahelian zone. Early forms typically predominate in the Sahelian climatic zone (Clement *et al.*, 1993).

2.3.6 Production and Economic Importance

After rice, wheat, maize and sorghum, pearl millet is the fifth important cereal in the world. It is a largely grown rainfed cereal crop in Africa in the semi-arid regions and in Southern Asia. In other countries, it is grown under intensive cultivation as a forage crop. Pearl millet is primarily grown for grain production on 25 million hectares in the semi-arid and arid tropical region of Africa and Asia (Rai *et al.*, 2007). Millet accounts for half of global millet production with 60 % of the cultivated land in Africa, followed by Asian countries with 35 %. European countries represent 4 % of millet production and only 1 % in the North of America mainly for forage (Chitalu, 2013). Pearl millet is the third major crop in Sub-Saharan Africa, the main producing countries are

Nigeria, Niger, Burkina Faso, Chad, Mali, Mauritania, and Senegal in West Africa and Sudan and Uganda in the East (Chitalu, 2013). In the Southern Africa because of commercial farming, maize has partially displaced millet cultivation (Basavaraj *et al.*, 2010).

In terms of both area and yield, 9.3 million ha and production of 9.3 million metric tonnes with an average yield of 1044 kg ha⁻¹, India is the largest producer of pearl millet (Chitalu, 2013). The trend in area, production and productivity of pearl millet suggest that area has increased marginally (2 %) and productivity has gone up by 19 % (Yadav, 2011).

Millet is one of the most important crops in Mali, where the economy is mainly based on the primary sector. According to the revised agricultural sector, millet represented 19.7 % of total GDP in 1985, during the agricultural season 1994-1995, millet production has experienced the highest growth rate of about 27 % against 14 % for maize. During the period with other productions falling considerably (DNS, 1995). In Mali, millet and sorghum are cultivated on about 1.5 million hectares (75 % of arable land). Millet production in Mali has evolved as follows: between 1978 and 1981, production of Mali reached 461,000 tonnes, it has increased in 1988 and increased to one million tonnes in 1990, against sorghum production which has remained stable since 1988 between 670,000 and 750,000 tonnes. It has doubled in time since the late 1970s (FAO and

ICRISAT, 1997).

Millet is a cereal plant cultivated mainly for its seed which is used primarily (80-90 %) for human consumption in the Sahel (Ben *et al.*, 2002). Beside, millet as food for consumption, its stems and leaves are used for various domestic purposes including construction of hut, walls, fences and thatches, and the production of mats, baskets, brooms, sunshades, manure and compost and so on. Millet is an energetic, nutritious food, especially prescribed for children and the older persons or convalescents. The grain of millet contains about 10.6 % protein; 5.1 % fat; 66.7 % starch; 1.3 % crude fibre and 1,9 % of mineral elements (ROCAFREMI, 2002). The intake of vitamins from the

grains from millet is also appreciable. The grains contain about 0.22 mg vitamin A per 100 g of seeds (Yahaya, 2009).

2.3.7 Adaptation and Agro-Climatological Requirements

Tolerant to dry and hot climates, pearl millet is able to produce reasonable yield on marginal soil where other crop will fail. High salinity and fertility often constitute problems in millet producing areas, however, it performs relatively well on sandy acidic soil conditions, and when available moisture and soil fertility are low (Leisinger and Schmitt, 1995). This adaptation reflects pearl millet origin in the Sahel regions of Africa, where growing conditions are difficult (Myers, 2002). It is tolerant to sub-soils that are acidic (pH 4-5) and high in aluminum content (Oushy, 2010). The five climatic parameters crucial for pearl millet production are air and soil temperatures, rainfall, and day length (photoperiod), radiation and wind. The effect of these variables is dependent up on the developmental stage of the crop (Felch and Traore, 2010).

Pearl millet development can be widely divided into three growth stages (Begg, 1965):Growth stage one or sowing to panicle differentiation;Growth stage two, or panicle initiation to flowering; Growthstage three, or flowering to grain.

a) Rainfall Requirements for Pearl Millet Growth

Water availability is an important factor for the success of millet in the Sahel (Bley, 1990). Several studies have shown that water shortage in the cycle can reduce or even negate the positive effects of yield growth and thousands of techniques such as the application of nitrogen fertilizers (Donald and Hamblin, 1976; Alhassane *et al.* 2006), tillage (Chopart and Koné, 1994), or the use of improved varieties (Vaksmann *et al.*, 1996). However, the distribution and amount of rainfall are essential factors in determining the ultimate productivity of a crop. A rainfall of 350 mm, well

distributed over 75 days at a minimum, can give satisfactory harvest of millet; the efficiency also declines when the rain becomes too much (Yahaya, 2009). The onset of rainy season differs considerably in West Africa, while the end of rain is more defined (Kowal and Kassam, 1975).

During panicle initiation, or the period of the vegetation, the crop is well adapted to water shortage (Mahalakshmi *et al.*, 1988) and can tolerate intermittent breaks in rainfall, which are a common feature of the climate of millet-producing areas. In early flowering and grain filling stages, the crop is most vulnerable to water shortage (Mahalakshmi *et al.*, 1988; Mahalakshmi and Bidinger, 1985). Both the timing of the stress in relation to flowering and intensity of the stress determine the decrease in grain yield (Mahalakshmi *et al.*, 1988). Most of the variation among environment in a multi-location trial was caused by the availability of water during early grain filling.

b) Temperature Requirements for Pearl Millet Growth

Millet demands temperature and is damp cold dread (Yahaya, 2009). Several studies have been carried out over the years on the impacts of air/soil temperature on the germination, growth and yield of pearl millets (Ong, 1983a; 1983b). Pearl millet development begins at a base temperature around 12 °C with an optimum temperature ranging from 30 - 35 °C and the lethal temperature around 45 °C (Felch and Traore, 2010). The temperature base has been found to be fairly stable, regardless of the stage of development. The temperatures in the Sahel are mostly high due to the importance of radiation load and the scarcity of rainfall. Regarding the germination and emergence stage, soil temperature must reach 12 °C for germination to begin (Felch and Traore, 2010). The germination rate increases linearly with temperature to a sharply defined optimum of 33 °C and then decline sharply as temperatures increase (Ong, 1983c). High temperatures > 45 °C and soil surface crusting following sowing may also result in poor crop establishment because of seedling death (Soman *et al.*, 1987).

At the stage of development, the temperature requirements of pearl millet rely on the cultivar.

Diop (1999) revealed an optimum range of 22 - 35 °C for plant growth and a maximum of 40 - 45 °C, the optimum temperature for root elongation is 32 °C. A World Meteorological Organization (WMO) (1996) report on the agro meteorology of millet, states that millet requires a temperature between 22 - 36 °C for a good photosynthetic response, with an optimum ranging from 31- 35 °C.

Cantini (1995) declared that leaf appearance and expansion rate are positively correlated with temperature, and the leaf area index (LAI) increases linearly with temperature in the optimum range. A tiller appears sooner and they form rapidly as temperature rises to about 25 °C (Ong, 1983a; Pearson, 1975). Temperatures above 25 °C, the time of appearance of the first tiller does not change, there is a decrease in the number of tillers (Ong, 1983b; Begg and Burton, 1971).

Pearson (1975) showed that the rate of leaf production was hastened at high temperatures, it even the number of the leaf primordia on the main stem apex which does not change from 18 °C to 30 °C (Theodorides and Pearson, 1981). The duration of development phase is very vulnerable to temperatures averaging 18 days in length (McIntyre *et al.*, 1993). Each one degree rise in temperature reduces the length of the period by about two days. There is also no doubt that the number of grain produced is determined during the growth stage two, and the amount of radiation intercepted during this phase is more important than the interception after anthesis (Ong, 1983b). In the stage of reproductive growth, both the rate of the spikelet production and the duration of the early reproductive phase are very vulnerable to soil temperatures since the meristem are at or close to the soil surface (Felch and Traore, 2010). Setting grain is optimum from 22 - 25 °C and decline temperatures from above and below this range, whereas grain mass steadily declines with rising temperatures from 19 - 31 °C (Ong, 1983b). Plant exposure to prolonged periods of low temperatures < 13 °C during the booting stage levels to low grain set (Felch and Traore, 2010). During flowering, high temperatures result in a decrease of pollen viability of florets and pollen grain induced by lower temperatures (Mashingaidze and Muchena, 1982; Fussell *et al.*, 1980)

c) Day Length/Photoperiod for Pearl Millet Growth

Day photoperiod, or length is an important control in the initiation of the reproductive phase of millet in many pearl millet cultivars (Felch and Traore, 2010). Photosensitive cultivar is grown as long season crop, whereas non-photoperiodic cultivars are grown as short – season crops (Syngenta Foundation for Sustainable Agriculture, 2005).

The two main millet growing zones of the world are located in different latitudes, from 11 °N to 14 °N in western and central Africa and between 25 °N and 30 °N in north-western India (Felch and Traore, 2010). In these two zones, the length of the growing season ranges from 10 to 18 weeks (Virmani *et al.*, 1982; Kassam and Kowal, 1975). The length of the growing season is inversely related to the latitude and this relationship is more pronounced in West Africa, where season length changes often over a relatively small distance in latitude (Felch and Traore, 2010). However, the photoperiodic roles responses vary in the regions. In West Africa, the onset of the rains is strongly variable, whereas the end of the rains is clearly defined (Kassam and Kowal, 1975). The growth cycle of local millet cultivars changes greatly with sowing date because of photoperiod sensitivity. If sown in May or June, when days are long, the millet plant remains in the vegetative state until day length reaches an inductive threshold (Felch and Traore, 2010). When sown in August or under shorter days on one hand, the duration of the vegetative phase is very short, despite the fact that it is the minimum value that represents the intrinsic earliness of the cultivar (Vaksmann and Traore, 1994). Moreover, Kouressy et al. (1998) showed that the number of leaves and the total biomass are greater with early sowing because of the extended development period. However, Bacci et al.

(1998) indicated that this higher biomass yield is primarily caused by stalks and not to grain yield, in other words, greater biomass does not necessarily mean greater grain yields.

d) Wind Requirements for Pearl Millet Growth

Heavy winds associated with thunderstorms are frequent during the crop season in West Africa. These winds are laden with the particle of dust that decrease visibility and incoming amount and radiation quality; these particles form deposits on the surface of leaf that may influence photosynthesis (WMO, 1996). On the sandy soils in the southern Sahel, wind erosion owing to frequent sandstorms, particularly at the beginning of the rainfall season, is one of the obstacles to crop growth (Michels *et al.*, 1993). If sufficiently buried, these pockets of plant must be replanted. From partially surviving plants covered pockets show delays in development and growth (Felch and Traore, 2010). The maximum leaf number and plant height are lower, with an important decrease in the leaf area index. Grain yield from unaffected pockets is nearly twice that of pockets that are partially covered (Felch and Traore, 2010).

Scholars in northern Nigeria shelterbelt have shown that *Eucalyptus camaldulensis* shelterbelt positively influences yield of millet crops planted close to the belts (Onyewotu *et al.*, 1998). It was shown from experience that the shelterbelt would not have to exceed 100 m apart to fully exploit the protection of the crop from adverted dry and hot air (Felch and Traore, 2010). Differences in substantial yield is as a function of the distance from the belts and could be shown by soil moisture at the sowing and the impacts on crop growth conditions resulting from dry turbulent and heat air generated by the belts (Felch and Traore, 2010). A number of the factors that should be taken into account in the design and development of shelterbelts are described by Stigter (2005). The shelterbelts settled undulation and drifting sand encourages the return of soil protection grasses (Onyewotu *et al.*, 2003).

e) Solar Radiation Needs of Pearl Millet

Millet is a sun-loving plant (which requires direct solar radiation) in which light plays a key role, in both the morphogenetic processed of growth on the determinism of the flowering. Indeed, the length of insolation or photoperiod is a real source of variability in flowering time-sensitive varieties (Alagarswamy *et al.*, 1998; Alhassane, 2009).

The incoming amount of radiation sets the limits for dry matter production. It has two roles in crop production; a segment of total radiation is called photosynthetically active radiation (PAR), which is required for photosynthesis (Felch and Traore, 2010).

Pearl millet is a C4-type plant, meaning it has a high photosynthetic efficiency, especially under high temperature condition, due to the decrease of photorespiration (WMO, 1996). The photosynthesis efficiency depends, on the age of leaves, the degree of their direct exposure to sunlight and genotype (Flech and Traore, 2010). Where direct sunlight is very important both in determining the flowering and in the morphogenetic process of growth of pearl millet. Within the plant cover, the redistribution of solar radiation involves plant architecture, leaf area density, planting density and leaf angle (Bégué, 1991). Several studies have been carried out to determine the radiation use efficiency of pearl millet (Bégué, 1991; McIntyre *et al.*, 1993). The efficiency conversion varies with the stage of development, being highest during tilling and then gradually declining as the crop matures (McIntyre *et al.*, 1993).

2.4 Potential Yields

Potential yields are yield of cultivar when grown in environments it is adapted to, with water nonlimiting and nutrients and with diseases, pests, weed and lodging, and other stresses effectively controlled. As such, it is observed from potential yields, which can be defined here as the maximum yield which could be reached by a crop in a given environment, as determined for instance by simulation models with plausible agronomic and physiological assumptions (Evan and Fischer, 1999).

World grain yield records have reached 25 t/ha for maize, 12.5 t/ha for sorghum and 6 t/ha for pearl millet. They were reported in the USA for all three cereals, as well as in India for pearl millet. In West Africa, maximum yields are far below these records. Research stations have typically recorded grain yields of 4 t/ha for sorghum, 6 to 7 t/ha weighted with early caudatum / kafir varieties. For pearl millet, grain yields of 3.2 - 3.5 t/ha have been reported respectively in Nigeria and Mali (Clerget and Traore, 2009).

During 2008, Malian farmers have planted millet (Pearl millet (*Pennisetum glaucum (L.) R. Br.*) on 1,615,450 ha and sorghum (*Sorghum bicolor L. Moench*) on 986,367 ha with respective average yields of 768 kg ha⁻¹ and 943 kg ha⁻¹ (FAOSTAT, 2008). In the Sahelian zone of Cinzana in Mali, these crops are the main staple crops and yields were lower than their reported potential (Coulibaly, 2010).

2.5. Other Background Information

2.5.1 Diseases, Pests and Weeds

The major deterrents to pearl millet production are downy mildew, *Striga*, smut, ergot and rust, with the first two being far the most important (WMO, 1996; Syngenta Foundation for Sustainable Agriculture, 2005).

Striga is a parasitic weed that originates and has major problems across many countries of Africa and parts of Asia (Felch and Traore, 2010). In Africa, 21 million hectares are said to be infested by *Striga*, leading to an estimated grain loss of 4.2 million tonnes each year (Sauerborn, 1991). It competes with the pearl millet plant for both nutrients and water. Therefore, low rainfall and low

soil fertility favour *Striga* infestation, however, it can be partially controlled by pre-treatment of seeds with herbicides that decrease or prevent the germination of the *Striga* seeds (Felch and Traore, 2010).

Downy mildew (*Sclerospora graminicola (Sacc*) is the most important disease risk to the successful cultivation of pearl millet (Web Based Information Service for Agriculture Research for Development (WISARD, 1999), especially in India. 30% of the harvest in India can be lost during years of serious attacks, with losses in individual fields reaching nearly 100 % (CGIAR, 2006). Despite of the fact that pearl millet hybrids often provide better in grain yield than local open pollinated cultivars, the genetically uniform single cross hybrid cultivars presently available in India are more sensitive to epidemic of pearl millet downy mildew (Felch and Traore, 2010).

Smut is a panicle disease which attacks the flowering head of the pearl millet plant. The main source of inoculum is spore balls in the soil from previously infected crop residue and surface contaminated seed used for sowing (Tharkur and King, 1988). Moderate temperatures $(25 - 30 \degree C)$ rather than cool temperatures, high relative humidity > 80 % and long days seem to be suitable for disease development (Thakur, 1990; Kousik *et al.*, 1988).

Rust is a foliar disease. Occurrence of the disease during the seedling stage can result in substantial decrease in grain and fodder yield and quality (Felch and Traore, 2010). High humidity and cooler temperatures favour disease development (Singh and King, 1991). When *rust* appears late in the season, grain yield may not be affected but the plant fodder is used to feed animals after harvesting of grain (Felch and Traore, 2010). The causes of disease are a severe decrease in digestible dry matter yield of forage (Felch and Traore, 2010). Of all the environmental parameters evaluated, only average temperature below 27 °C were consistently correlated with the onset of *rust* epidemics (Panwar and Wilson, 2001).

2.5.2. Insect and Pest

Relatively, pearl millet has few insect pest problems. In the Asian subcontinent, white grubs are the main pest (Rachie and Majumdar, 1980). In West Africa, there is a range of insect pest that destroy the crop, resulting in economic losses; the main ones are the ear head caterpillar (*Raghuva*), stem borer (*Acigona*), midge (*Geromya penniseti*) and many species of grasshoppers. In West Africa, pests' surveys show that crops are affected by infestation of ear head caterpillars (Felch and Traore, 2010). The number of surviving diapausing pupae emerging from the soil is correlated with soil temperature and moisture at different depths from November to May. Furthermore, there is a close relationship between onset of rain and soil moisture and month of emergence, which are the key factors in diapause termination (Felch and Traore, 2010). Soil moisture content increasing and lower soil temperature in the upper soil layer are linked to earlier termination of diapausing pupae in the soil layer (Nwanze and Sivakumar, 1990).

Rainfall onset time and amount during the crop season is linked to the stem borer population (Nwanze, 1989). There is a need for knowledge of diapausing populations and the relationship between insect pest and rainfall during the season in regions where sporadic outbreaks happen in order for the weather parameter to be related to the population dynamics of the pest (Felch and Traore, 2010).

2.5.3 Drought Tolerance

A major aspect of pearl millet ability to survive under high stress is the deep root penetration. Pearl millet roots can go down up to 180 cm deep, with approximately two thirds of the root system in the top 45 % of the soil zone (Mangat *et al.*, 1999). This root depth penetration can help millet species to exploit soil water more effectively and therefore survive drought stress, in the same time pearl millet root system also have the capability to penetrate through hard clay pans in the soil

(Felch and Traore, 2010). Furthermore, the photosynthetic rate is maintained through periods of serious drought (Zegada-lizarazu and Iijima, 2004). Pearl millet has a monocotyledonous behaviour type of root system made of a primary or seminal root, adventitious root from the nodes and the base of the stem, and the crown root from many lower nodes at below the soil surface. These lateral roots develop with four days after the radicle comes out and helps in the first establishment of the seedling. The seminal root is active up to 45- 60 days, after which it starts to degrade (Felch and Traore, 2010).

Concerning the structure of leaf, stomata are seen on both leaf surfaces. The colour of the leaves is vary from light green to yellow to deep purple. The maximum leaf area occurs at the time of 50 % flowering, when the majority of the tillers have produced leaves. After flowering, there is a decrease in leaf area and at the same time the leaves beginning senescing (Felch and Traore, 2010). Physiological maturity only on the upper 3 - 4 leaves can be green on the main stem (Mangat *et al.*, 1999).

Stomatal sensitivity to evaporative demand is dependent upon leaf area and age of the crop (Felch and Traore, 2010). This means that the degree to which water use is controlled by stomata and leaf area is influenced by the way ontogeny has to optimize the growth of water use (Winkel *et al.*, 2001). The conception is that millet does not tend to keep moisture but rather transpire freely as long as the root system is able to provide the water needed (Wallace *et al.*, 1993).

2.6. Gaps in Knowledge

From the literature research, the following gaps in knowledge were found that there is lack of scientific information on farmers' perception of climate change and variability in Cinzana. In addition, in Mali, limited knowledge relating to rainfall and temperature extremes study for climate related to decisions making. And, there is very little literature on the validation work of the DSSAT

Model on Pearl millet potential yield in the Sudanian and Sahelian zone of Mali using long term scenarios (30 years).

The gaps were tatted by extracting farmers' perceptions of climate change using questionnaires, assessing rainfall and temperature trends to compare with farmers' perceptions, modelling effect of climate change and variability for millet grain yield with DSSAT crop model in tandem with field experimentation millet performance under historical climatic data and different scenarios from 2040 to 2069.

CHAPTER 3: MATERIALS AND METHODS

3.1. Introduction

The present chapter gives details on the study area, an overview of the data used and the methods applied for data gathering and processing as well as the models employed for the assessment of rainfall and temperature changes and assessment on changes in potential yield of millet crop under a scenario of climate change. To achieve the planned objectives, field surveys and field trials were conducted. The experimental trials were conducted during 2013 and 2014 in the rainfall season at Cinzana and Sotuba. The details of all fields work are also described.

3.2. Study Area

The study was performed in two agro-ecological zones of Mali as defined by Traoré *et al.* (2010). The two zones involved in implementing the activities were at:

ANF

- Sahelian (Cinzana) and
- Sudanian (Sotuba).

In each agro-ecological zone, sites were basically chosen, on the basis of representativeness and data availability, for implementing the activities. These zones were also, deliberately selected because of their main economic activity. These agro-ecological zones have more than 60 % of the population engaged in agriculture and characterized by a single, relatively long rainy season from April to early July and ending between September and October. Key food staples include millet, sorghum, maize, rice, cowpea and groundnut produced predominantly under smallholder production systems. The agriculture sector is affected negatively by climate variability, including drought, flood, heat waves and other extreme weather events. Consequently, cropping systems are characterized by low production due to erratic rainfall, poor soil fertility and poor crop management.





Figure 3. 1 Study Area showing Sotuba (left) and Cinzana(right)

3.3 Cinzana Commune

Cinzana (Figure 3.1) is a village and rural Community in the Region of Segou in the Segou Region of southern-central Mali. Etymologically, Cinzana is the distortion of "Zele and Zan" Zele being women and Zan her husband. The name was assigned to the village of Cinzana by the population coming from Séribougou village (Cinzana Community). There is a research station situated 5 km from the Cinzana village. Research activities are performed at the Agricultural Research Station de Cinzana (CRRA). It is one of eight (8) major research stations of the Institute of Rural Economy (IER) and reports of the Regional Agricultural Research Centre (CRRA) Niono, one of six regional research centres of IER since its restructuring in 1992. In 1979, the initiative was born of a creation for the benefit of small farmers, a research station focused on the promotion of farming systems based on millet in the Sudano-Sahelian zone. Inaugurated in 1983 and operating to date, the station has devolved and experienced a diversification of its partners, including Ciba-Geigy, USAID, ICRISAT, IFAD, IITA, Netherlands, European Union, SYNGENTA Foundation. Cinzana Gare is located in the rural town of Cinzana 45 km from Segou and 5 km off the national paved road (Axis Segou-Mopti) on longitude 5 ° 57 'West and latitude 13 ° 15' North and at an altitude of 280 m.

3.3.1 Physical Characteristics

a) Climate

From a climate perspective, Cinzana Community is situated on latitude 13° 15N and on longitude 5 ° 57 W which makes it a Sahelian zone type semi-arid area. The mean characteristic feature is the change of a long dry season from November to May and a rainy season from June to October. The range annual rainfall of the last ten years is 600 to 700 mm. This amount is unevenly distributed over the period, is insufficient for the needs of cropping which largely justifies the successive years of crop failure. The average temperature ranges between 28 °C and 33 °C with 39 - 40 °C high and thresholds of 8 ° - 12 ° C.

b) Vegetation

Cinzana vegetation is dominated by tree strata, shrubs and grasses. The tree layer includes species rarely reaching 20 m. The species distinguished there are: *Acacia albida, Borassus aethiopium,*

SANE NO

Tamarindus indica, Bombax costatum, Combretum glutinosum, Pterocarpus lucens, Adansonia digitata, Vitellaria paradoxa and Parkia biglobosa. Among the Poaceae (Perennial grasses) are: Andropogon gayanus, Hyparrhenia dissoluta, Cymbopogon giganteus, Ctenium newtonii and Loudetia simplex. Among the annual grasses are distinguished: Andropogon pseudapricus, Eragrostis Tremula and Ctenium elegans. The flora is composed of a slightly degraded wooded savannah. The fauna is mostly represented by small species (hares, squirrels, birds). c) Soil Kouyate et al., (2014) and Kablan et al., (2008) reported that soil of the area is classified as a leached tropical ferruginous soil with spots and concretions and Alfisols according to Soil Taxonomy (SoilSurvey Staff, 1999), with many Paleustalfs and frequent Plinthustalfs. The Ustalf classification indicates that the soils are, indeed, highly weathered and highly leached. The classification of Plinthustalfs is of special concern because it indicates that the soils contain a plinthite layer of soft Fe and Al oxides that will harden irreversibly into lateritic stone if exposed.

3.4. Sotuba Commune

Regional Agricultural Research Centre (CRRA) at Sotuba (Figure 3.1) is located on the left bank of the Niger River of the Bamako District. It is bounded to the east by the village of Sotuba, to the west by the industrial area south of the Niger River and to the north by the districts of Djélibougou and Boulkassoumbougou. Taking the Bamako District topo-sequence, the Sotuba Agronomic Research Station is located in the lower glacis and on the floodplain of the River Niger, and between two sandstone series.

First, the sandstone series so-called Sotuba is buried under the alluvium of the plains. This emerging in the bed of the River Niger consists of Cambrian sediments located below the first set with a tabular morphology. Along the river ends the perimeter by a water catchment with semipermanent flooding regime. Second, the sandstone - shale series of Koulouba limit the alluvial

plains in the north and comprises several layers of various sandstones intercalated with two beds of shale. These sandstones are overlain by powerful ironstone.

The topography is gently undulating with an average downward slope from north to south ranging from 0.3 to 0.5 %. It is generally flat in the east direction except for the corner -west South -east of the resort where the slopes exceed 0.5%, and the area is highly sensitive to erosion. The existing topography and layout of the station Sotuba allow the development of fields in regular plots.

3.4.1. Physical Characteristics

a) Climate

The station is located in the Sudano-Sahelian zone typically characterized by semi-arid tropical rainfall. The average annual precipitation is about 969 mm. Rainfall mainly occurs between April and October. The wet best periods occur in August and September. During this period the normal is 883.5 mm. The highest rainfall of the station was recorded in 1954 reaching 1,425 mm. The lowest was 591.9 mm in 1996. The frequency of rainfall intensities for short periods is available to the station, and 90 mm / h intensity. The maximum temperature varies between 30 °C and 40 °C during March and April, the minimum temperature varies between 24 and 14 °C and occurring between December and January. The humidity varies between 19 and 78 % and is higher during the rainy period. The hours of sunlight vary between 5 and 9 hours per day and are longer during the dry period.

b) Vegetation

Natural vegetation hardly exists in the areas under cultivation. It was destroyed by continuous cultivation and other uses; but fallow exist in the south-western area of the station. Some observation showed *Kaicedra, Vitellaria paradoxa, Thorny, Parkia biglobosa, Fficus Guierra Senegalensis*, and several species of herbs. The density of trees and shrubs varies across the station;

the density of trees represents a significant obstacle to the normal course of farming operations. Areas including a high density of mature trees are excluded from the court in search areas, but will be developed for general production goals and grazing as they do not affect normal agricultural activities. Trees located in or around research areas is habitat for birds and shade for experimental crops.

3.5. Perception of Famers on Climate Change/ Variability and their

Adaptation

3.5.1. Target Population and Sampling

A study on evaluation of farmers' perceptions of climate change and their adaptation strategies was conducted at Cinzana, a village located in the Sahel zone and Kassela and Baguineda located in South-East in Sudano-sahelian of Mali. A series of on-farm surveys were conducted. One hundred and nineteen farmers' household (119) including women and youth were randomly selected in the villages and were interviewed using structured questionnaires. Information in the structured questionnaire was collected on faming systems, perception on climate change and the concerns for climate change. Data were also collected through a field survey using face to face interviews with farmers. The questionnaire was designed in French and English but interviews were conducted in the local language, *Bambara*.

Interviews were conducted in November 2013. Before being administrated to all the interviewers, the questionnaire was pre tested for is suitability. Precipitation and temperature data were also obtained from the Mali Meteorological station over the period 1961 to 2014. The data gave an overview of the trends of precipitation and temperature and the vulnerability of the region to drought.

3.5.2. Data Analyses

Data from the questionnaire were entered and analysed using the MS Excel and the Statistical Package for Social Sciences (IBM, SPSS version 20) software. Descriptive statistical tools such as, percentages, mean, frequencies and standard deviations were used to summarize and categorize the information gathered. Cross tabulation was used to compare group's means. They were used to identify the differences that prevail between different categories of farmers such as age, gender, level of education and their adaptation strategies. Analysis of correlation was used to determine whether household income rather than education level significantly influenced adaptation strategies to climate variability and change. For the climatic variables analysis, the annual rainfall average was calculated and index of Nicholson (RI) which is a reduced centre variable calculated on annual precipitation to highlight fluctuations in rainfall patterns. It is the ratio of the difference in inter-annual to average standard deviation of annual precipitation amounts. Nicholson indices play a very important role in determining precipitation fluctuations (wet, normal and dry period).

3.6. Assessment of the **Trends in Extreme Precipitation and Temperature** Indices for Sotuba and Cinzana

Historical daily rainfall and temperature data for Segou and Senou meteorological station were analysed for significant trends. The latitude and longitude of the stations are 12. °53 N and -7. °95 and 13. °40 N and -6. °15 W respectively and the elevations are 320 m and 289 m respectively. The length of historical observation data record available for analyses was 53 years covering the period of 01/01/1960 to 31/12/2014. Climate data were obtained from Tu Tiempo World Weather - locale Weather Forecast at <u>http://en.tutiempo.net/climate</u> and NASA POWER a renewable energy resource web site of global meteorology and surface solar energy climatology from NASA satellite data on 1 by 1-degree resolution

(http://power.larc.nasa.gov/common/php/Agro_ExSummary.php).

Extreme temperature indices were computed from minimum and maximum daily and mean, rainfall and temperature indices from daily rainfall data. The data were from the AGRHYMET Regional Centre of Niamey Niger and RClimdex software was used to compute the indices.

3.6.1. Tu Tiempo and NASA POWER data

Tu tiempo data and NASA POWER data were obtained through the World Wide Web, http://en.tutiempo.net/climate http://power.larc.nasa.gov/common/php/Agro_ExSummary.php

Climate variables include in the dataset for each station included mean and daily temperature (minimum and maximum); amount of precipitation; wind speed and visibility; relative humidity and insolation. So far in this study unit, temperature and rainfall was used to fill the gap for time series data obtained at AGRHYMET Regional Centre.

3.6.2 Brief Description of RClimDex

ClimDex was developed by Byron Gleason at the National Climate Data Centre (NCDC) of NOAA, and has been used in CCl/CLIVAR workshops on climate indices from 2001. It is a Microsoft Excel based programme that provides an easy- to use software package for the indices calculation of climate extremes for monitoring and detecting climate change. The objective was to port ClimDex into an environment that does not depend on a special operation system. It is very natural to R as our platform, and it is a free and yet very robust and powerful software for statistical analysis and graphics. It runs under both UNIX and Windows environment. It was discovered in 2003 that the method for computing percentile-based temperature indices in ClimDex and other programme resulted in inhomogeneity in the indices series. A fix to the problem requires a bootstrap procedure that makes it almost impossible to implement in an Excel environment. This has made it more urgent to develop this R based package. RClimDex (1.0) is designed to provide a user friendly interface to compute indices of climate extremes. It is composed of 27 core indices recommended by CCI/CLIVAR Expert Team for Climate Change Detection Monitoring and Indices (ETCCDMI) as well as some other temperature and precipitation indices with user defined thresholds. The 27 core indices (Table 3.1) include almost all indices calculated by RClimDex (Version 1.3). This version of ClimDex has been developed under R 1.84 and should be run with R 1.84 or a later version.

The main objective of constructing climate extremes indices is to use them for climate change monitoring and detection studies which require the homogeneity of the indices. Data homogenization has been planned but is not implemented in this release. Present RClimDex only includes a simple data quality controlled before the indices can be computed, this users' manual provides step-by-step instructions on

- the installation of R and setting up the user environment,
- quality control of daily climate data,
- calculation of the 27 core indices.

Climate indices details computed and analysed for presence or absence of trends and seasons considered in this study are explained in Chapter 5.

3.7. Assessment of The Changes on Yields of Millet Crop under Different Scenarios of Climate Change Using DSSAT Model

3.7.1. Experimental Design and Treatments

Two varieties of millet were planted with all cultivation practices in two agro ecological zones of Mali; Sotuba in the Sudanian zone and Cinzana in the Sahelian zone. Latitude and longitude for Sotuba are latitude 12° 39 N and longitude 07° 56 W Cinzana commune is situated on latitude 13° 15N and on longitude 5° 57 W. Daily records of weather parameters like temperature (maximum and minimum) and rainfall amount were collected from each station respectively.

The experiments were fully rainfed. Trials were thinned to two plants per stand by using hand pull up to 14 days after sowing (DAS). At 21 days (DAS), the first weeding was done by the use of hoe thereafter weeding was carried out at two week intervals until the plant reached the flowering stage. Four soil nutrients amendment applications were used with control which had no application, organic manure (OM), NPK and combination of NPK and OM. Phenological data from the time the millet was planted to harvesting were obtained in the following order. Date of planting, date of emergence (50%), dates of maturity, date of harvest. Yield and yield parameters were taken: number of panicles per plot at harvest, panicle weight per plot at harvest, straw weight at harvest, 1000 grain weight per variety and grain weight per plot at harvest.

The experimental design and treatments are described in detail in Chapter 6.

3.7.2. Brief Description of Crop Model 'DSSAT'

Decision Support System for Agro Technology Transfer (DSSAT) is a software package that integrates the effect of crop phenotype, soil, weather, and crop management system through a database system and allows users to simulate experiments on desktop computers in a minute, which would take significant quantity of time to conduct. DSSAT enables the user to study the "what if" results of different management options and strategies through its different independent programmes that operate together (Jones *et al.*, 2003). These programmes include crop sub modules and databases that describe weather, soil, experimental conditions and measurements, and genotype information (Figure 3.2). The software also enables users to prepare inputs for each of the programmes and compare simulation results with observation, giving users confidence in the models or determine possible modification to achieve improved accuracy. In addition, DSSAT programmes allow users to assess risks associated with different crop production strategies through its multi-year simulation option. Conversely, DSSAT also has a built-in function to specify changes in weather variables without directly modifying the original weather file which suits it for climate change impact studies. In recent updates of the software, it can also directly read historical atmospheric carbon dioxide data from Mauna Loa, Hawaii (Hoogenboom *et al.*, 2010).

DSSAT was developed through collaboration between scientist and researchers of different institutions such as University of Florida, University of Georgia, University of Hawaii, University of Guelph, International Centre for Soil Fertility and Agricultural Development, Iowa State University, and the International Consortium for Agricultural Systems Application (Tsuji et al.,

1998; Hoogenboom *et al.*, 2003; Jones *et al.*, 2003). DSSAT consists of different crop models (Jones *et al.*, 2003) such as CERES-Maize (Jones and Kiniry, 1986) for maize, CERES-Rice (Hoogenboom *et al.*, 1994) for rice, CANEGRO (Hoogenboom *et al.*, 1994) for sugarcane,

PNUTGRO (Boote et al., 1986) for peanut, and a model for tomato (Hoogenboom et al., 1994).

Actually, DSSAT crop models have been cited by UNFCC (UNFCC, 2007) as a tool which can be combined or integrated with other tools or methods to evaluate impacts, vulnerability and adaptation to climate change. Details of the simulation in DSSAT (Figure 3.2) computed and analysed for the millet crop are described in Chapter 6.



52





4.1. Introduction

Climate change causes extreme weather and unpredictable events which impact and increasingly affect crop growth, availability of soil water, soil erosion, droughts and dry spells, floods, sea level rise with prevalent of pest infestations and diseases (Adejuwon, 2004; Zoellick and Robert, 2009). Many studies have shown that climate change will highly affect the African continent and will be one of the most challenging issues for future economic development, particularly in the subSaharan Africa (Roudier *et al.*, 2011). Because most of the population rely on natural resources, they are

often practical affected by climate variability and change, especially the poorest (Morton, 2007). Particularly, smallholder farmers in sub- Saharan Africa are strongly impacted (Sivakumar *et al.*, 2005). Local societies already have in-depth knowledge of local climate change and variability as parts of their local ecological knowledge, obtained and transferred through generations (Berke *et al.*, 2000). Several scholars on climate change perception deals with precipitation and temperature in terms of the annual amount, the length and annual distribution of rainfall (Deressa *et al.*, 2009; Fisher *et al.*, 2010). Meteorological data are used to confirm local farmers' assessment of climate change. However, some authors emphasized the need to consider climate in a wider context such as in health or polices (Mubaya *et al.*, 2012; Shackleton and Shackleton, 2012).

According to Parrry *et al.* (2007), climate change and agriculture are interrelated processes, both of which take place on a global scale. It is predicted that global warming will have important effect on agriculture (McCarthy *et al.*, 2001; Funk *et al.*, 2008). These determine the biosphere's capacity to produce enough food for the human population and domesticated animals. Overall impact of climate change on agriculture will depend on changes in average temperatures, rainfall, and changes in pests and diseases; (Fisher *et al.*, 2002). Studies show that Africa's agriculture will be affected negatively by climate change (McCarthy *et al.*, 2001; Dinar *et al.*, 2008; Pearce *et al.*, 1996). About 25-42 % of species in Africa could be lost, affecting both non-food and food crops. Habitat change has already been observed in some areas leading to species shifts, changes in plant diversity including plant-based medicine and food (McClean *et al.*, 2005). Eleven per cent of arable land will be affected by climate change in the developing countries, which might lead to a decrease of cereal production in up to 65 countries and about 16 % of agricultural GDP (FAO, 2005).
Climate change will hardly harm the agricultural sector in West Africa if any adaptation actions are not taken (Rosenzweig and Parry, 1994; Adger *et al.*, 2003), but the negative impact of climate change can significantly be reduced if appropriate adaptation practices are adopted (Waha *et al.*, 2013). The type of adaptation practices and strategies to be undertaken is to some extent first determined by famers' perceptions of climate change (Roncoli *et al.*, 2001; Thomas *et al.*, 2007).

It is widely acknowledging in the Intergovernmental Panel on Climate Change (IPCC, 2007) that global climate is changing. In the same way, there is general consensus that the Sahel will get hotter as a result of global warming. The temperature in the Sahel will increase by 1.2 °C in 2030 as compared to current temperatures. (Butt et al., 2006). The Sahel is expected to heat up more than the rest of the globe because in-land areas will become warmer than temperatures over the oceans. In the Sahel, temperatures are already close to the maximum for plant growth, especially at the starting of the season. Experiments with different levels of shading showed that temperature has a pronounced effect on millet production in the Sahel (Vandenbeldt and Williams, 1992). There is therefore, the need to know farmers perceive climate, changes in climate and the environment (Kemausuor et al., 2011). In fact, perception of farmers of climate variability and how these perceptions determine the choice of coping or adaptation strategies (Vedwan, 2006), have been investigated by previous studies in West African Sahel (Akponikpe et al., 2010; Kyekyeku, 2012; Zampaligré et al., 2013; Sanfo et al., 2014; Traore et al., 2014, Kima et al., 2015). Climate change confirmed by most of the farmers up to 98 % of respondents was dependent on the geographical area and prevailing climate across five 5 countries of West Africa (Akponikpè et al., 2010). In the Sahel, more proportion of farmers perceived the change to have started between 20-30 years ago or more while the majority of them perceived it to be less than 10 years ago in the Guinean areas. Farmers in Burkina Faso understood climate change variability primarily based on weather-crop interactions and on events that are associated with climatic fluctuations. Their perceptions were additionally shaped by the cultural frame. Farmers mentioned more erratic rainfall patterns, decreased rainfall amounts, increased temperatures, winds and radiation.

In Mali, small scale rural farmers are depended to agricultural sector which depend on rainfall for crop production. According to Butt *et al.* (2005), national agricultural production earnings in Mali will likely decrease from US\$ 417 million in 1996 to US\$ 256 million by 2030 because of climate change. In the Sahelian zone of West Africa, the rural societies whose livelihoods strongly rely on natural resources are already facing many challenges due to their harsh environment (desertification, recurrent droughts and sometimes floods) and poor socio-economic conditions (Mortimore and Adams, 2001).

This study aimed at assessing Cinzana farmers in the Sahel zone in Mali their perception on climate change variability and their adaptation practices to overcome or reduce the negative impact of climate change on their farming system as well as their livelihoods.

4.2 Material and Methods

4.2.1. Overview of the Study Area

AND SANE

This research was conducted in the Cinzana commune in the circle of Segou in the Segou Region of south-central Mali (Figure 4.1).

LEADY



Figure 4. 1: The Location of the Study Area

Cinzana is located around latitude 13'15 N, longitude 5'58 W with an altitude of 289 m. The characteristic feature is the alternation of long dry season from November to May and raining season from June to October. The average annual rainfall of the last ten years is 650-750 mm. This amount which is unevenly distributed over the period is insufficient for the needs of crops which largely explains the successive years of crop failure. The average temperature ranged between 28-33 °C with 39-40 °C as the highest and lower threshold of 8 °C to 12 °C. Cinzana rural town has a population of 37,572 composing of 18,727 (52%) males and 18,844 (48

%) females. The main economic activity of the commune is agriculture and the main crops are millet, sorghum and maize. The produce is primarily intended for home consumption because of poor rainfall, low soil fertility, and high cost of inputs which makes the production expensive. However, some farmers using improved technologies developed by the Institut D'Economie Rurale du Mali are able to generate surpluses that are traded on the weekly market at Cinzana. Farmers grow secondary crops like fonio, groundnuts, bambara groundnut, cowpea and sesame.

A series of on-farm surveys were conducted at Cinzana in 2013. One hundred and nineteen farmers' household including women and youth were randomly selected in the villages and were interviewed using structured questionnaire. The following formula was used to select the number of farmers:

 $n = t^2 x N/t^2 + (2e)^2 (N-1)...$ Equation 4.1

t: Confidence level (value of the confidence level of 95% which is 1.96) N= Size of the

population (number of households in this case) n: minimum sample size to obtain significant

results for an event and obtain significant result for an event and a given level of risk e= level of

precision (state to 10% in this study) Then the formula will be:

 $1.96^{2}N$

..... Equation 4.2

n =

 $1.96^2+(2e)^2(N-1)$ The structured questionnaire collected information on farming systems, perception on climate change and concerns for climate change. The questionnaire was designed in English and French but interviews were conducted in the local language, *Bambara*. The questionnaire was designed under the following 6 sections (see Appendix):

Section A describes the general information related to the farmers which included the socioeconomic characteristics of the households and farming activities. Data collected were age groups, gender, educational level, secondary activity practices, household size, main crop, land holding, average duration of continuous crop farming, fallow use and fallow age.

Section B concerned farm production, which focused on food coverage, food self-sufficiency, crop yield and yield patterns for the last ten years.

Section C dealt with farmers' perceptions on climate change. The data collected were on climate change impact, the different measures taken by farmers against the climate change impact, access to meteorological data and meteorological data type, and the identification of traditional methods for predicting weather.

Section E was on vulnerability and climatic risk management. Data recorded were on mitigation and adaptation strategies, food security and main climate change adaptation strategies.

Interviews were conducted in November 2013. The questionnaire was pretested for its suitability before interviews were conducted. The pretesting include behaviour coding of

interviewer/respondent interactions, interviewer debriefings, respondent debriefings and the analysis of item nonresponse rates and response distributions. Precipitation and temperature data were sourced from the Cinzana Meteorological station over the period from 1961 to 2014. The data gave an overview of the trends of precipitation and temperature and the vulnerability of the region to drought.

4.2.2. Data Analyses

Data from the questionnaire were coded and then analysed using the Statistical Package for Social Sciences (SPSS version 20) and MS Excel software. Descriptive statistical tool such as means, standard deviations, frequencies and percentages were used to summarize and categorize the information gathered. Cross tabulation was used to compare group's means. Correlation analysis was used to determine whether household income rather than education level significantly influenced adaptation strategies to climate variability and change. For the climatic variables

analysis, the annual rainfall average and index of Nicholson (RI) which is a reduced central variable calculated on annual precipitation to highlight fluctuations in rainfall patterns were calculated. It is the ratio of the difference in inter-annual to average standard deviation of annual precipitation amounts. Nicholson indices play a very important role in determining precipitation fluctuations (wet, normal and dry period). Nicholson *et al.* (1988) cited by Paturel *et al.* (1997) have defined an index which is calculated for each year as

 $RI = \int_{\sigma}^{Xi - \bar{X}} Equation 4.3$

Where:

RI: Rainfall Index

Xi: Rainfall in year in millimeters

X: Average height of average rainfall over the period of study in millimeters σ = standard deviation of the rainfall over the study period.

The Nicholson index determines a reduced central variable (Lamb, 1982) as quoted by (Servat et

al., 1998). Inter-annual average of a series is the index zero (0) by the method of Nicholson.

4.3. Results and Discussion

4.3.1 Socio-Economic Characteristics of Respondents

About 92.7 % of famers interviewed in the survey were male headed households and aged between 36 and 55 years. Data showed that about 13.4% of household heads attended secondary school, 31.8 % attended primary school and 48.7% were illiterates. The average household size was more than 20 people by 55.6 %. Also, 52.1 % of the household heads were polygamous. The most

experienced farmer in term of average duration of continuous farming was 50 years. Millet was the main cereal crop grown in the locality and is grown by 63 % of the farmers followed by sorghum 27 % and maize with 8 % (Figure 4.2).



Farmers found that millet contributed the most to satisfy their subsistence needs and income followed by sorghum and maize. The reasons for high cultivation of millet among farmers is attributed its high productivity, consumer preference and drought tolerance. About 46 % of farmers grew millet because of its high productivity, 16 % grew it because of its consumer preferred traits and 21 % cultivated millet because of its tolerance to drought (Figure 4.3).



Figure 4. 3: Cinzana Farmer's Reasons for Crop Choices

The population of Cinzana is dominated by relatively young people of less than 60 years old and male headed households. These male headed households are the active population in the locality. The main socioeconomic factors that impacted on adaptation decisions were level of education, gender and age of the household head, household size and years of farming experience. Effect of age on farmers' adaptation to technology contradicts existing in literature (Teklewold *et al.*, 2006; Glwadys, 2009). In the view of Teklewold *et al.* (2006) and Glwadys (2009) adaption is mostly influenced by location or technology and specific findings of age are interesting empirical results. This is because the decision to adopt a technology is not easily accepted by older farmers compared to younger farmers.

The average size of a household in this locality was very large (20 persons) compared to many households in other locations in Mali. In Cinzana, female-headed households were non-existence as females are supported by men for subsistence according to their culture. However, women can practice some activities to improve their main source of income. Household size influenced the decision to adapt. Therefore, household size is a proxy to labour availability and may influence adoption of the new technology positively as its availability reduce the labour constraints (Teklewold *et al.*, 2006), However, there is a possibility that households with many family members may distract the labour force to off-farm activities in attempts to earn income to ease the consumption pressure imposed by the large family size (Tizle, 2007).

The gender of household head influenced the decision to adopt the change. Several scholars in Africa have shown that women have access to critical resource (cash, land and labour), which often undermines their ability to carry out intensive labour agriculture innovation (Quisumbing *et al.*, 1995; De Groote and Coulibaly, 1998). According to Nhemachena and Hassan (2007), femaleheaded households are more likely to take up climate change adaptation methods. A number

of studies found that, the possible reason for this observation is that in most local smallholder farming communities in the region, men are more often based in towns and much of the agriculture work is done by women (Nhemachena and Hassan, 2007; Quisumbing *et al.*, 1995; De Groote and Coulibaly, 1998).

However, level of education and literacy rate provides information about development at regional and national levels and they reflect people's chances of alternative sources of income and livelihood. The literacy rate among farmers in Cinzana is substantially very low, only 13.4 % of people have attained secondary education. The farmers who attained secondary education level rely firstly on subsistence farming as their main source of income. Education level is often an assumption to increase in the understanding of adoption a new technology (Adesina and Forson 1995; Daberkow and McBride, 2003). This is because education is expected to increase the ability to receive, and understand information relevant to making innovative decision (Wozniak, 1984).

With an average rainfall of 650 mm in Cinzana, millet is largely cultivated, particularly on marginal lands. These areas are further characterized by variable and a short rainy season, irregular rainfall and a high evaporative demand (high radiation and temperatures). Lastly, the soils are acidic with low mineral fertility (especially low in phosphorous) and organic matter content. For all of these reasons, millet is considered to be the farmers preferred crop in Cinzana because of the low rainfall and adverse soil factors. Experience of farming probably increase the uptake of all adaptation option because of farmers' experiences that make them have better knowledge and information on change in climatic conditions as well as crop and livestock management practices (Hassan and Nhemachena, 2008).

4.3.2. Farmers' Perceptions of Climate Change and Variability and its Impacts on Their Farming System

All the respondent in the region believe that the climate is changing and is no longer as it was some years back. Each farmer gave their perception on each of the following factors: length of rainy season, soil fertility, crop yield, vegetation cover, temperature and drought. Figure 4.4 indicates that 94.1 % of the respondents perceived a decrease in the length of the rainy season.



Farmers' perceptions

Figure 4. 4: Farmers' Perceptions of Climate Change and Variability in Cinzana Community in 2013

The respondents perceived a decrease in soil fertility as represented by 68.9 % (Fig. 4.4). Most of the soils in the locality were very poor in fertility. About 27.7 % of the farmers interviewed indicated that the loss of the vegetation cover was due to climate change. Surprisingly, only 15.9 % of the respondents' had perception of climate change indicated a decreased crop yield; 10.9 % of respondents observed temperature while 8.4 % found drought change over the years.

The farmers were using local knowledge to predict weather. More than 50 % of the farmers traditionally referred to tree flowering in the region for weather prediction while 38% use animal migration. Other less important factors are direction of wind (4.2 %) and change in temperature (1.6 %) (Figure 4.5).



Figure 4. 5: Traditional Methods to Perceive Climate Change and Variability for Cinzana Commune Farmers

Furthermore, the respondents were asked about their perception on climate change and climate variability and their effect on the environment. Most of the farmers (87 %) indicated that agricultural areas have expanded followed by more vegetable cultivation and lowland cultivation with 69% (Fig. 4.6) They also found that forest areas (trees) have reduced by (65 %) and the savannah (the grazing area) has also decreased (61 %). Figure 4.4 indicates that 94.1 % of the respondents perceived a decrease in the length of the rainy season.

LEADW

SAP J W J SANE



Figure 4. 6: Effects of climate change as perceived by farmers in Cinzana

Smallholder farmers in Cinzana are aware of climate variability and change, principally through their experiences by appreciating the length of the rainy season, soil fertility, crop yield, vegetation cover, temperature and drought. Some changes were observed on farmers' knowledge as factors used to predict weather. These included change in flowering period for some trees in the region, animal migration especially some birds, wind intensity and direction and high temperatures. This perception agrees with the general opinion that climate is changing (IPCC, 2007; McCarthy *et al.*, 2008; Orindi and Eriksen, 2005; Lobel *et al.*, 2013; Alexander, 2013). Furthermore, studies on farming perception in semi-arid environment of Africa (Nyanga *et al.*, 2011; Osbahr *et al.*, 2011) and farmer perception on climate variability and change in semi-arid Zimbabwe e (Moyo *et al.*, 2012) showed similar findings.

This study indicated that farmers at Cinzana believe that climate is changing, and it was perceived to affect agricultural productivity negatively. There is, however, divergence of perceptions amongst the farmers, as indicated by the results of what is causing the change in agricultural productivity. Famers reported a number of changes within their location; however, contradictions were apparent among them about the exact nature and the intensity of the changes. This may explain the different perceptions farmers have on effect climate variability and climate change have on agricultural production and productivity. Therefore, climate change and variability is often given as the main reason for crop failure and food shortage (Traore *et al.*, 2014; Mishra *et al.*, 2008; Sultan *et al.*, 2005).

4.3.3. Climatological Evidence Compared to Farmers' Perceptions

a) Rainfall and Temperature Variations

There were two periods regarding the rain season in the study area (Fig. 4.7). The dry period from 2000 to 2007 showing rainfall index supported by negative value index of this period, while the wet period from 2008 to 2012 showed a positive rainfall index value. The analysis showed the beginning of the dry period was supported by negative rainfall index for the 2013 year; which is in agreement with farmers' perceptions in this study. Overall there was a decrease of precipitation distribution during the 14 years of observations since 8 out of the 14 years indicated a negative index against 6/14 for the positive index. Rainfall is the most important variable that farmers perceived since it had a clear signal in the record of climate.

The analysis of temperature means during the same period showed a clear increase in temperature from 2000 to 2007 while from 2008 to 2012 it decreased (Figure. 4.8.). This was not in agreement with farmers in the study area. Few of the respondents perceived that temperature had changed, and could be explained by the fact that in the Sahelian agro ecological zone, farmers focus more on rainfall distribution than temperature.



Figure 4. 5: Annual Rainfall Indexes (mm) from 2000 to 2013 in Cinzana (Data source: Cinzana Meteorological Station)



Figure 4. 6: Annual Temperature Means from 2000 to 2013 (Data source: Cinzana Meteorological Station).

The decrease of annual rainfall indicated by farmers was confirmed through an analyses of observed historical data in southern Mali (Mouhamed *et al.*, 2013; Traore *et al.*, 2014), showing an increase in the inter-annual rainfall variability. Furthermore, a study by Cooper *et al.* (2008) supports the same results for the Sub-Saharan Africa. In several countries, significant results between observed and farmers' perception of climate change and variability were found (Apata *et*

al., 2009; Deressa *et al.*, 2009). However, most farmers perceived a continuous decrease of annual rainfall which was not confirmed by the analysis of climate data. No clear change in total precipitation was observed over the last five decades in the Sudanian zone of Mali (Traore *et al.*, 2013). Changes in annual rainfall that farmers think they have been experiencing could be caused by the increase and decrease in the year to year variability of seasonal rainfall (Balme *et al.*, 2006). However, farmers may be reporting overall decline which could be attributed to temperature increase. For temperature, some farmers' perception of a substantial increase matches the empirical observations for the region very well (Hulme *et al.*, 2001). Farmers linked the temperature increase to a change in the number of drought days in the growing season, as these are perceived to be hot. Osbahr *et al.* (2011) showed that temperature increases led to increased evapotranspiration rates by linking to the faster depletion of soil water. Researchers found the high levels of depletion of soil water resulted from high rates of evapotranspiration normally leading to crop wilting, and causing crop failure, which the farmer may be attributing to a decline in rainfall.

4.3.4. Coping and Adaptation Mechanisms to Climate Change and Variability

To cope with the negative impacts of the perceived change in climate patterns on their farming systems, the majority of the farmers in the study areas have adopted some adaptation strategies and coping practices. Among these practices, the most commonly adopted strategies were the use of adapted crop varieties such as drought or pest tolerant crop varieties, early sowing and the use of organic mature as fertiliser. Other practices were the systematic use of chemical fertilizers, change in cropping system, more labour investment and rainwater harvesting. Table 4.1 indicates that about 75 % of the respondents interviewed used adapted crop varieties that are the most tolerant to drought for the zone while 73.1 % use early crop sowing for the different crop varieties to optimise the available moisture in the soil. Concerning organic manure use as an adaptation

practice, 32.8 % of the respondents used this to reduce the impact of climate change and climate variability. Other strategies mentioned by respondents were the use of chemical fertilizer (28.6 %), cropping system (13.4 %) and rainwater harvesting (1.7 %).

Adopted practices	Farmer's response (%)
Adaptation practices	Yes
Variety adapted	75.6
Early sowing	73.1
Organic manure use	32.8
Mineral fertilizer use	28.6
Cropping system changes	13.4
Rainwater harvesting	1.7
Coping practices	Yes
Cropping system	65.5
Organic manure	63.5
Use of adapted crops	41.2
Rainwater harvesting	3.4
Livestock selling	0.8
Buying supplies	0.8

 Table 4. 1 Coping Strategies to Climate Change and Variability of Farmers in Cinzana

 Commune in 2013

Results of adaptation strategies to climatic stressors undertaken by households are given in Figure

4.9. The major adaptation mechanisms include selling livestock especially small ruminants and

poultry (42 %), utilization of improved crop seeds and seeking other activities outside agriculture

(11.7%), getting credit (10%) and use of manure (8.4%).



Adaptation Strategies

Figure 4. 7: Famers' Adaptation Strategies to Climate Change and Variability Stressor in Cinzana community in 2013

Descriptive statistics results have shown that farmers in the study area have adapted to the effect of climate change through a number of mechanisms. Households have multiple strategies related to agriculture and engaged in off-farm activities to complement their household incomes and food in case of adverse conditions in agriculture. The majority of respondents reported that they have used more than one type of adaptation and coping strategies. This decision implies that a single strategy is inadequate in adapting to the impact of climate change as a combination of several strategies is likely to be more effective than a single strategy. The major coping mechanisms to climate change undertaken by households were adapted crop seed varieties; which is different from the studies on coping mechanisms by Fana and Asnake (2012) in Ethiopia, Quay (2008), Kyekyeku (2012) in Ghana, and Bardege *et al.* (2013) in where? Who obtained various diversified coping strategies in response to climate change, which most of them related to the coping mechanism found in the study.

The findings reveal that households in Cinzana commune had several adaptation strategies in response to the effect of climate change and variability. The major strategies included use of adapted varieties, use of organic matter and selling of animals. This implies that climate variability and change is not a new phenomenon to the Cinzana commune. Those strategies are not strange but have a close relationship to those carried out elsewhere. However, a study done in 11 African countries by Nhemachena and Rashid (2008) found that diversified adaptation mechanisms to climate change such as diversifying production, using different improved varieties, changing planting dates, increased irrigation, use of insurance, water conservation, prayers, soil conservation were used., The adaptation strategies adopted in different regions or countries depend on their level of economic development, technology, financial capacity, institutional support and traditions. Therefore, each region tends to have most similar adaptation strategies adopted.

4.3.5 Correlation Estimation of Some Factors Influencing the Farmers' Adaptation Strategies to Climate Change Effects

To diagnose correlation, the contingency coefficient test was applied and omits independent variables that are strongly correlated and dependent to each other (Table 4.2). The study observed multi-correlation between age and farming experience, sex and farming experience, household size and number of farms, secondary activities and farming experience, sex and education, education and household size, sex and number of farms.

Table 4. 2 Correlation of factors studied on adaptation mechanisms in Cinzana commun								
	Ag	Gen	Edu	Hldsize	Adpstra	Farsize	Farexp	Income
Age	1		5	1		2		
Gender	-0.019	1						
Education	-0.127	-0.177*	1	10				
HldSiz	0.169*	-0.122	.164*	1				1
Adstra	0.057	0.099	-0.005	0.073	1-4	1		
Landsize	0.132	-0.184*	0.110	<mark>0</mark> .355**	0.013	1	F	3
Farexp	0.208**	-0.228*	* 0.089	0.255**	-0.038	0.262**	1	
Income	0.004	-0.161	0.068	0.112	0.170	-0.057	-0.050	1

Table 4. 2 Correlation of factors studied on adaptation mechanisms in Cinzana commune

* Correlation is significant at 0.05 level

** Corelation is significant at 0.01 level

NB: Ag= age, Gen: gender, Edu= education, Hldize= household size, Adpstra= adaptation strategies, Farsize= farm size, Farexp: farming experience

The results of correlation (Table 4.2.) indicated a positive correlation (0.01 %) between age and farming experience. This means that farming experience significantly increase with increase in the number of years of the respondent in farming. Experienced farmers see the need to adapt to climate variability and change effects. Furthermore, these older farmers may be more interested in the following traditional methods familiar to them rather than adopting modern farming techniques. Acquah (2011) and Quayum *et al.* (2012) found similar results but negative significance. Their findings may be because young adults have more motivation to act on perceived changes in order

to cope with it. The ability to adapt depends on individual's motivation to act (Bandura, 1997). Therefore, young adults are able and have energy to get jobs outside agriculture and can also diversify agricultural production which can help them get more income to adopt other adaptation. The influence of farmers' gender was negatively correlated to the educational level, the number of farms and highly correlated to the farming experience negatively. This influence of gender is because the males were more likely to adapt than their female counterparts since in many African traditions the females have less access to land and other socio-economic resources constraining the adaptive capacity of females to adapt to climate change and variability. It was reported by International Food Policy Research Institute (IFPRI, 2001) and Meinzen-Dich *et al.* (2010) on gender studies that unequal distribution of assets between males and females in rural households was in favour of males. The results from this study agrees with those results by Tenge and Hella (2004) and Nabikolo *et al.* (2012).

The study also showed a high correlation between the household size and the number of farms and farming experiences. Households with high number of people have more land and more experiences in farming. They are more likely to adapt than their counterparts with small land because those farmers may open up large gardens and plant more crops. People who own more land may rent out some of their land in order to generate more income for the household and be able to adapt more appropriately than those who have less land (Nabikolo *et al.* (2012). Advancing Capacity to Support Climate Change Adaptation (ACCCA, 2010) reported that large farm sizes positively influences adaptation strategies such as the use of growing trees and improved varieties of crops and livestock.

4.4 Conclusion

In Conclusion, the agricultural sector of Mali is largely led by small scale rural farmers and it is seriously affected by rainfall fluctuation. Millet was the mean cereal crop growing in Cinzana

commune because of its high productivity, their own consumption and because millet is tolerant to drought. Cinzana farmers feel and perceive changes in the climate patterns. They do take measures to deal with the changing climate though they have limited knowledge of climate change as a scientific concept Farmers used various measures to deal with the changing climate and the most preferred measure is the use of improved crop varieties was the main adaptation. Gender, size of land and age of household head were the significant determinants of adaptation strategies.



CHAPTER 5: ASSESSMENT OF CHANGING TRENDS OF DAILY PRECIPITATION AND TEMPERATURE EXTREMES IN SOTUBAAND CINZANA IN MALI

5.1. Introduction

The anthropogony's problem of climate change has been identified by researchers and is a widely acknowledged issue in the political sphere during the last decades. The different emissions of greenhouse gases linked mainly with industrial production increase the global mean temperature via the greenhouse effect, affect the whole Earth system (Lavinia, 2011). These changes imply

risks for human lives as well as natural ecosystems. In different parts of the world, climates are characterized by their variability which affects many sectors. Since the 1950s this variability has increased and particularly so during the latest decade, principally because of the increased concentration of anthropogenic greenhouse gases in the atmosphere as argued by some scientists (Trenberth *et al.*, 2007a; Hegerl *et al.*, 2007; Stott *et al.*, 2011; Solomon *et al.*, 2009, 2010). It is argued in this context that surface temperature has increased by 0.7 °C over the last century with a significant warming in several regions and with the land areas warming faster than the oceans (Trenberth *et al.*, 2007a).

IPCC has provided since the early 1990's proof of accelerated global warming and climate change. The global average temperature in the last 100 to 150 years has increased by 0.76 °C (0.57 °C to 0.95 °C) (IPCC, 2007a). However, global rainfall trends are complex because of wide regional differences, gaps in spatial coverage and temporal shortfalls in the data. In addition, Traore (2014), showed evidence for increases in the frequency of both heavy rains and severe droughts in many regions of the world. It has been shown by all 21 General Circulation Models (GCMs) used by IPCC to project that in sub-Saharan Africa a temperature increase in the order of 3.3 °C and this will happen by the end of the 21^{st} century. Regarding projected change in the amounts of rainfall in sub-Sharan Africa, the uncertainty is considerably greater and in several models do not agree on whether change in rainfall will be either negative or positive (Cooper *et al.*, 2008). Agricultural production in Africa is uncertainty associated with between and within season rainfall variability and remains a fundamental constraint to several investors who often overestimate the negative impacts of climate induced uncertainty. Climate change is likely to make matters worse with increases in rainfall variability being projected. A combination of internal and external forces in West Africa makes the climate of the region one of the most erratic in the world (Zeng, 2003). Several scholars have characterized the rainy season in West Africa; most of them described the start and end of rainy season (Dodd and Jolliffe, 2001; Omotosho *et al.*, 2000; Diop, 1996) However, other studies were based on decadal, total or monthly annual rainfall analysis (Nicholson, 1980; Sivakumar and Salaam, 1994; Ati *et al.*, 2002). A good pattern of perception of the seasonal variability is critical because of the wide unstable rainy onset season and high temporal and spatial variability and by alternation between wet and dry seasons (Servat *et al.*, 1998). According to Traoré *et al.* (2007) present knowledge on the regional climate in Sudano-Sahelian area West Africa revealed that rainfall remains unpredictable. This unpredictability of rainfall is the most constraint for farmers who have to plan at the beginning of the cropping season (Piéri. 1989). The first rain is not always followed by the full start of the monsoon (Sultan and Janicot, 2003), afterward dry spell can occur, for instance during the early stages of the crop growth so that germinated plants may die off or seeds may not germinate properly. Otherwise, if sowing is delayed, the land may be too wet to till (Traore, 2014).

According to a publication of CEDEAO-Club Sahel/OCDE/CILSS on Climate and Climate Change (2008), in the West African Sahel, an increasing trend was observed in both maximum and minimum temperatures for all the three ecological zones (Sudanian, Sahelian and SahelSaharan) with temperature increasing at a faster rate. Therefore, climate change is perceived through extreme events which tend to modify the magnitude of the predicted climate, while this can be supported by the number of flood events having increased on average from less than 2 per year before 1990 to more than 8 or 12 on average per year during the 2000s (Sarr, 2011). In addition, the annual total rainfall evolution has been characterized by succession of wet years from 1950 to 1969, followed by a period with the persistence of dry years from 1970 to 1993. The persistence of dry years from 1970 to 1993 (Ali, 2011; L'Hote *et al.*, 2002) was found in the southward movement of isohyets by about 200 km (Diouf *et al.*, 2000). Ali (2011) showed that in the region after 1993, another type of variability characterised by an alternation between very dry and wet years to have begun. Furthermore, the same author indicates that while there is tendency of persisting drought in the Western Sahel, the east is experiencing gradual return to wetter conditions.

The occurrence of extremes is usually the result of different factors at different time scales. A large amount of the available scientific literature on climate extremes is based on the use of the so-called 'extreme indices', which can either be based on the probability of occurrence of given quantities or on threshold exceedance (Ly *et al.*, 2013). The same authors argued that several publications related to those extremes emerged in some specific countries in particular in the developing world where most of the time, data availability is always distributed by the meteorological service. In Africa particularly, very few studies document that, such changing climate is because of several difficulties in accessing daily meteorological data (Lamptey, 2009).

Therefore, the World Meteorological Organization (WOM), through the Climate Variability and Predictability (WMO/CLIVAR) Expert Team on Climate Change Detection, Monitoring Indices (ETCCDMI), coordinated a series of workshops in different regions of the world to help scientists and national experts discuss and assure quality control of climate data. Some regional analyses were thus undertaken for the understanding of climate extremes and trends (Easterling *et al.*, 2003; Vincent *et al.*, 2005; Aguilar *et al.*, 2005). However, recent studies in Africa, when analysing daily climate in terms of trends and extreme indices revealed some significant increases and decreases in annual precipitation; increases in longest wet spells, increases in high daily precipitation amounts and average rainfall intensity (New *et al.*, 2006; Alexander *et al.*, 2006). Minimum temperature was found to have increased faster than maximum temperature, thus contributing to narrow the diurnal temperature range (Easterling, 1997; Caesar *et al.*, 2006). Thus, many indicators in the extreme indices can be analysed to further investigate the perception of changes in climate in widespread area in the West African Sahel where a few studies have documented the changes that occurred in those extremes (Ly *et al.*, 2013). This is why, it's important to set trends in climate variables as this will indicate the nature of climate-related adaptation and decision strategies employed by various stakeholders to improve their livelihoods in an agricultural-based economy (Makuvaro, 2014).

Annual rainfall in Mali is highly variable, ranging from less than 200 mm to 1300 mm. Climate change threatens to increase air temperatures and evapotranspiration, increase the risk of intense rainstorms, and increase the risk of heat waves associated with drought (Baptista *et al.*, 2013). According to Diarra *et al.* (2007); Butt *et al.* (2005); and Jalloh *et al.* (2013), the literature on climate change in Mali makes a variety of predictions about the specific changes that will occur. In Mali, the impact of climate change on precipitation is varying on which global and regional combination climate models are used (Baptista *et al.*, 2013).

Climate change is expected to increase vulnerability in all agro-ecological zones of Mali through rising temperature and more erratic rainfall, which will have drastic consequences on food security and economic growths (Butt *et al.*, 2006; Dell *et al.*, 2012). Climate change projections from the Hadley Centre Coupled Climate Model (HADCM) and Canadian Global Couple Climate (CGCM), suggest that by the year 2030, Malian average temperature may increase by 1-2.75 °C and 2-4 °C before 2060 (Butt *et al.*, 2005; 2006; Konate, 2010). The impact of temperature on agriculture is of major significance since 1 °C rise is linked to a 2.7 % reduction in growth in agricultural outputs (Dell *et al.*, 2012). Rainfall characteristics are particularly affected by these changes in rainfall: the

onset and ending date of the rainy season, the distribution and intensity of rainfall events (Traore *et al*, 2000; Konaté, 2010). The impact of these changes on rainfall characteristic is 20–60 % yield losses for agricultural productions by 2025 (Butt *et al.*, 2005). Here too, the biophysical basis is fairly clear-cut that rainfall characteristics affect water balance, more specifically water availability and evapotranspiration, plus plant physiology directly (Dell *et al.*, 2012). The objective of this study was to assess the changes in trends of daily temperature and precipitation extremes in Sotuba and Cinzana in Mali during the period 1961 to 2014.

5.2. Materials and Methods

5.2.1. Study Area

The study was conducted in Mali in two locations, Sotuba in Sudanian agro-ecological zone and Cinzana in Sahel agro-ecological zone. Annual maximum and minimum temperature means vary between 27- 44 °C and 11- 30 °C, respectively in Sotuba whilst in Cinzana they vary between 26-47 °C for maximum mean and 13-14 °C for minimum mean. The rainfall season is unimodal, with the onset of the rains occurring in the main agricultural areas from May to July and ending in

September–October.

With an area of 245 km², Sotuba is located in the Sudan zone and has a tropical dry and wet climate with average highs for over 30 °C. Sotuba is very hot on average all years round with the warmest month being March, April and May. November to February is the coolest months. During October to April, rainfall is severe with little rain falling and virtually none between December and February. The rainy season occur with the peak in July, August and September.

Ségou is a town and an urban commune in south-central Mali that lies 235 kilometres northeast of Sotuba on the River Niger. The region of Ségou is characterised by a semi-arid climate and drained by two important waterways: Bani and Niger River. It has two seasons: a dry and rainy season.

The dry season includes a cold period and a period of heat. The rainy season starts in June and lasts about four months until September and yearly rainfall average is about 600 mm. The harmattan is dominant in the dry season and it blows from north to south. The monsoon blowing from south-west is more frequent during the rainy season. The list of the meteorological station used in this study is given in Table 5.1.

Table 5. 1: Meteorological Sations Used in the Study						
Station Name	Longitude	Latitude	Elevation (m)			
Bamako	-7.95	12.53	289			
Ségou	-6.15	13.40	320			

5.2.2 Analysis of Extreme Events

a) Descriptive Indices of Extremes

To improve a constant perspective on observed change climate and weather extremes, ETCCDI (Expert Team on Climate Change Detection and Indices) has defined a core set of descriptive indices of extreme. The indices describe special characteristics of extremes including amplitude, frequency and persistence. The core set includes 27 extreme indices for precipitation and temperature. R-based software (RClimDex) which is user friendly was downloaded on line from http://www.r-project.org.

The most important key of the indices concept involves calculation of the number of day in year exceeding specific thresholds. Example of such "day-count" indices are the number of days with minimum temperatures below the long-term 10th percentile in the 1961-1990 base periods or number of days with rainfall amounts greater than 20 mm. Several ETCCDI indices are based on percentiles with thresholds set to assess moderate extremes that typically occur a few times every year rather than great impact, once comes in a decade weather events. The percentile thresholds

for precipitation is calculated from the sample of all wet days in the base period and for temperature are calculated from five-day windows encountered on each calendar day to account for the mean annual cycle. The reason for choosing mostly percentile thresholds rather than fixed thresholds is that the number of days exceeding percentile thresholds is more evenly distributed in space and is meaningful in every region. Such indices allow straightforward monitoring of trends in the intensity or frequency of events.

b) Data, Homogeneity Testing and Quality Control

Daily minimum and maximum temperatures and daily precipitation for the two stations were provided by the Centre of Aghrymet of Niamey for Bamako and Ségou Meteorological stations for 1961-2014 periods and were chosen to investigate extreme precipitation and temperature and to optimise the number of significant trends and data coverage throughout the study area. Data quality and homogeneity testing as well as the detection and adjustment of inhomogeneous time series were carried out using RClimdex 3.2.0 package (Zhang and Yang, 2004) which is freely downloadable from the ETCCDMI website at: http://ccma.seos.uvic.ca/ETCCDMI/index.shtml. As a first step, temperature and precipitation time series with more than 20 % missing values within the analysis period 1961–2014 were excluded. Outlier in the time series has been identified and temporal consistency was tested according to Aguilar *et al.* (2003). In the data quality control subroutine, unrealistic data was found such as:

- days with negative or greater than 500 mm rainfall amount,
- days with minimum temperature equal to or greater than maximum temperature and
- minimum temperatures greater than maximum temperature values were detected and corrected.

81

Observational climate data can be influenced by different climatic effect, such as the relocation of weather station, land use changes, adjustments in instruments and observational problems. These effects result in homogeneity problems causing a shift in the mean of a time series, which could result in first order autoregressive errors. Also, in the quality control process, each time series was tested to ensure homogeneity in the whole dataset. The erroneous observations (i.e. the unrealistic data) were replaced with a code (-99.9) which is recognized as missing data by the RClimDex software package to avoid errors in the computation. After ensuring these measures, the stations data was assumed 100 % consistent to further execute the study hours (Peterson *et al.*, 2002; Aguilar *et al.*, 2003).

c) Climate Extreme Indices and Analytical Methods

The 27 ETCCDI indices agreed by the international community aim to monitor changes in "moderate" extremes and to enhance studies on climate extremes using indices that are statistically strong and cover a large range of climates and have a great signal-to-noise ratio (Min *et al.*, 2011). The indices are calculated from daily precipitation data (Peterson *et al.*, 2002; Karl and Easterling, 1999). From the core extreme indices, 4 extreme precipitation indices and 5 extreme temperature indices were selected for the present study (Table 5.2). All trends for indices chosen were calculated annually using the Software RClimDex 3.2.0.

Percentile indices were calculated using the standard reference period 1961 to 2014 to make results easily comparable with other studies using the same reference period. During the trend estimation of the percentiles – based indices, RClimDex 3.2.0 uses the boot- strapping approach to avoid possible bias within the reference period associated with the existing inhomogeneity (Zhang and Yong, 2004). During the calculation process, particular data requirement must be met in order to calculate indices value using RClimDex. An annual value is considered as incomplete if more than 15 days are missing in a year. A month will not be calculated when \geq 3 days are missing, a year will only be calculated when all months are present. A percentile bases index will only be calculated if there is 70 % data present at least within the reference period. Additional ETCCDIrecommended standard criteria are applied as described in the RClimDex user manual online at: http:// etccdi.pacificclimate.org/RClimDex/RClimDexUserManual.doc.

In the study, statistical analyses were done over the last 53 years since 1961 including the current climatological base period of 1981–2010 as defined in the source code of the RClimDex. Plots and tables of trends for the period 1961–2014 for the selected indices (Table 5.2) were generated using the RClimDex, a developing script in R open source software (Core team, 2012). To determine the significance of the trends, Student *t*-test was performed and the resulting *P* value, which now serves as a criterion to define the class boundary, was used to analyse the null hypothesis that the trend is equal to 0. The trend for each index was considered significant when found ≤ 0.05 .

	Contraction of Contraction	and the second sec	
Doinfall		Annual total	Annual total PRCP in wet days (RR \geq
Kaiman	PRCPTOT	rainfall	1 mm) Mm
Rainf <mark>all</mark>	RX5day	Maximum 5-day precipitation	Maximum 5-day precipitation Mm
Rainfall CW	CWD	Consecutive wet	Maximum number of consecutive wet days
	CWD	days	Days
Rainfall	R99p	Extremely wet	Annual total PRCP when RR >99th Mm
		days	percentile
Tn 7	Tn10p	Cool night	Percentage of days with TN <10th ₀₆
		frequency	percentile of 1961–2014
Tn	Tn90p	Warm night	Percentage of days with TN >90th ₀₄
		frequency	percentile of 1961–2014
Tx	Tx10p	Cool day	Percentage of days with TX <10th ₀₄
		frequency	percentile of 1961–2014

Descriptive name

Index

Element

 Table 5. 2 Summary of rainfall and temperature indices used in the study

Units

Definition

$\mathbf{T}_{\mathbf{v}}$	TyQOn	Warm	day	Percentage of days with TX >90th _{0/2}
	тхлор	frequency		percentile of 1961–2014
				Annual count of days with at least 6
Tx	WSDI	Warm spells		consecutive days with TX >90th Days
			percentile of 1961–2014	

Tn and Tx are daily maximum and minimum temperatures, respectively

5.3. Results and Discussion

5.3.1. Temperature and Rainfall Indices

a) Temperature Indices

Temperature indices showed a general warming trend throughout the region during the period from 1960 to 2014 (Figure 5.1 and 5.2). The stations with negative trend (significant at the 0.05 level) and positive trends (significant at 0.05 levels) in some of temperature indices extremes are shown in Table 5.3. The regional annual series for temperature indices during 1960-2014 are shown in Figure 5.1 and 5.2. Over the 1960-2014 period, cold day (TX10p) and cold night (TN10p) showed decrease in warming with the trend of -0.028 and -0.029 respectively in Sotuba station but was not significant (Figure 5.1 a and Figure 5.1 c). However, warm day (TX90p), warm night (TN90p) showed positive increase in warming at the rate of 0.081 and 0.029 respectively (Figure 5.2 a, figure 5.2 c) and warm spells (WSDI) showed positive significant trends over the periods 1960-2014 cold day (TX10p) and cold night (TN10p) (Figure 5.1 b and d), warm day (TX90p), warm night (TN90p) and warm spells (WSDI) with 0.331, 0.631, 0.392, 0.292 and 0.78 respectively (Figure 5.2 b).

Indeed, in Sotuba the result of the frequency of cold nights and cold days appears to be negative and slope meaning that the nights have become warmer. Warm nights, warm days and warm spells have also become more frequent. Figure 5.1 and 5.2 illustrate that, through the analyses of some temperature indices during the study period the frequent cold nights and of cold days' decrease while that of warm days and warm nights and also of warm spells increased. Whereas, in the Ségou region, the results showed a general positive trends in cold nights, cold days, warm nights and warm days and also warm spells (Figure 5.1 and 5.2). While the frequency of cold ones decreased the warm ones increased. These results were not the case for only this study or for Sahelian countries, but also for those in coastal countries because of temperature regime caused by the influence of ocean. Thus, the increase in temperature was much higher than the global warming trends.

Generally, positive trend in cold and warm extremes, found in this study are consistent with increasing trends in both warm days and warm night temperatures and decreasing trends in both cold days and cold night as well as in frequency of both extremes established for West Africa by New *et al.* (2006) who showed the number of cold nights and days have decreased and the number of warm nights and days have increased between 1961 and 2000. Several of these trends are statistically significant at 90 % level. Kruger and Sekele (2013) also found that in South Africa, warm extremes increased and cool extremes decreased for all the weather stations during 19622009 periods. Therefore, the warming trend observed means a higher demand on domestic energy consumption for cooling, higher evaporation rates from water bodies and irrigated crops, and a lower performance from livestock and agricultural crops. Furthermore, old persons are particularly vulnerable in such increases in temperature and may increase the level of mortality.

indices extremes in Soluba and Cinzana for 1900-2014 periods							
	Sotuba (1960-2014)			Cinzana (1960-2014)			
	Mean		P-	Mean	D voluo		
Indices		Slope	value	Slop	pe r-value		
Cold day (%)	10.48	-0.028	0.487	13.48 0.33	81 0**		

Table 5. 3 Trend per percentage and day with positive and negative slope of temperature indices extremes in Sotuba and Cinzana for 1960-2014 periods

Warm day (%)	9.15	0.081	0.084	6.95	0.392	0**
Cold night (%)	12.35	-0.029	0.685	17.71	0.631	0**
Warm night (%)	8.36	0.029	0.55	7.41	0.292	0**
Warm spell (day)	4.15	0.049	0.049*	3.67	0.278	0**

*Statistically significant trend at p-value = 0.05; **highly statistically significant trend at p-value = 0.05

In addition, a study done by Ly *et al* (2013) in the West Africa Sahel showed a general warming trend throughout the period from 1960 to 2010. Through some selected locations in the West Sahel, Ly *et al.* (2013) found steady warming during the study period. Indeed, while the cold night and cold days' frequency decreased that of warm nights, warm days and warm spells increased. This temperature increase can result in reduced fodder yield, an increase in evapotranspiration, a possibility of migration and conflict between livestock and crop farmers as well as in the economical, physical and psychological cost (Thornton *et al.*, 2009; Sirohi and Michaelowa, 2007). Further, this increase in temperature also has an impact on human health (McMichael *et al.*, 2012). However, some authors (Ly *et al.*, 2013) found out that in the last two decades, the frequency of cool nights decreased about twofold going from 10.1 % to 2.9 %, 10.3 % to 7.4 %,

10.1 % to 4.2 %, and 10.2 % to 04.8 %, for Nouakchott, Bamako, Ouagadougou and Niamey respectively. The result in Ségou is consistent with this later finding in that station for the most general significant warming trends. This was not observed in Sotuba where only warm spells were significant probably because of the reduced ocean-land temperature gradient that is characteristic of the monsoon regions as shown by Trenberth *et al.* (2007b) and Giannini *et al.* (2008).

NO BADH

ASAP J W J SANE



NO

Figure 5. 1 Evolution of the percentage of cold nights (Tn10p) and that of cold days (Tx10p) in Sotuba (a and c) and Cinzana (b and d) from 1960 to 2014 period

AP J W J SANE



Figure 5. 2 Evolution of the percentage of warm nights (Tn90p) and that of warm days (Tx90p) and warm spells (wsid) in Sotuba (a and c) Ségou (b and d) from 1960 to 2014 period

NO

WJSANE



Figure 5.3: Evolution of the percentage of warm spells (wsid) in Sotuba (a) Cinzana (b) from 1960 to 2014 period

b) Rainfall Indices Extremes

Table 5.4 shows the regional trend in some extreme rainfall indices and the stations with negative (significant at the 0.05 level), no trend and positive trends (significant at the 0.05 level). The regional annual series for rainfall indices during 1960-2014 are shown in Figure 5.4. However, the results of precipitation extremes for Ségou showed positive significant decrease of 0.009 with negative slope trend, while consecutive wet days (CWD) (Figure 3b) and cumulated rainfall extremely wet days (RX99p) showed a positive slope and significant decrease with a rate of 0.00.9 and 0.029 respectively (Figure 5.4a). Whereas maximum 5 day's precipitation (RX5 Day) showed positive insignificant slope increase with 0.057 (Figure 5.3a) and the total annual precipitation (PRCPTOT) showed positive insignificant decrease with 0.541 with negative slope trend (Figure 4c). While in Sotuba, all the precipitation extremes showed negative slope trend. The consecutive wet day (CWD) (Figure 5.3c) and RX5p (Figure 5.3a) and total annual precipitation (Figure 5.4d)

showed positive insignificant decrease at rates of 0.431, 0.459 and 0.97 respectively. Cumulated rainfall extremely wet day (R99p) showed positive insignificant increase with a rate of 0.852.

Rainfall trends related indices are not as uniform as the ones on temperatures (Figure 5.3 and 5.4). In Sotuba as well as in Ségou the results showed an overall reduction of rainfall over the 1961 to 2014 period. All indices in Sotuba revealed statistically insignificant, yet all the indices decreased in a moderate way (Table 5.4). However, in Ségou cumulated rainfall, extremely wet days and consecutive wet days showed statistically significant trends. Results showed that all rainfall indices, including both regions have declined over the last 53 years. That may be explained by natural variability and or a high response to increased greenhouse gases by anthropology or reduced aerosols (Haarsma *et al.*, 2005; Mohino *et al.*, 2011; Ackerley *et al.*, 2011; Biasutti, 2013). Rainfall represent meteorological element that best defines the climate of the tropics in general and of Mali in particular. Therefore, that decrease in rainfall clearly impacts well- being activities which rely on rainfed agriculture. The crop cycles follow the rhythm of rainfall. Precipitation is the main source of water during refuelling and layers. If any of the precipitation ever were to change, the consequence will be multiple for human life. Elsewhere, increased frequencies of extreme rainfall events such as long dry or wet spells mean weaker productive systems.

This decrease in annual total rainfall is confirmed in most part of Africa (Frappart *et al.*, 2009; Ozer *et al.*, 2009; Hountondji *et al.*, 2011). Similar finding was observed of the overall reduction in rainfall in the studies over the Sahel (Biasutti, 2013; Mohamed, 2011; Ackerley *et al.*, 2011; Lebel and Ali, 2009; Nicholson *et al.*, 2000). It was also found in the study done by New *et al.* (2006) non- significant, but in total rainfall positive trends from heavy events for the West Africa and Southern Africa region. Therefore, for the increasing trend of cumulated rainfall of extremely wet days from the late 1980s, this is in accordance with the findings of Sarr (2011) and Ly *et al.* (2013).
However, extreme rainfall events became more frequent in the West African Sahel during the last decade, compared to the 1961 to 1990 period (Ly *et al.*, 2013). Furthermore, studies in other parts of Africa showed in some cases similar trends, but in other cases the trends were different. For instance, Kruger (2006) found some significant increase and decrease occurred in annual precipitation, some increases in the longest annual wet spells and some increases in high daily precipitation amounts in southern Africa over the 1910- 2014 period.

Table 5. 4 Trend per percentage, days with positive and negative slopes of rainfall indicesextremes in Sotuba and Cinzana for 1961-2014 period

	Sotu	ba (1960-	-2014)	Cinz	ana (1960)-2014)
Indices	Mean	Slope	P-value	Mean	Slope	P-value
Maximums 5 days' rain (mm)	120.56	-0.209	0.459	100.16	1.68	0.057
Consecutive wet days (day)	5.11	-0.013	0.431	4.18	-0.029	0.009*
Extremely wet days (mm)	49.21	-0.105	0.852	22.76	2.156	0.029*
Annual total precipitation (mm)	929.64	-0.049	0.97	635.13	-0.856	0.541

*Trend statistically significant at p-value = 0.05





Figure 5. 3 Evolution of the maximum 5 day, precipitations (RX5 Day) and that of consecutive wet days (CWD) in Sotuba (a, b and d) Cinzana (a and c) from 1960 to 2014 period

BADW

C M C C A SHAL

SANE

NO



Figure 5. 4 Evolution of extremely wet days (R95p) and that of total annual precipitation (PRCPTOTE) in Sotuba (a, b and d) Cinzana (a and c) from 1960 to 2014 period

5.4 Conclusion

All temperature extremes in Sotuba and Cinzana showed warming trends in 1961- 2014 period. The warming trend was greater and significant in Cinzana than Sotuba. The finding of the study showed that there is a general decreasing trend for cold nights (TN10p) and cold days (TX10p) in Sotuba, while Segou showed high decreasing trend. A general warming trend in the temperature indices for warm nights (TN90p) and warm days (TX90p) since 1961 were observed. The significance of changes in precipitation extremes in Cinzana in 1960 and 2014 period was low, and only the trends in CWD and R99p were significant. In general, all precipitation extremes decreased, but the changes were not statistically significant.

CHAPTER 6: ASSESSING THE CHANGES IN YIELDS OF MILLET CROP UNDER DIFFERENT SCENARIOS OF CLIMATE CHANGE USING DSSAT MODEL

6.1 Introduction

In several region of the world, climate will probably adversely affect food production, especially in developing countries where a large fraction of the population already faces permanent hunger (Lobell and Burke, 2008). According to Dai *et al.* (2004) in sub-Saharan West Africa, the observed decrease in rainfall has been associated with an increase in temperature since the 1970s which has led to a decline in production (Barriose *et al.*, 2008; Traore *et al.*, 2013). IPCC (2013) showed that the projection of climate by the end of the century shows an increase of temperature by (1.1 °C-4.8 °C) and an increase of contrast between wet and dry season for the Sahelian region. Many studies assessed the impact of current and the future climate on crop production (Roudier *et al.*, 2011; Sultan *et al.*, 2013). Crop yields would decrease by 15 % probably because the effect of temperature decreasing the length of crop growth cycle and increasing water stress through higher evaporation losses even though the rainfall amount remains unchanged (Schlenker and Lobell, 2010). The impact of crop yield studies for Africa illustrates a wide dispersion of changes in yields ranging from -50 % to 90 % under various climate change scenarios (Roudier *et al.*, 2011), while the reported changes in crop yields are mostly negative (Challionor *et al.*, 2007). The projected impact is wider in the Sudano-Sahelian countries in West Africa, with an average yield loss of 18 % compared with an average yield loss of 13 % in the southern Guinean countries (Sultan *et al.*, 2013) and this difference is likely caused by the warmer and drier climates more in the northern countries.

Agriculture in Mali is very vulnerable to climate change. Rainfall remains uncertain for the future (Sultan and Janicot, 2003), while the increase in temperature is almost a certainty (Schwartz and Randall, 2003). This is a major challenge for agriculture which depends almost entirely on rainfall, as average temperatures would be close to highest values (1 to 2,75 °C or more by 2030) (Oxfam, 2007) that could be harmful to the growth and development of crops. Crop production in Mali is essentially characterized by subsistence farming based on millet and sorghum in the Central and Northern region, and commercial agriculture dominated by growing cotton and rice in the southern regions and the centre. The yield of rice and cotton have seen a considerable increase (cotton 1 t/ha and rice 5 t/ha), while those of rainfed food crops (sorghum 0.8 t/ha and millet, 0.7 t/ha.) evolved slowly in the last 50 years, despite being the base of food for more than 80 % of the population (DNS, 1995).

In Mali, future crop yield will vary between -17 % and +6 % at the national level (Butt *et al.*, 2005). Negative impacts of climate change on crop productivity increase in severity as warming intensifies, emphasizes the importance of coping with global warming (Traore, 2014). The IPCC's fifth Assessment Report (AR5) presents new evidence of climate change (IPCC, 2013) and adapting cropping systems to the likely climate change is essential. Several adaptation options which help Malian farmers to cope with current climate variability could be considered (Traore, 2014). Farm production practices are in general: water management, fertilisation, crop land increases but also asset management/income as diversification of activities and migration (Chuku and Okoye, 2009). Changing the sowing date to the starting of the season is another common

practice of farmers in the semi-arid regions (Muller *et al.*, 2010). Studies for West Africa's crop simulation studies showed that sowing date and cultivar type adaptation can reduce the negative climate change impacts and increase crop yields (Tingem and Rivingto, 2009). The same authors showed a simulated 15 % and 40 % decrease in maize and sorghum yield respectively caused by climate change was converted to 32 % and 18 % increase respectively with the use of different variety with a longer crop growing period. Similarly, Butt *et al.* (2005) argued that by implementing adaptive responses such as the use of high-temperature-resistant crop varieties together with addressing soil fertility decline and economic gains could exceed losses caused by climate change in Mali.

Crop models simulation is the complex interaction between soils properties, management factors genetics, pest and weather that influence performance of crops. Simulation of crop modelling has developed over several years critical with advances in crop physiology, ecology and computing technology (Mukkar and Hassana, 2011). However, among the numerous crop growth models the most largely used are the Decision Support System for Agro-Technology Transfer (DSSAT) model which were designed to stimulate development, growth and yields of crop growing on a uniform area of land as well as the changes in carbon, nitrogen and soil water that take place under the cropping system over time (Jones *et al.*, 2003). An important task in models experiments is the testing of their performance in a large range of circumstances to identify their scope of limitations and validity (Mukktar and Hassana, 2011). The objective of this current study was to assess the performance of DSSAT model in simulating the yield of two millet varieties in the two selected agro-ecological zones of Mali under different scenarios of climate change.

6.2 Materials and Methods

6.2.1. Experimental Design and Crop Management

The computed data were obtained from two experiments conducted at Agricultural Research Station of Sotuba and Cinzana in 2013 and 2014. A combination of two pearl millet varieties and four fertilizer treatments were used. The fertilizers treatments used were: T1 Control (no application); T2 MANURE of 23:10:17 (5000 kg/ha); T3 NPK of 15: 15: 15 (50 kg/ha) and T4 combination of NPK + MANURE The experiments were conducted in a randomised complete block design (RCBD) with four replications in a plot size of 10 m x 4.2 m. The varieties evaluated at Sotuba station were Sanioni, an improved local variety and Cho, a local variety while at Cinzana station Sosat, an improved variety form IER /ICRISAT and IBV8001, and an improved variety from ICRISAT were used.

The land was ploughed, harrowed and ridged in the two locations. The experiments were established at Sotuba on 15th July, 2013 and 11th June, 2014 and at Cinzana on the 22th July, 2013 and 15th July, 2014. The plants were sown on 6 rows 5 m long, 0.80 m between rows and 0.50 m within rows (between hills). Plants were thinned to 2 plants per hill. Planting depth was approximately 3 - 5 cm and seeds covered with soil to enhance good germination and good seedling establishment.

Table 6. 1	Experime	ent layout	for randor	nised comp	plete bock	design wit	<mark>h four re</mark> p	licates
Replication I	V 1 T 3	V 1 T 2	V_1T_1	V 2 T 4	V1T4	V ₂ T ₃	V ₂ T ₁	V ₂ T ₂
	~ >				5	aP	e	
Replication II	V 2 T 3	V 1 T 1	V 1 T 3	V 2 T 1	V1T2	V ₂ T ₄	V 1 T 4	V1T2
	-	14	2 SAL	JE P		P		
Replication II	V ₂ T ₄	V ₂ T ₁	V ₁ T ₂	V ₁ T ₁	V ₁ T ₃	V ₂ T ₃	V 1 T 3	V ₂ T ₃
Replication IV	V ₁ T ₄	V 1 T 1	V 2 T 4	V 2 T 3	V1T2	V ₂ T ₃	V 1 T 3	V ₂ T ₁

Note: $V_{1=}$ (SANIONI an improved local variety in Sotuba site, and SOSAT an improved variety from IER in Cinzana site) $V_{2=}$ (CHO a local variety in Sotuba and IBV8001 an improved variety

from ICRISAT) T₁₌ Control, T₂₌ MANURE, T₃₌ NPK, T₄₌ NPK +MANURE 6.2.2 Data Collected

Data were collected on all plants from the four central rows per plot in both locations. Data on seedling vigour were collected two weeks after planting and scored on a scale of 1 to 5 (1 = excellent, 2 = good, 3 = average, 4 = poor, 5 = very poor). The plant stand was evaluated as the number of plants of millet in each plot after thinning. The recording of day of 50 % heading was carried on the day at which 50 % in experiment unit plants started to have panicle from boot. The flowering was recorded when50 % of in an experimented unit had flowered.

The plant height was measured from the base of plant to tip of head at maturity. The data was taken on three plants per experimental unit and average three plants height were taken per plot. The number of harvested panicles was obtained by counting the number of productive panicles from the experimental plot. Panicle weight is the total of dry weight of the harvested panicle from the experimental plot. The numbers of hills harvested were collected by considering: numbers of harvested hills per plot that were taken; grain weight is the total grain weight of grain obtained after threshing the harvest panicles. Threshing and winnowing was done manually using sack and plastic bowls. After threshing and weighing, the seeds from the experimental unit, the weight of the 1000 grain was obtained and recorded. Grain yield was calculated using the following formula:

- Grain yield = Grain_____ weight per plot x 10000...... equation 6.1
- Straw weight was recorded by weight of all plant harvested and dried from the harvest unit.

6.2.3 Crop Model

The decision support system for agro-technology transfer (DSSAT)/CSM simulates growth, development and yield of a crop growing on a uniform area of land under prescribed or simulated management as well as the changes in soil water, carbon, and nitrogen that take place under the cropping system over time. A detail description of the model has been provided by Ritchie and Alagarswamy, (1989) and Hoogenboom *et al.* (2003).

6.2.4 Selection of Pearl Millet Varieties

DSSAT model is variety-specific and is able to predict millet yield and millet response to different environmental conditions. In projecting crop growth and yield, the model takes into account crop management, genetics, effect of weather and soil water, C and N. The model uses a detailed set of crop specific genetic coefficients, which allows the model to respond to diverse management and weather conditions. However, in order to get trustworthy results from the model simulations, it is important to have the appropriate genetic coefficient for the selected cultivars. The two pearl millet varieties SANIONI and CHO for Sotuba site and IBV8001 and SOSAT for Cinzana have been selected in the present study. A specific cultivar coefficient for the genotypes used in these experiments was not in the list of cultivars available within the model. The cultivar coefficients were adjusted, until main development and growth stages were made for parameters of the development and growth, the aim being sensitivity analyses of the model and improvement of the coefficients. The coefficients were decreased or increased if needed using a small step.

6.2.5 Soil and crop management input.

Soils in these zones are mainly sandy and loamy of low fertility and seasonally waterlogged or flooded clays. They are classified as Lixisol in the FAO classification. Soil-related modules were parameterized mainly with measured data from experiments carried out under optimal growth

conditions, and from related literature. Disturbed and undisturbed soil samples which were taken in soil profiles (0–10, 10–20, 20–40, 40–120 cm) prior to sowing, were analysed for organic carbon (OC%), pH in water, and particle size distribution as described in Hoogenboom et al. (1999). Input data related to soil characteristics include soil texture, number of layers in soil profile, soil layer depth, pH of soil for each depth, clay, silt and sand contents, organic matter, cation exchange capacity, etc. The soil profile data used in the parameterization of the model is presented in Table 6.2 and Table 6.3.

Depth	pН	pН	Sand	Silt %	Clay	OM	N %	CEC
(cm)	(w)	(KC)	%		%	% C		meq/100mg
0-10	5.51	4.54	81.67	11.00	7.17	1.89	0.15	2.62
10-20	5.44	4.37	76.67	9.67	13.00	1.76	0.14	3.43
20-40	5.45	4.35	70.17	9.00	20.83	1.46	0.15	3.00
40-120	5.45	4.35	70.17	9.00	20.83	1.46	0.15	3.00
	-							

Table 6. 2 Chemical properties at Cinzana site for model evaluation experiment

Table 6. 3 Chemical properties at Sotuba site for model evaluation experiment

Depth	p	H	pH	Sand	Silt	Clay	OM	N %	CEC	
(cm)	(v	N) ((KC)	%	%	%	% C	- 1	meq/100	mg
0-10	5.	77	4.83	72.06	21.28	6.59	1.32	0.88	5.47	
10-20	5.	70	4.63	70.13	19.25	10.69	1.24	0.85	5.10	
20-40	5.75	4.61	66.59	19.19	14.13	1.14 (0.73 5.4	44 40-120) 5.75	4.61
<u>66.59</u>	19.19	14.13	1.14	0.73	5.44	_			1	
Z				5	-		1		12	1

6.2.6 Experiment for model parameterization for crop yield simulation

In the present study, DSSAT v 4.6 (CERES-Millet) (Jones et al. 2003) was used to simulate crop yields as a function of current as well as future climatological conditions. Data from an experiment carried out between 2013 and 2014 at Sotuba and Cinzana Agronomic Research Station under rainfed conditions were used to parameterize the models. Daily weather data during the growing

season, were obtained from observations at the experimental stations. These include minimum and maximum temperatures, rainfall and sunshine hours which in turn are used to estimate solar radiation. Phenological data including planting date, date of flowering, date for start of grain filling and date of physiological maturity were collected. These were noted when 50% of plant population per plot attained each of these stages. The calibration was done using trial and error method of iteratively adjusting the parameters to obtain as close as possible the simulated and observed values of phenology (i. e. anthesis and maturity dates) and grain and biomass yields.

6.2.7 Modelling of Effect of Climate change and variability on millet grain yield

For climate change impact study, historical weather data was derived from the Agronomic Research Station for both sites for the period of 30 years from 1983 to 2012 and another 30 years' period projected weather from 2040 to 2070. The historical data were used to simulate grain yield variability on different pearl millet varieties for different treatments and also used as Baseline for the projected scenarios. Five scenarios were used based on GCMs (Global Circulations Models), where CMCC-CMS, CESM1-BGD, CCSM4, ACCESS1-0 and MRI-CGCM3 were described as Cold-dry, Hot-dry, Cold-wet, Hot-wet and Middle respectively. Cold and hot were defined as changes in temperature while wet and dry as changes in rainfall. Future weather scenarios used in the simulation for grain yields experiments were:

- Temperature increase (1.4° C) and (no change) in rainfall,
- Temperature increase in (2.4° C) and 10 % decrease in rainfall,
- Temperature increase in (1° C) and 30 % increase in rainfall,
- Temperature increase in (2.4° C) and 30 % decrease in rainfall, [] Temperature increase in

 (1.6° C) and 15 % decrease in rainfall.

The five GCMs were chosen based on their use in previous studies in the region and their better representation of projected climate, in terms of temperature and rainfall patterns in West Africa. The simulations were made using a fixed concentration of atmospheric [CO2] of 499 ppm for the Baseline (the value reported for the year 2010 in the fourth assessment report of IPCC.

6.2.8 Statistical Analyses

a) Experimental treatment

The effect of the treatments and their interactions were analysed for the four following parameters: plants height, 1000 grain mass, grain yield and straw yield, with ANOVA performed with GENSAT V.9 Discovery Edition 4, Release 10.3DE (PC/Windows 7), Copyright 2011, VSN International Ltd. (Roth Amsted Experimental Station).

Using the General Treatment Structure (in Randomized Complete Blocks), ANOVA model included treatments, year and their interactions. Duncan Significant Difference (DSD) test was used for post-ANOVA multiple comparison (P < 0.05) between means. Duncan significant tests were chosen because it can be used on raw data or in conjunction with an ANOVA test (Random complete block).

b) Data for model evaluation

The calibrated model was evaluated by comparing observed values for parameters of grain yield with those from model simulations. Model performance was assessed through various statistical parameters viz., coefficient of correlation, standard deviation, root mean square error (RMSE), relative error, model efficiency, were used to evaluate overall model performance (Loague and Green, 1991; Lecina *et al.*, 2003; Dust *et al.*, 2000; Ali *et al.*, 2004; Liu *et al.*, 2011)

i. Root Mean Square Error (RMSE)

The Root Mean Square Error also called the Root Mean Square Deviation (RMSD) is a frequently used measure of the difference between values predicted by a model and the values actually observed from the environment that is being modelled. The individual differences are also called residuals and, the root mean square serves to aggregate them into a single measure of predictive power

i.
$$RMSE = \sqrt{\frac{\sum_{i=1}^{n} (P_i - O_i)^2}{n}}$$
Equation 6.2

where *n* refers to the number of observations, P_i and O_i signifies the simulated and observed values, respectively, and *M* denotes the observed mean value, according to Loague and Green (1991).

ii. Model Efficiency

$$EF = \Sigma$$
 (measured- Σ measure(measured)

mean-measured)² - Σ (simulated mean)₂-measured)² Equation 6.3

6.3. Results and discussion 6.3.1. Inter-Annual Rainfall Variability, Rainfall Distribution and its Relation with Crop

Yields for Sotuba and Cinzana

The rainfall distribution during the cropping season of 2013 and 2014 at Sotuba is illustrated in Figure 6.1 A and B. Overall, 60 to 75 % of the total rain fell between June and September in Sotuba. In the same location in that year, 2013 was the wettest year with 713 mm, compared to 2014 year with 640 mm. The distribution of rainfall also varied from year to year and the analyses showed that in 2014 the rainfall was better distributed than in 2013. In 2013, it was observed that a 7 days' dry spell occurred in September, while in 2014 no day dry spell occurred even though the rainfall started in April with 0.4 mm of rainfall. In both years, the end of the rainy season did not show any dry spell period.

Figure 6.1 C and D illustrate the rainfall distribution at Cinzana during the cropping season in 2013 and 2014. Though, 60 to 85 % of the total rain occurred between July and September in Cinzana.



Figure 6. 1 Rainfall distribution in 2013 and 2014 in Sotuba (A and B) and in Cinzana (C and D)

Contrary to Sotuba, the year 2014 at Cinzana was wetter with 443 mm compared to the rainfall of 382 mm recorded in 2013. Two dry spells periods occurred in 2013 with 9 and 18 days' dry spell period in July and October respectively. In 2014 two dry spell periods was also observed with 9

and 11 days' dry spell periods in August and September. In both years, the rain started in April with rainfall of 0.4 and 6.6 mm for 2014 and 2013 respectively. The end of the rainy season was dry in 2013, with only 0.63 mm of rain in October against 0.91 mm in 2014. The rainfalls in both years at Sotuba were higher than that for Cinzana. This is expected because Sotuba is located in Sudanian zone while Cinzana is in the Sahelian zone. In 2013, dry spell periods were observed in both locations. Sotuba experienced only one dry spell period in year 2013 while it appeared twice in both years at Cinzana. The dry spell period at Sotuba that occurred in September corresponded to the flowering period of the crops. In 2013 at Cinzana the dry spell periods occurred in July corresponding to planting period of the crop while the one in October corresponded to the plant maturity period. The dry spell periods occurred in 2014 at Cinzana in August and September are the periods of the crop flowering and grain filling. Study done by Le Barbé et al. (2002), on rainfall variability in West Africa during the years 1950-90 showed similar results, where they defined that the ration between the number of rain events and the number of rainy day will depend on the location and period of year considered. In the Southern part of the West African Sahel the rainfall season starts earlier and end later than in the north (Traoré et al., 2000; Traoré et al., 2007).

The analysis of rainfall revealed sequences of dry spells which may affect crop yields. The highest in dry spell of 5 to 10 occurred in May and October (result not shown in the report). This showed the irregularity of rainfall in May which represents the land preparation in Sudanian-Sahelian zones of Mali, and which may delay the planting date and reduce the rainy season length for cropping period. Many farmers agreed that the rainy season which start early are mostly better for crop production than late rainy season (Sivakumar and Hatfield, 1990; Stewart, 1991). This is confirmed by the finding in 2014 at Sotuba by the researcher where the rainy season started early without any dry spell period in the cropping season resulting in higher yield. The date of started of rainy season was demonstrated to be the key variable to all other seasonal rainfall (Traoré *et al.*, 2014).

6.3.2. Analysis of Variance of Agronomic Parameters Studied

a) Plant Height

At Sotuba, the average plant height was 2.9 m and 3.6 m for 2013 and 2014 respectively (Figure 6.2) and CV of 10.9 %. The analyses showed high significant differences (<0.001) between the years for plant height (Table 6.4). It also showed a high significant difference (<0.001) between years and varieties interaction (Table 6.4). The plant height of CHO and Sanioni varieties varied according to year. At Cinzana, the average plant height observed was 2.3 m in 2013 and 3.6 m in 2014 and a CV of 11.7 % (Table 6.5). The analyses showed high significant differences (<0.001) between year 2013 and 2014 and did not show any statistical differences for any interactions.

Table 6. 4 Summary of analysis of variance on Yields and Yield components of two (2) varieties evaluated in Sotuba in the Sudanian agro-ecological zone

				F pr.	1
		Grain Yield	l Straw Y	ield 1000 grain weight	Plant
Source of variance	df	kg ha ⁻¹	kg ha ⁻¹	(g)	height (m)
Years (Y)	1	0.002	<.001	0.001	<.001
Varieties (V)	1	0.683	<.001	<.001	0.094
Fertilizers (F)	3	<.001	<.001	0.223	0.256
Y x V	1	0.349	0.003	<.001	<.001
Y x F	3	<.001	0.261	0.397	0.449
V x F	3	0.631	0.816	0.245	0.944
Y x V x F	3	0.698	0.866	0.887	0.857
Residual	45				
Total	63				-
CV%		18.1	20.0	8.4	10.9



106

Figure 6. 2 Varieties for plant height yields in 2013 and 2014 *b) 1000 Grain Weight*

At Sotuba, the average 1000 grain weight was 9.6 g and 8.7 g for CHO and SANIONI, respectively. The analyses revealed significant differences between the two years (0.001) (Table 6.4). Highly significant differences were observed between varieties (<0.001) and the interaction between years and varieties (<0.001) was significant. In 2013, the variety CHO showed higher 1000 grain weight with 10.17 g compared to variety SANIONI (7.53 g) while in 2014 the variety SANIONI had higher 1000 grain weight (9.94 g) compared to variety CHO with (9.12 g). The 1000 grain weight gap between the year 2013 and 2014 for variety SANIONI was higher than that of the variety CHO for the same years (Figure 6.3). On the contrary, at Cinzana, the analysis did not show any statistically significant differences for the variable 1000 grain weight.



Figure 6. 3 Varieties 1000 grain weight in 2013 and 2014 in Sotuba

NE

			<u>F p</u>	<u>r.</u>	
	Dí	Grain Yield k	g Straw Yield k	g 1000 grain	Plant
Source of variation	Dī	na	na	weight (g)	neight (III)
Years (Y)	1	0.094	0.04	0.925	<.001
Varieties (V)	1	0.779	0.129	0.328	0.244
Fertilizers (F)	3	0.055	0.058	0.8	0.33
Y x V	1	0.02	0.452	0.791	0.236
Y x F	3	0.002	0.387	0.984	0.448
V x F	3	0.711	0.399	0.742	0.985
Y x V x F	3	0.2	0.331	0.691	0.821
Residual	45				
Total	63				
CV%		22.4	30.9	16.7	11.7

Table 6. 5 Summary of analysis of variance on Yields and Yield components of two (2)varieties evaluated in Cinzana a Sahelian agro-ecological zone

c) Straw Weight

At Sotuba, the average straw yield was 9714 kg ha⁻¹ and 31487 kg ha⁻¹ for year 2013 and 2014 respectively. The analyses revealed high significant differences between years (<0.001), between varieties (<0.001) and among fertilizer treatments (<0.001). Table 6.4 showed significant differences between years and varieties interaction (0.003) with a CV of 20 %. Figure 6.4 showed an interaction between years and varieties. The total straw produced in 2014 (34999 kg ha⁻¹) by the variety SANIONI was 3 times higher than produced in 2013 (1000 kg ha⁻¹) by the same variety. In 2014, the local variety CHO produced higher straw production with 27975 kg ha⁻¹. In 2014, the quantity of straw produced by SANIONI was higher than that produced by CHO while in 2013 both varieties had similar amounts of straw produced.

At Cinzana, the average straw quantity produced was 9270 kg ha⁻¹ in 2013 and 7867 kg ha¹ in

2014 and a CV of 30.9 % (Table 6.5). The analyses showed significant differences (0.04) between year 2013 and 2014 and did not show any statistical difference for the other factors and their interactions.



Figure 6. 4 Straw yields in 2013 and 2014 in Sotuba

d) Grain Yield kg

At Sotuba, the average grain yield was 1293 kg ha⁻¹ and 1503 kg ha⁻¹ for year 2013 and 2014 respectively. The analyses revealed significant differences between years (0.002). Table 6.4 showed high significant differences among fertilizer treatments (<0.001) and between years and fertilizer interaction (<0.001) with a CV of 18.1 %. In 2013, the analyses revealed high grain yield in the combined organic manure and NPK (NPK+MANURE) treatment with 1948 kg ha⁻¹, followed by NPK (1281 kg ha⁻¹), MANURE (1130 kg ha⁻¹) and Control (813 kg ha⁻¹). In 2014, the grain yield with fertilizer treatment (NPK+MANURE) was 1602 kg ha⁻¹, followed by NPK with grain yield (1502 kg ha⁻¹), MANURE was (1466 kg ha⁻¹) and control was (1440 kg ha⁻¹) (Figure 6.5). The grain yield gap for the Control treatment in 2013 (813 kg ha⁻¹) and in 2014 (1440 kg ha¹)

was high compared to the NPK+MANURE, MANURE and NPK treatment grain yield gap for the same years.



Figure 6. 5 Fertilizer and years for grain yields in 2013 and 2014 in Sotuba

At Cinzana, the average grain yield observed in 2013 and 2014 was 1390 and 1530 kg ha⁻¹ respectively. The analyses did not show any significant differences between the varieties and among the fertilizer treatments. However, significant differences were found in the following interactions: years x varieties (0.02) and years x fertilizer treatments (0.002) with a CV of 22.4 % (Table 6.5). In 2013, variety SOSAT showed higher grain yield (1447 kg ha⁻¹) than IBV8001 (1333 kg ha⁻¹) while in 2014, variety IBV8001 showed higher grain yield (1614 kg ha⁻¹) than SOSAT (1545 kg ha⁻¹). Figure 6.6a showed an interaction between years and varieties; thus, the grain yield gap of the variety IBV8001 in 2013 (1333 kg ha⁻¹) and in 2014 (1614 kg ha⁻¹) was higher compared to SOSAT in year 2013 (1447 kg ha⁻¹) and in 2014 (1545 kg ha⁻¹). In 2013, the analyses revealed higher grain yield with the treatments of NPK+MANURE, (1440 kgha⁻¹), NPK (1429 kg ha⁻¹), MANURE (1416 kg ha⁻¹) compared to Control (1276 kg ha⁻¹). In 2014, the Figure 6.6b showed the highest grain yield with NPK+MANURE (1915 kg ha⁻¹) followed by NPK (1561 kg ha⁻¹),

MANURE (1523 kg ha⁻¹) and Control (1121 kg ha⁻¹). The grain yield gap for the combined NPK+MANURE treatment in 2103 (1440 kg ha⁻¹) and in 2014 (1915 kg ha¹) was high compared to the others for the fertilizer grain yield gap for both years.



Figure 6. 6a and b. Year x varieties and year x amendments for grain yields in 2013 and 2014 in Cinzana

The rainfall at Sotuba in 2013 (713) and 2014 (640) were lower for the average rainfall of that location which was between 800 and 1100 mm. The same trend was observed at Cinzana with rainfall in 2013 (382 mm) and 2014 (443 mm) compared to the average rainfall between 650 and 800 mm. Rainfall is process by nature; therefore, lack of reliable rainfall covers the positive response of a crop to soil fertility management strategies for the crop productivity is increasing in West Africa Sahel (Bationo *et al.*, 1990; Rezaei *et al.*; 2014). At Sotuba the highest grain yield was obtained with NPK+MANURE treatment. The differences among the grain yields of the different fertilizer treatments compared to the Control were higher in 2013 showing the advantage of using fertilizer in the cropping system. At Cinzana, beside the low rainfall, the varieties showed differences according to the years and the fertilizer treatments. The differences among the grain yields showing the grain yields of the different fertilizer treatment fertilizer treatments compared to the Control were higher in 2013 showing the grain yields of the differences according to the years and the fertilizer treatments. The differences among the grain yields showed differences according to the years and the fertilizer treatments. The differences among the grain yields showed higher the different fertilizer treatments compared to the Control were higher in 2014 showing the grain yields of the different fertilizer treatments compared to the Control were higher in 2014 showing the grain yields of the different fertilizer treatments compared to the Control were higher in 2014 showing the grain yields of the different fertilizer treatments compared to the Control were higher in 2014 showing the grain yields of the different fertilizer treatments compared to the Control were higher in 2014 showing

the advantage of using fertilizer in the cropping system. Furthermore, pearl millet is adapted to dry conditions and is sensitive to water stress particularly when this occurs during the flowering stage as was the case in the present study in 2013 and 2014 for the Cinzana location.

Millet is one of the most important foods in the Sahelian and the Sudano-Sahelian zones of Mali. The lower growth and yield parameters recorded in varieties could be attributed to uneven rainfall distribution, which cause water availability problems that affect soil aeration, amendment use efficiency and plant metabolism such as photosynthesis thus inducing good plant growth and development. Therefore, in this study, pearl millet responded positively to different fertiliser treatments imposed and manure application improved millet yield production significantly as compared to the no fertilizer (Control) application for both sites. There was a significant increase of grain yield production when manure was added to NPK.

6.3.3. Validation and Evaluation of the CSM-CERES Millet Model for Sotuba and Cinzana

The model adequately simulated the grain yields. The linear regression of simulated versus observed grain yield for Sotuba location genotype CHO (0.98 and 0.88) and SANIONI (0.96 and 0.92) for 2013 and 2014 respectively. While in Cinzana location, genotype SOSAT (0.68 and 0.99) and IBV8001 (0.91 and 0.96) for 2013 and 2014 respectively. The correlation had intercepts not significantly different from zero and the slopes not significantly different from one (P > 0.05).

The analysis coefficient of determination (\mathbb{R}^2) showed that the value of \mathbb{R}^2 for all genotypes grain yields are most often, within the acceptable limit ($\mathbb{R}^2 > 0.5$). The values of \mathbb{R}^2 for all genotypes indicated that a good correlation exists between the observed and simulated yields (Figure 6.7 a, b, c and d). The highest grain yield observed corresponded to the highest grain yield simulated and vice versa. It so happened for the corresponding measures taken for both simulated and observed. Figure 6.7 shows that for the 2 years' period of experimental results for 2013 and 2014 in the in the two sites, the DSSAT models caught yields variabilities for different genotypes. Therefore, when analyzing year by year, a very good correlation was found in Sotuba location with ($R^2=0.98$) and 0.88) for CHO genotypes and SANIONI ($R^2=0.96$ and 0.92) in 2013 and 2014 respectively. The same case was found in Cinzana location with genotype Sosat (R²=0.68 and 0.99) and IBV8001 (R²=0.91 and 0.96) in 2013 and 2014 respectively. Therefore, in both years and among genotypes and each location the model performance (\mathbb{R}^2) was high and there was a systematic overestimation of millet yield. But with SOSAT genotype (R^2) there was a difference between the two years, in 2013 which was in the acceptable range while for 2014 the yields were overestimated. This could be explained by the relatively even rainfall distribution throughout the season in 2013 and 2014, when very erratic rainfall patterns were observed with heavy downpours alternating with long early cessation or dry spell of rains (Traoré et al., 2010). The same authors found out that, these discrepancies in the performance of the model could be explained by the poor assessment of the soil and crop water status during the reproductive and maturation phases. Therefore, sensitivity analysis showed that the response of the model is quite dependant on parameters such as runoff coefficients, rooting depth and soil water holding capacity, none of which were measured during our experimental trials.





Figure 6. 7 Comparison between simulated and observed grain yields for Sotuba (a and b) and Cinzana (c and d) location with different varieties

Table 6.6 showed the values of root mean square error (RMSE) of Cinzana location for the grain yield for SOSAT and IBV8001. For the experimental year 2013 and 2014 for varieties SOSAT and IBV8001 the RMSE were (19.9 and 16.9 %) and (11.3 and 19 %), respectively. The overall percentage value of model efficiency above 50 % indicated good model performance thus for SOSAT and IBV8001 in 2013 (14.4 and 51.9 %) and 2014 the model efficiency (50.0 and 41.1%) indicated an overall good model performance for the Cinzana location.

	Varieties	RMSE %	
Years			Model Efficiency (%)
2012	Sosat	19.9	14.4
2015	IBV8001	16.9	51.9
2014	Sosat	11.3	50.0
2014	IBV8001	21.1	41.1

Table 6. 6 Statistical parameters of simulation performance for the Cinzana location

Sotuba location (Table 6.7) shows the values of root mean square error (RMSE) for the grain yield for SANIONI and CHO for the experimental year 2013 and 2014 for varieties SANIONI and CHO. The RMSE was (19.4 and 20.0 %) and (16.6 and 20.3 %) for 2013 and 2014, respectively. The overall percentage of model efficiency value above 50 % indicated good model performance thus for SANIONI and CHO in 2013 (85.3 and 14.4 %) and 2014 the model efficiency (70.1 and 68.9 %) indicated an overall good model performance for Sotuba location.

Years	Varieties	RMSE %	Model Efficiency (%
-	Sanioni	19.4	85.3
2013	C X	2	CAR.
	Cho	20.0	14.4
	Sanioni	16.6	70.1
2014	Sanioni	10.0	/0.1
	Cho	20.3	68.9

Based on the regional climate change conditions of the two agro-ecological zones, pearl millet has been simulated using DSSAT crop model. The results showed that the simulated achieved higher potential yield compared to the observed yields. The statistical parameters (Table 6.6 and 6.7) suggest that overall performance of the DSSAT model in the simulating chemical and organic fertilizer amendments under rainfed pearl millet production is good and able to project grain yields with high accuracy. And the analyses indicated good model performance. Therefore, with regards to agronomic conditions for the observed experiments and simulated situation cannot make a one to one mapping possible, then it can be argued that the value RMSE and model efficiency indicate good model performance. However, for the perfect models, a 100 % value should be observed for model efficiency. Higher percentage values indicate good model performance, thus the results of the model efficiency for Sotuba and Cinzana indicate an overall good model performance for all varieties.

Since the DSSAT was run using real field conditions, hence, it does not account for yield loss caused by harvest practices, diseases and pest and birds attacks; it can then be concluded that variations between observed and simulated is sufficiently accounted. The yield differences can also be attributed to soil types and degradation caused by unsustainable agronomic practices. However, the same approach was found in a study on rice simulation done by Oteng-Darko *et al.* (2012) where good correlation was found between observed and simulated yields with an average relative difference of 12.28 (<30 %).

6.3.4 Simulated millet grain yield under climate variability over 1983 – 2012 in Sotuba and Cinzana

Since the model is able to give good simulation of millet grain yield for all the treatments at both site for both varieties, the grain yields were simulated for the period 1983 - 2012 under historical weather conditions. High variability was observed in the grain yield over the past 30-years simulation period for all the treatments (Figure 6.8). For both sites and all varieties, the trend of simulated grain yields was Control < MANURE<NPK<MANURE+NPK. Furthermore, the lowest grain yields varied between 200 and 300 kg whereas the highest values were about 1500 and 2250 kg for Sotuba and Cinzana, respectively.



- CONTROL ---- NPK ---- MANURE ---- MANURE+NPK

Figure 6.8. Simulated millet grain yield under historical weather data for Cinzana (a and b) and Sotuba (c and d)

The study first established where and how much crop yield varied with the varieties and then identify how much of the year to year variation in crop yields with the different treatment was explained by year to year variation in the climate. In general, the coefficients of variation (CV) or yield variability normalized by mean yields were lower in Cinzana with IBV8001variety under NPK and MANURE+NPK treatments by 14 and 12 %, respectively. Similarly, the CV for SOSAT

variety were 18 and 14 %, respectively with the same treatments at the same site (Table 6.8). As far as Sotuba is concerned, CV were slightly higher for CHO variety with CV of 23 and 18 % under the NPK and MANURE+NPK treatments, respectively. However, SANIONI variety showed low variability (14 % for both) under the same treatments (Table 6.9). Conversely, higher variability was found in Control and MANURE in both locations and both varieties. Higher CV (30 % for both treatments), which indicates the greatest variability in millet grain yield, were observed in Sotuba with SANIONI variety under Control and MANURE treatments (Table 6.9). Globally, the highest yield variability was found in Sotuba experimental site.

The results showed that millet yield responded to the different historical climatic conditions under all treatments. In line with the experimental observed data, the highest historical grain yields were achieved when the combination NPK +MANURE was applied. This shows the beneficial effect of combined application of organic and mineral fertilizer inputs the integrated soil management for Lixisol soil fertility. The relative increase in yield by NPK+MANURE combination application could be attributed to improvement in soil structure, nutrient retention and water for plant use (Arunah et al., 2006). The study showed how much of the year-to-year variability in crop yield was associated with climate variability within the study areas. As reported by Tingen et al., (2008) in Cameroon, this study also found that millet yield variability was explained only by a complex relationship between both temperature and precipitation variability. In conjunction with fluctuation in agricultural production as affected by climate variability, low food stocks can particularly contribute to changes in food price. Also location with high crop yield variability would disproportionately contribute to this effect especially if it is also the major granary. In Mali, semiarid to arid conditions are depend with seasonal rainfall distribution. The study showed that an important climatic feature which is high annual variability in grain yield in Sotuba (Sudanian zone) than Cinzana (Sahelian zone). However, this may explain that the actual effect on millet grain

yields variability depend on the agro-ecological zones for the specific crop and management approach. Pearl millet is more adapted in the Sahelian zone than the Sudanian zone.

	/		IBV8001	
	Control	NPK	Manure	Manure+NPK
Mean	784	1391	1010	1726
stdev	182	192	228	202
CV (%)	23	14	23	12
			SOSAT	
Mean	808	1357	1039	1743
stdev	182	243	225	242
CV (%)	23	18	22	14
Table 6.9	9. Historical aver	age crop yi	elds and coefficie	ent of variation in Sotul
		S.	СНО	
	Control	NPK	Manure	Manure+NPK
Mean	580	1015	717	1232
stdev	144	231	187	222
CV (%)	25	23	26	18
~	10	D-3	SANIONI	177
Mean	496	1011	607	1154
stdev	146	138	183	162
CV (%)	30	14	30	14

Table 6.8. Historical average crop yields and coefficient of variation in Cinzana

Simulation of climate change effects on millet grain yield showed that all scenarios underestimated crop yield compared to the Baseline for all treatments and both varieties in Cinzana (Figure 6.9a and b). For the variety IBV8001, the Control and MANURE treatments had the lowest grain yield (< 1000 kg/ha) whereas NPK +MANURE and NPK yielded over 1000 kg/ha. Crop yield under the Control and NPK treatments were closer among the scenarios and compared to the Baseline. However, the outputs for all the scenarios under NPK+ MANURE and MANURE and MANURE were very low compared to the Baseline, which reached about 2000 kg/ha. As indicated in Table 6.10, ANOVA

^{6.3.5.} Effect of climate change scenarios on millet grain yield for Cinzana and Sotuba

test and LSD confirmed the significant differences (P < 0.05) between the scenario outputs under all treatments. Among all scenarios, ACCESS1-0 (Hot-Wet) showed the lowest grain yields under the four treatments (Figure 6.9a). The percentage of grain yields decrease as simulated by the scenarios varied between 3% for the Control and NPK under CMCC-CMS, and 24% for Manure under ACCESS1-0 (Figure 6.10a). Slight increase of 1 % was observed for the NPK treatment under CESM1-BGD.

Regarding the SOSAT variety, all the scenarios for the four treatments estimated low grain yield compared to the Baseline, except Control and NPK for which the CMCC-CMS (Cold-Dry) and CESM1 -BGC (Hot-Dry) yielded higher than Baseline for lower probabilities (Figure 6.9b). Similar to IBV8001 variety, all the scenarios under NPK+ MANURE and MANURE simulated very low grain yields compared to the Baseline. The lowest grain yields were of 200 kg/ha for Control and MANURE, and 900 kg/ha for NPK and NPK+ MANURE. ANOVA test and LSD revealed the significant differences (P < 0.05) among the scenario outputs for Control, NPK+MANURE AND MANURE (Table 6.10). Meanwhile, no significant difference was observed among the average grain yields for all the scenarios under NPK treatment. Similar to IBV8001 variety, SOSAT showed lowest grain yields under the four treatments for ACCESS1-0 (Hot-Wet). The grain yields decrease reached a percentage of 1% for the NPK under CCSM4, and 40% for Manure under ACCESS1-0 (Figure 6.10b). Increases of 1 to 5 % were observed for the NPK and Control under CMCC-CMS, CESM1-BGD and MRI-CGCM3.

IBV8001 Variety (a)



Figure 6.9 a and b. Effect of climate change scenarios on millet grain yield for Cinzana



SCONTROL SMANURE SNPK MANURE+NPK

Similar to Cinzana site, simulation of climate change effects on millet grain yield in Sotuba, showed an underestimation of crop yield for all scenarios compared to the Baseline for all treatments and both varieties (Figure 6.10a and b). For CHO variety, all the scenarios for the four treatments estimated low grain yield compared to the Baseline, apart from Control and NPK for which the CMCC-CMS (Cold-Dry) and CESM1 -BGC (Hot-Dry) yielded higher than Baseline for very low probabilities (Figure 6.10a). NPK+ MANURE and MANURE simulated extremely low grain yields compared to the Baseline. The lowest grain yields were of 150 kg/ha for both Control

Figure 6.11 Percentage Changes of Millet Grain Yields for Cinzana (a and b) and Sotuba (c and d) under Different Scenarios

and MANURE, and about 400 kg/ha for both NPK and NPK+ MANURE. Similar to IBV8001 variety in Cinzana, ANOVA test and LSD revealed significant differences (P < 0.05) among the scenario outputs for all treatments (Table 6.11). Meanwhile, CHO variety showed lowest grain yields under all treatments for ACCESS1-0 (Hot-Wet). The decrease in grain yields reached a percentage of 2% for NPK under MRI-CGCM3, and 46% for Manure under ACCESS1 -0 (Figure 6.10c). NPK and Control under CMCC-CMS and CESM1-BGD induced increases in grain yields by 1 to 4%.

As far as SANIONI variety is concerned, all the scenarios and treatments estimated low grain yield compared to the Baseline, apart from Control and NPK for which the CMCC-CMS (Cold-Dry) and CESM1 -BGC (Hot-Dry) showed higher grain yield than Baseline for very low probabilities (Figure 6.10b). MANURE+NPK and MANURE simulated under all the scenarios very low grain yields compared to the Baseline. The lowest grain yields were about 100 kg/ha for both Control and MANURE, and about 600 kg/ha for both NPK and NPK+ MANURE. For scenario ACCESS10 (Hot-Wet), SANIONI variety showed lowest grain yields under all treatments. ANOVA test and

LSD revealed significant differences (P < 0.05) among the scenario outputs for all treatments (Table 6.11). Simulations for the Control under CMCC-CMS, and MANURE under ACCESS1 0 decreased by 4 % and 46%, respectively (Figure 6.10d). Control under CESM1-BGD induced increases in grain yields by 4%.

CHO Variety (a)

123



Table 6.10 Changes in grain yields for different climate scenarios and treatments in Cinzana.

	IBA	/8001	
CONTR	<u>OL MANURE</u>	NPK	MANURE+NPK

Scenario	Mean	Mean	Mean	Mean
Baseline	784±182 A	1009±228 A	1390±192 A	1726±2-2 A
Cold/Dry	757±167A	754±168 BC	1344±151 AB	1343±151 B
Hot/Dry	624±190 B	620±190 CD	1250±162 BC	1246±163 BC
Cold/Wet	778±142 A	775±142 B	1237±145 BC	1236±145 BC
Hot/Wet	605±179 B	601±179 D	1158±133 C	1154±134 C
Middle P-	<u>685±180 AB</u>	<u>681±179 BCD</u>	1281±156 AB	1278±156 B
value	0.000	0.000	0.000	0.000
		SOSAT		
	CONTROL	MANURE	NPK	MANURE+NPK
				_
Scenario	Mean	Mean	Mean	Mean
Scenario Baseline	Mean 808±182 A	Mean 1038±25 A	Mean 1356±243 A	Mean 1743±242 A
Scenario Baseline Cold/Dry	Mean 808±182 A 787±168 A	Mean 1038±25 A 785±169 BC	Mean 1356±243 A 1398±191 AB	Mean 1743±242 A 1394±190 B
Scenario Baseline Cold/Dry Hot/Dry	Mean 808±182 A 787±168 A 655±193 B	Mean 1038±25 A 785±169 BC 651±193 CD	Mean 1356±243 A 1398±191 AB 1313±187 AB	Mean 1743±242 A 1394±190 B 1309±187 BC
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet	Mean 808±182 A 787±168 A 655±193 B 808±141 A	Mean 1038±25 A 785±169 BC 651±193 CD 805±142 B	Mean 1356±243 A 1398±191 AB 1313±187 AB 1386±145 AB	Mean 1743±242 A 1394±190 B 1309±187 BC 1383±145 BC
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet Hot/Wet	Mean 808±182 A 787±168 A 655±193 B 808±141 A 635±183 B	Mean 1038±25 A 785±169 BC 651±193 CD 805±142 B 631±183 D	Mean 1356±243 A 1398±191 AB 1313±187 AB 1386±145 AB 1260±158 B	Mean 1743±242 A 1394±190 B 1309±187 BC 1383±145 BC 1256±158 C
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet Hot/Wet <u>Middle</u> P-	Mean 808±182 A 787±168 A 655±193 B 808±141 A 635±183 B 714±180 AB	Mean 1038±25 A 785±169 BC 651±193 CD 805±142 B 631±183 D 711±180 BCD	Mean 1356±243 A 1398±191 AB 1313±187 AB 1386±145 AB 1260±158 B 1346±174 AB	Mean 1743±242 A 1394±190 B 1309±187 BC 1383±145 BC 1256±158 C 1343±173 BC
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet Hot/Wet <u>Middle</u> P- value	Mean 808±182 A 787±168 A 655±193 B 808±141 A 635±183 B <u>714±180 AB</u> 0.000	Mean 1038±25 A 785±169 BC 651±193 CD 805±142 B 631±183 D 711±180 BCD 0.000	Mean 1356±243 A 1398±191 AB 1313±187 AB 1386±145 AB 1260±158 B 1346±174 AB 0.051	Mean 1743±242 A 1394±190 B 1309±187 BC 1383±145 BC 1256±158 C 1343±173 BC 0.000

		SANIONI		
	CONTROL	MANURE	<u>NPK</u>	MANURE+NPK
Scenario	Mean	Mean	Mean	Mean
Baseline	496±146 AB	606±184 A	1170±211 A	1153±162 A
Cold/Dry	471±128 AB	470±130 B	952±114 B	947±118 B
Hot/Dry	347±140 C	345±140 C	854±104 C	845±109 C
Cold/Wet	507±120 A	505±122 AB	956±118 B	951±121 B
Hot/Wet	337±136 C	335±136 C	804±117 C	796±121 C
Middle P-	<u>400± 141 BC</u>	399±142 BC	887±124BC	880±128 BC
value	0.000	0.000	0.000	0.000
3		СНО	5	1
E	CONTROL	MANURE	NPK	MANURE+NPK
and the second second				
S	-			2
Scenario	Mean	Mean	Mean	Mean
Scenario Baseline	Mean 580±144 A	– Mean 717±186 A	Mean 1015±231 A	Mean 1231±221 A
Scenario Baseline Cold/Dry	Mean 580±144 A 548±132 A	Mean 717±186 A 547±134 B	Mean 1015±231 A 1017±179 A	Mean 1231±221 A 1012±180 B
Scenario Baseline Cold/Dry Hot/Dry	Mean 580±144 A 548±132 A 419±142 BC	Mean 717±186 A 547±134 B 417±143 C	Mean 1015±231 A 1017±179 A 930±155 AB	Mean 1231±221 A 1012±180 B 920±157 BC
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet	Mean 580±144 A 548±132 A 419±142 BC 580±126 A	Mean 717±186 A 547±134 B 417±143 C 578±127 B	Mean 1015±231 A 1017±179 A 930±155 AB 1028±171 A	Mean 1231±221 A 1012±180 B 920±157 BC 1023±172 B
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet Hot/Wet	Mean 580±144 A 548±132 A 419±142 BC 580±126 A 399±141 C	Mean 717±186 A 547±134 B 417±143 C 578±127 B 397±141 C	Mean 1015±231 A 1017±179 A 930±155 AB 1028±171 A 881±139 B	Mean 1231±221 A 1012±180 B 920±157 BC 1023±172 B 873±142 C
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet Hot/Wet Middle	Mean 580±144 A 548±132 A 419±142 BC 580±126 A 399±141 C 473±144 BC	Mean 717±186 A 547±134 B 417±143 C 578±127 B 397±141 C 472±144 BC	Mean 1015±231 A 1017±179 A 930±155 AB 1028±171 A 881±139 B 962±169 AB	Mean 1231±221 A 1012±180 B 920±157 BC 1023±172 B 873±142 C 954±170 BC
Scenario Baseline Cold/Dry Hot/Dry Cold/Wet Hot/Wet Middle P-value	Mean 580±144 A 548±132 A 419±142 BC 580±126 A 399±141 C 473±144 BC 0.000	Mean 717±186 A 547±134 B 417±143 C 578±127 B 397±141 C 472±144 BC 0.000	Mean 1015±231 A 1017±179 A 930±155 AB 1028±171 A 881±139 B 962±169 AB 0.007	Mean 1231±221 A 1012±180 B 920±157 BC 1023±172 B 873±142 C 954±170 BC 0.000

. Figures 6.9 (a and b) and Figures 6.10 (a and b) showed the simulated effects of the scenarios of rainfall and temperature changes on millet grain yields for the period 2040 to 2069. The changes are computed as averages across the 29 GCMs simulations for each agro-ecological zone. Future climate projections from GCMs showed decreasing millet grain yields over 2040 – 2069 compared to the Baseline period (1983 - 2012) at both sites. Simulations showed higher grain yields under historical weather data compared to the situation with the changing climate future scenarios. The study showed that an increase in temperature and precipitation would be the main unfavourable driver of the future change in pearl millet yields. The effects of climate change and variability are more acute in the Sudanian zone for both varieties and treatments. This may be explained by the fact that in the Sudanian zone, a semi-arid area with mean annual rainfall between 800 - 1100 mm, the increase in temperature in that among of rainfall may provoke excessive and useless soil moisture damageable to pearl millet production. These findings are consistent with previous studies for sorghum using another crop model (Sultan *et al.*, 2014) and also for millet and sorghum (Sultan et al., 2013) and for maize (Schlenker and Lobell, 2010). All these findings confirmed that temperature increase is the main driver decreasing the yield for future climate change. Higher temperatures generally decrease the yield by increasing the plant growth rate, reducing thus the period available for biomass production (Chmielowski et al., 2004; McCarthy et al., 2012).

Furthermore, the result showed that millet grain yields respond positively i.e. an increase of grain yield under CMCC-CMS and CESM1-BGC scenarios than the three others. The increase in grain yield for those scenarios were observed under the Control and NPK treatments. This may be explained by the fact that in CMCC-CMS, the increase in temperature by 1.4° C and no change in rainfall may not affect the grain yields under Control and NPK treatments. Similarly, increase in temperature by 2.4° C and 10 % decrease in rainfall (CESM1-BGC scenario) may also not affect the grain yield. Millet yield decreased significantly under climate change scenarios, particularly
under NPK treatment and the ACCESS1-0 scenario, which is based on increase of both in temperature and precipitation (2.4° C and 30 %, respectively) for the period 2040 - 2069. As indicated in previous researches (McCarthy et al 2010; Eyshi Rezaie et al., 2013), this may be due to the high average temperature and the short phenological phase duration of pearl millet, especially in critical growth stages such as grain filling period under mineral fertilisation. The frequency of extreme events like warm day and warm night could therefore affect the productivity of millet in the Sudanian and Sahelian zones. In addition, excessive rainfalls (about 30 % in this study) can have severe impacts on millet production through the reduction of nutrient availability due to soil loss and fertilizer leaching.

6.4 Conclusion

ANSAP J

In conclusion, the study found that DSSAT model is a good crop modelling for selection of crops, soil and management options to suit the semi-arid and arid zone in Mali. The crop model showed good performance in reproducing the pearl millet yields to climatic variables under present treatment applications. It is able to capture the grain yield under different historical climate condition and also under different future climate conditions. Whereas in the future scenarios the simulation showed an excessive decreasing of grain yield. All the varieties showed lowest grain yields under the four treatments for ACCESS1-0 (Hot-Wet) among the scenarios

CHAPTER 7 CONCLUSIONS AND RECOMMENDATIONS

7.1 Introduction

This section of the study summarise the main findings of the study as presented in previous chapters and further develop necessary recommendations for farmers, policy makers and other Stakeholders.

7.2 conclusions

Based on the findings of this study, effect of climate change and variability on pearl millet (*Pennisetum glaucum* (l.) R. Br.) Production in the Sudanian and Sahelian agro-ecological zones in Mali, the study concluded that:

Farmers in Cinzana commune are conscious and aware of climate variability and change and see it as a real risk to their livelihood. A number of different strategies have been adapted by farmers in response to environmental degradation as a result of climate change impact. Perceptions of farmers of climate variability are in line with climatic data records. Indeed, farmers in Cinzana were able to recognize reduction of rainfall. The study found that the use of improved crop varieties was the main adaptation strategy among farm adaptation strategies in Cinzana because yield can be increased through improved crop management. Gender, size of land and age of household head were the significant determinants of adaptation strategies. The study analysed some factors likely to be influenced by climate variability and change in the study area. From the correlation analysis, it was revealed that factors such as household size, gender, education, age and farm size were statistically significant in relation to the adoption of strategies to mitigate climate change. The study analysed changes in precipitation and temperature extreme indices in Sotuba and Cinzana based on daily maximum and minimum temperature indices and precipitation series for the period from 1961 to 2014. The data were controlled and data series were homogenised using the software RClimdex 3.2.0. The finding of the study showed that there is a general decrease in the cold nights (TN10p) and cold days (TX10p) in Sotuba, while Segou showed high decreasing trend. A general warming trend in the temperature indices for warm nights (TN90p) and warm days (TX90p) since 1961 were observed. Although, there was a general tendency of decreases in annual total rainfall, the observed trends are not as uniform as the one in temperature. The observed trends have an implication on the economies of the two regions most of which are not strong enough to mitigate climate effects. Therefore, the warming trend observed means a higher demand on domestic energy consumption for cooling, higher evaporation rates from water bodies and irrigated crops, and a lower performance from livestock and agricultural crops. Furthermore, older persons are particularly vulnerable to such increase in temperature and may increase the level of mortality. Elsewhere, increased frequencies of extreme rainfall events such as long dry or wet spells mean weaker productive systems. Despite the small number of homogenous temperature and rainfall indices series, the study could represent a proportion of significant extremes in Ségou station average trends. The study provides evidence that during the last 53 years, Ségou was particularly affected by warm extremes based on night time indices rather than cold extremes based on day time indices.

The low productivity of millet crop in the Sahelian and Sudanian zones is caused by poor soil fertility due to uncertain rainfall conditions. A research trial addressed the issue to alleviate the agricultural production constraints with variable results depending on the agro-ecological environment and years. The results of the present study showed that millet production could be improved further by combining inorganic fertilizers with the placement of manure. The highest grain yield was obtained in straw yield and grain yield by OM+NPK combined application and the control as expected obtained the lowest yield. The output of this study clearly showed that module DSSAT simulated millet crop growth and yield with good accuracy. However, studies have indicated crop simulation models as potential decision making agronomic tools to understand

millet crop bio-dynamism under variable climatic conditions of agriculture. Moreover, the model evaluation over the two agro-ecologies enhances our knowledge to build resilience in various situations and to pick variability and rainfall regimes, prove a better tool for crop yield and rainfall forecast of the regions. The statistical parameters suggest that the overall DSSAT performance model in simulating chemical and organic fertilizer amendments in rainfed and irrigated maize is good and able to predict biomass and grain yields with high accuracy. The study showed that increase in temperature and precipitation in future would affect millet grain yield in climate.

7.3 Recommendations

This section outlines recommendations based on the outcomes of the study. The study acknowledges the extent of research on climate change that has been done across Africa, and a number of recommendations have therefore been made to policy makers to address climate change issues. There has been limited research on the subject of study in the selected study area and it's therefore important to provide the necessary recommendation to help in mitigating climate change and its impact in these areas.

It is therefore recommended that researchers, policy makers and the necessary institutions educate and provide information to farmers in order to enable them better understand climate change and its impact on their farming activities so as to adopt the necessary farming practices to mitigate these challenges. It is further recommended that researchers must design programmes that build on farmers' perceptions and plan climate awareness campaigns. Policy makers should also purpose at strengthening institutions such as meteorological services and agricultural extension so as to give real time information on climate change to farmers. Moreover, further studies must explore the role issues play in intensifying climate change impact and their subsequent effects on farmers. Policies relating to climate change are most often country or regional specific however, this study has demonstrated the need for a more tailored policy for specific geographical areas, it is therefore recommended that policy and decision makers must draw up policies for specific areas that have greater impact of climate change.

There is no doubt of global warming especially during day time and the frequency of night days' increase in temperature over the period 1961 – 2014 for the two agro-ecologies zones of Mali. In the current study, trends in extreme temperature and rainfall were established from historical climate data for two meteorological stations. Further research work could improve the value of the current finding by analysing climate data from more stations, especially selected from and representing the entire two natural regions in Mali. The findings could contribute to Geographical Information Systems (GIS) and provide information into the remapping and reformulation of farming recommendations in the natural regions. However, for agriculture, research for more heat resistant crops, animal species and cultivars should be promoted. Also the production systems need to be changed to reduce higher risks of crop losses due to climate change

Based on the simulation results from this study, the DSSAT crop model is suitable to simulate the impact of climate change on millet yield for the conditions in the two agro-ecological zones. However, the model performance in simulating on a long term basis needs to be further evaluated. From the point of view of information generation, DSSAT model will be able to explore climate change effect on plant and yield growth of another important cereal crop namely, pearl millet in future. This crop is presently better suited to the drier regions. The model can be used for management, planning and operation of irrigation projects for efficient use of water. The model can simulate and compare crop yield under different climate scenarios and cropping system, for instance, intercropping, rotation and sole cropping. Therefore, simulated crop yield for smallholder

farmers under climate change could be fed into economic models to estimate the economics of climate change implications for instance losses or/and gains. These subjects hold potential for further research.



APPENDIX

Questionnaires for evaluating farmers' perceptions of climate change and their adaptation strategies and evaluate change in land use, especially cropped areas and yields.

RegionDepartment:Commune:	
Village	
SECTION A GENERALE INFORMATION	
I- SOCIO-ECONOMIC CHARACTERISTICS OF HOUSEHOLDS	
1- Sexe : M	
2- Ethnie et nationalité/ Ethnic group/nationality	
Bamanan Senoufou Maninke Peulh Autres Preciser	•••••
Etrangers/ foreign	
3- Dans quelle tranche d'âge vous situez-vous?/ <i>Age</i>	1
18 à 35 ans/years old	5 ans et
4- Statut matrimonial/ Marital status	
Marié(e)/ married Célibataire/single Veuf/ widow	v
Monogame/ monogamous Divorce/ divorced Dolyg polygamous Nombre de femmes/ Number of wives	yne/
5- Confession religieuse/ Religion	
Chrétien/ Christian Musulman/ Muslim Polythéïste/ Polytheist Autres/ Others Précisez/ specify	/
6- Statut social du chef de ménage/ Household's head	
Chef de cours chef de famille chef de quartier Notable	
Chef religieux Chef de communauté Autres Précisez/	

7- Niveau d'instruction/ Education level

Illettré/ <i>illiterate</i>	Primaire/ primary level		Secondaire/	secondary	level
8- Revenu mensuel d	u chef de ménage/ <i>Househ</i>	old's income	-		
Moins de/ less than 50 De/between 100 000F between 200 000F à 29	000F De à 149 000F De/b 99 000F	e/ between 50 000 between 150 000f Plus de/mo)F à 99 000F à 199 000F ore than 300 000	De/ DF	
9- Avez-vous une acti a second activity? (vité secondaire ou dispose Or other sources of income	z-vous de revenu	is additionnels/	Do you have	
Oui/ yes	Non/ no 🗔				
Si oui, laquelle?/ if ye	s, name it	1.3			
A combien pouvez vo ?	us estimer ces revenus par	• mois/ What is yo	our income for	this activity	
10- Que faites-vous et	n dehors de cette activite /	What are the oth	er activities?	1	
Peche/ <i>fishing</i> Elevage/livestock	Artisanat/ <i>local craft</i> Autres/ <i>others</i>	Cueillette-chas	se/hunter-gathe	rer	
Pourquoi/ why ?					
11- Taille du ménage	 (nombre de p <mark>ersonnes ha</mark> l	bitant la maison)	/ size of the hou	isehold	
(number of people		ST		34	
Moins de/ <i>less than</i> 05 entre/ <i>between</i> 15 et 19	entre/ betwee more than 20 et pl	en 05 et 09 us	entre/ bei	ween 10 et 14	
II- CARACTERISTI CHARACTERISTICS	QU <mark>ES DES ACTIVITES</mark>	AGRICOLES/ F.	ARMING ACT	IVITY	
12- Quand avez-debu	te votre activite agricole /	When did you st	art farming ?		

Depuis au moins 30	ans/years	30 et 20 ans/years	20 et 10 ans/years	10
et 5 ans/years	moins de/ les	s than 5 ans/years		

13- Quelles sont les cultures que vous pratiquez / What crops do you cultivate ?
Riz/rice Mil/Millet Sorgho/ Surghum Coton/Cotton
Autres/others (préciser/Specify name them)
14- Combien de champs ou de parcelles possedez-vous/ How many farms do you have ?
15- Quelle est la superficie totale de votre champ/ What is the total area of your farmland ?
Moins de/ less than 1 ha de/ from 1 - 2 ha de/ from 3 - 5 ha de/
from 6 - 10 ha plus de/more than 10 ha
16- Depuis combien de temps cultivez_vous en continue ce champ? / Since when have you been cultivated this farm ?
17- Quelle est la rotation culturale que vous pratiquez dans votre exploitation? /What is the cultural rotation what you pratice in your farm ?
18- Est-ce que vous faites la jachère? / Do you do fallow land ?
19- Si oui quel est l'âge de la jachère ? / If yes what is the age ?
Moins de 05 entre 05 et 10 entre 11 et 20 entre 21 et 30 30 et plus
20- Parmi ces cultures, quelle est celle qui vous procure le plus de satisfaction/ Among your crops, which one give you more satisfaction?

21- Pourquoi/ why ?

.....

SECTION B: AGRICULTURAL PRODUCTION/ PRODUCTION AGRICOL

22. What are the priority crops grown now and 30 years ago?/ Quels sont les priorites

Crop grown cu	irrently	Crops grown	30 years ago	
Crop grow now/ culture cultive maintennt	How important is the crop for food security (see code below)/ Quelle importance à la culture pour la securite alimentaire (voir code dessous) par ordre d'importance	Crop grown five years ago/ culture cultive il ya 5ans	How important was the crop for food security (see codes below /Quelle importance a la culture pour la securite alimentaire (voir code dessous)/	If there is a change in priority of crop, why?/ S'il ya changement dans ces dites cultures, pourquoi?

Codes for importance of crop 1=Very importance 2=Moderate importance 3=Not important

23- Pendant combien de mois dans l'année votre exploitation agricole n'est pas autosuffisante ?/ *How many months in the year your farm is not self-sufficient*

24. What are the indicators of a good crop production year? *Quels sont les indicateurs d'une bonne annee de production*?

Indicator (e.g rainfall)/ Indicateur	Description/ Description

25.In the last 10 years, which years would you consider as having been good? Quel annee a ete bonne après 10 ans passes?

26. What are the indicators of a poor crop production year? Quels sont les indicateurs d'une mauvaise annee de production?

Indicator (e.g rainfall)/ Indicateur	Description/ Description

27. What are the average yields for the following major crops in a good crop production year and a poor crop production year? *Quel sont les rendements moyennes pour les cultures majeur d'une bonne et mauvaise annee de production*?

	Home field	10 E E	Out field	
Crop/Culture	Amount in kg/acre in a good crop production year/ <i>le montant</i> <i>de kg/hectare</i> <i>d'une bonne</i> <i>annee de</i> <i>production</i>	Amount in kg/acre in a bad crop production year/ le montant de kg/hectare d'une mauvaise annee de production	Amount in kg/acre in a good crop production year/ le montant de kg/hectare d'une bonne annee de production	Amount in kg/acre in a bad crop production year/ / le montant de kg/hectare d'une mauvaise annee de production
Maize/ Maise	1.1.1			
Millet/ Mil				
Sorghum/Sorgho			1	
Groundnuts/Arrachide				
Cowpea/ Haricot			1	
Bambar <mark>a nut/</mark>		12 8		
Other 1	1-3-Y		3	1
Other 2	- AN		X	

29. In the last 10 years, what has been the change in production of the following crops?/ Quel a ete les changements dans la production de ces cultures après 10ans passes?

How would you rank the changes you have mentioned above in terms of their contribution to change in agricultural productivity? *Comment notez yous le changement que yous avez mentionnez pour la contribution du changement de la productivite agricole?*

Cause of decline in crop production/ Cause de	Rank these factors starting with 1= most
la baisse dans la production des culture.	critical and 2= very most critical/ Notez ces
- A	facteurs commencant avec 1 = plus critical et
W JEAN	2= tres tres critical
SAI	AF 1

30. What are the main changes that you have made in the way you farm in the last ten years? /Quels sont les principaux changements que vous avez effectues dans la facon don't vous cultivez au cors des 10 dernieres annees.

a) Things that you are doing now that you were not doing before, varieties? Techniques culturales, engrais, herbicidage?/ Choses que vous faites maintenant que vous ne faisiez pas avant, les variétés? Techniques culturales, engrais, herbicidage?

Changes/ Changements	When did you make the	Why did you make the
	changes?/ Quand est ce vous	changes?\ Pourquoi avez vous
	avez fait les changements	fait les changements
	NIN	
	NUM	

b) Things that you were doing before and have now stopped, varieties? Techniques culturales, engrais, herbicidage? / *Choses que vous faites maintenant que vous n'avez arrte pas avant, les variétés ? Techniques culturales, engrais, herbicidage?*

Changes/ changements	When did you make the	Why did you make the
	changes? /Quand aviez vous	changes?/ Pourquoi vous avez
	les changements	fai les changements
		773
N		XX

SECTION C. HOUS<mark>EHOLD INCOME AND CAPITAL ASSETS/ REVE</mark>NU DES MÉNAGES ET IMMOBILISATIONS

31. What are your main sources of income in the past year and how important are these sources to your livelihood? Quelles sont vos principales sources de revenu de l'année écoulée et quelle sont l'importances de ces sources à votre vie?

Income Source/ Source de revnue	Yes/ No	Priority/ Priorite
Sale of crop/ vente de culture		
Sale of livestock/ vente de betail	K	1
Informal work(maricho)/ le travail informel	NO NO	3
(maricho)	2	
Formal employment/ le travail formel		
Remittances/ versement de fonds		
Old age pension/ la pension de vieillesse		

Pension fund from workgift received in kind/		
fonds de pension de recompense de travil recu		
No income at all/ aucun revenu		
Gardening/ gardinage		
Others (sepecify)/ autre (preciser)	1211 - 21 H	

Codes: Yes=1. No= 2/ Codes: Oui = 1, NO = 2

28. What livestock do you own?/Quels betails avez vous?

Assets/Biens	Do you	If yes, how	Source:/ Source	Purpose for keeping/But
	own?/	many?/ Si	1=bought,/achete	<i>de leur garder</i> 1=Mainly
	Avez vous?	oui,	2=gift,/cadeau	for
	1= yes/oui	combien?	3=inheritance,/heritage	food/essentielement pour
	2=no/non		4=other source/autre	nourriture
	1.1		source	2=Mainly for cash/
			17	essentiellement pour
				l'argent
		1000		3=Equally for cash
		1/0		and food/pour l'argent
		1		<i>et nourriture</i> 4=For
				prostige/Pour
		- 11		accumulation ou prestige
		1 62	8 37	accumulation ou prestige
Cattle/Boeuf	1	111-	1175	2
Goats/sheep/ chevre,				
mouton	1 - C		- Land	
Poultry/volaille			And	N
(chickens/poule,	-1/1	1		N
Guinea	- uu	A		
fowls/pindade,			11	2
donkey/ane)	1			
Pigs/ cochon				
Others				Z
(specify)/autre				21

29. What major agricultural assets/implements do you have? /*Quel outillage majeur vous avez*?

Asset/Bien	Do you	Number/Nombr	Source:/ Source
	own/Ave	е	1=bought/ achete,
	z, vous		2=gift/cadeau,

	1=		3=inheritance/heritag
	yes/oui		e, 4=other/autre
	2=no/no		Source
	n		
Ox-drawn plough/ Charrue de boeuf			
Harrow/ herse	N I	10	
Ridging plough/ pulverisatoire			
Oxcart/ char a boeuf			
Sprayer/ pulverisateur			
Cultivator/ cultivateur	6		
Hoes/ houe			
Irrigation equipment (e.g. treadle			
pump,	M	100	
water pump, etc)/ materiel d'irrigation			
(pompe a eau)	Sec. 20		
Other/autre(specify)	1	1.1	
Planter/ planteur		1	
Ripper/ eventeur	19		
Axe/ axe			
Other/ autre (specify)	17 M	14	
	13		TT

30. What major domestic assets do you have?/Quels sont les biens domestiques que vous possedez?

Assets/Biens	Do you own?/Avez yous?	If yes, how many?/ Si	Source:/source
	1= yes/oui	oui,	2=gift,/cadeau
	2=no/non	complen?	3=inheritance,/ <i>heritage</i>
	21	2/	source
Bicycle/Bicyclette			E
Sewing machines/ machine a			5
coudre			and I
Radio/TV/Radio/TV		6	8
Mobile phone/telephone portable		~	
Watch/clock/horloge	SANE	NO	-
Paraffin stove/ rechaud a petrole			
Other/ autre (specify)			

SECTION D: FARMER PERCEPTIONS OF CLIMATE CHANGE/ PERCEPTION DES PAYSANS SUR LE CHANGEMENT CLIMATIQUE

31. Are you aware of climate change? If yes/ *Etes-vous conscient du changement climatique? Si oui*

.....

- 31. Have you noticed any significant changes in weather patterns over the years in relation to agriculture? 0= No, 1= Yes/ Avez-vous remarqué des changements significatifs des conditions météorologiques au cours des années par rapport à l'agriculture? 0 = Non, 1 = Oui
- 32. For the changes mentioned above, what are some of their impacts in your household, the environment etc?
- 33. If any of these occurrences happens what actions do you take related to agricultural

production?

34. Do you have access to the weather forecasting data/information? 0=No, 1=Yes

Type of information	Source of information
	1=Radio,
	2=Extension
	3=Fellow farmer
	4=Television
	5=other (specify)
	TH I CON
	LAND

If yes, what different kinds of information do you get and where do you get it from?

35. How would you rate the weather information that you receive?

			and the second s
E	Rating	What are the	What are your
15	1=Poor,	reasons for your	suggestions for
Ab.	2=Average	rating?	improvement?
	3=Good	P.B.	
Timlineness	SANE	NON	
Adequacy			
Frequency of dissemination			
Usefulness			
General content			

Delivery channel		
Language of presentation		

36. If the forecast information is positive i.e it predicts that the rainfall will be enough and will be on time, what are some of the actions that you take in your farm?

	Ū.	Č .
Action	Do you take this action? (tick if farmer mentions)	Why do you take this action?
	KINLI	

37. If the forecast information is negative i.e it predicts that the rainfall will not be good or reliable, what are some of the actions that you take in your farm?

Action	Do you take this action? (tick if farmer mentions)	Why do you take this action?

38. Do you have any traditional / indigenous ways of predicting the weather patterns?

Weather pattern	Prediction Indicators
Drought Year	
Normal year (Rainfall)	
Flood Year	1 377
Very cold winters	A A A A A A A A A A A A A A A A A A A
Normal winters	March and
Very hot summer	" me
Normal summer	

39. What are the trends that you have observed in the following in the last ten years?

Variables	Increase	Same	Declined	What would you say is
Z	d (tick)	(tick)	(tick)	the main causes of this
121		1	- 8	change?
Crop yields		-	- /	244
Crop types, varieties			0	5
Crop pests and diseases		h	Br	
Livestock populations		2	1	
Livestock diseases	ANE	7		
Quality of pastures				
Rainfall amounts				
Water availability (for				
Soil erosion				

Wind erosion				
domestic use)				
Food availability for household				
consumption				
Water erosion	-	11121		
income from agriculture			CT	

40. For those variables where there has been a change, how are you coping with these changes?

Variables	How are you coping with change?
Crop yields	
Crop types, varieties	
Crop pests and diseases	
Livestock populations	
Livestock diseases	
Quality of pastures	
Rainfall amounts	
Water availability	
Soil erosion	
Water erosion	21-1
Wind erosion	8 7 5 5
Income from agriculture	1273
Food availability for household consumption	SAN S

41. Are you using any of the following farming practices in your farm as a result of the changes inn weather patterns?

Farming practice	Do you use? ((Tick as if	When do you use?
	farmer mentions)	1=All the time
		2=During drought years
Z		3=During good rainfall
13.		years
Potholing		
Ripping		5
Crop residues		B
Chemical weed control	04	5
Tied ridging	SANE NO	
Winter ploughing		
Conservation basins		
Using drought tolerant varieties		
Changing crops		

Mulching	
Intercropping	
Mono cropping	
Fallowing	
Other	

42. Are there some crop production practices that you use in good rainfall years and avoid in drought years? If yes, which ones

Cropping practice	Do you use in good rainfall years? 0=No 1=Yes	Do you use in drought years? 0=No 1=Yes	Reasons
Use of fertilizers	1-1,1	4	
Use of cattle manure	11	4	
Hire of labour for farming activities			
Use of irrigation		100	
Purchase of improved seeds	//9		
First weeding			
Second weeding		1	
Other specify			

REFERENCES

- ACCCA (Advancing Capacity to Support Climate Change Adaptation). (2010). Improving decision making capacity of small holder farmers in response to climate risk adaptation in three drought prone districts of Tigray. Paper presented at the Advancing Capacity to Support Climate Change Adaptation, Northern Ethiopia. Project No. 093.
- Ackerley, D., Booth B.B.B., Knight S.H.E., Highwood E.J., Frame D.J., Allen M.R. and Rowell D.P. (2011). Sensitivity of Twentieth-Century Sahel Rainfall to Sulphate Aerosol and CO2 Forcing. *Journal of Climate*, 24(19): 4999-5014.
- Acquah, H.D. (2011). Farmers' Perception and Adaptation to Climate Change Effects: A Willingness to Pay. *Journal of Sustainable. Development of Africa*, 13: 150–161.
- Adejuwon, S.A. (2004). Impacts of Climate Variability and Climate Change On Crop Yield in Nigeria. Paper Presented at the Stakeholders' Workshop On Assessment of Impacts and Adaptation to Climate Change (AIACC), Conference Centre, Obafemi Awolowo University, Ile-Ife, 271279.

- Adesina, A.A, and J.B. Forson. (1995). Farmers' Perceptions and Adoption of New Agricultural Technology: Evidence from Analysis in Burkina Faso and Guinea, West Africa. Agricultural Economics, 13:1–9.
- Adger, W.N., Huq, S., Brown, K., Conway, D. and Hulme, M. (2003). Adaptation to Climate Change in The Developing World. Progress in Development Studies, 3(3): 179-195.
- Aguilar, E., Auer, I., Brunet, M., Peterson, T.C. and Wieringa, J. (2003). Guidelines on Climate Metadata and Homogenization. World Meteorological Organization, Geneva, Switzerland 55 (WMO-TD No. 1186, WCDMP No. 53).
- Aguilar, E., Peterson, T. C., Obando, P. R., Frutos, R., Retana, J. A., Solera, M., Soley, J., García, I. G., Araujo, R. M., Santos, A. R., Valle, V. E., Brunet, M., Aguilar, L., Álvarez, L., Bautista, M., Castañón, C., Herrera, L., Ruano, E., Sinay, J. J., Sánchez, E., Oviedo, G. I. H., Obed, F., Salgado, J.E., Vázquez, J. L., Baca, M., Gutiérrez, M., Centella, C., Espinosa, J., Martínez, D., Olmedo, B., Espinoza, C. E. O., Núñez, R., Haylock, M., Benavides, H. and Mayorga, R. (2005). Changes in Precipitation and Temperature Extreme Sin Central America and Northern South America, 1961–2003. Journal of Geophysical Research, 110, D23107, 10.1029/2005 JD006119.
- Akponikpè, P.B. I., Peter J. and Agbossou, E. K. (2010). Farmers' Perception of Climate Change and Adaptation Strategies in Sub-Saharan West-Africa. 2nd International Conference: Climate, Sustainability and Development in Semi-Arid Regions August 16 - 20, 2010, Fortaleza -Ceará, Brazil.
- Alagarswamy, G., Reddy, D. M. and Swaminathan, D. (1998). Duration of The Photoperiod-Sensitive and Insensitive Phases of Time to Panicle Initiation in Sorghum. Field Crops Research 55: 110.
- Alexander, L.V., Zhang, X., Peterson, T.C., Caesar, J., Gleason, B., Klein Tank, A.M.G., Haylock, M., Collins, D., Trewin, B., Rahimzadeh, F., Tagipour, A., Rupa Kumar, K., Revadekar, J., Griffiths, G., Vincent, L., Stephenson, D.B., Burn, J., Aguilar, E., Brunet, M., Taylor, M., New, M., Zhai, P., Rusticucci, M. and Vazquez-Aguirre, J.L. (2006). Global observed changes
 - in daily climate extremes of temperature and precipitation. Journal of Geophysical Research, 111, D05109. http://dx.doi.org10.1029/2008JD011010.
- Alexander, H. (2013). Asian scientists mekong region facing six degrees warming, climate extremes. New York: USAID.
- Alhassane, A. (2009). Effets Du Climat Et Des Pratiques Culturales Sur La Croissance Et Le Développement Du Mil (Pennisetum Glaucum [L.]. R.BR.) Au Sahel : Contribution A L'amélioration Du Modèle SARRA-H De Prévision Des Rendements. UFR BiosciencesLaboratoire De Physiologie Végétale-Université De Cocody-Abidjan. 236.
- Alhassane, A., Traoré, S.B., Zouzou, M., Sarr, B. and Amadou, M. (2006). Effets De La Fertilisation Azotée Et De La Densité De Semis Sur La Consommation Hydrique D'une Variété De Mil Au Niger. Science Et Technique, Sciences Naturelles Et Agronomie, 28: 99-116.
- Ali, A. (2011). Climate Variability and Change In The Sahel. Understanding the Current Situation by Observing. Climate Change in The Sahel. A Challenge for Sustainable Development. AGRHYMET Monthly Bulletin (Special Issue), 17-20. (Http://Www.Agrhymet.Ne/PDF/Bulletin%20mensuel/Specialchcang.Pdf).

- Ali, M.H., Amin, M.G.M. and Islam, A.K.M.R. (2004). Comparison of various methods for estimating reference crop evapotranspiration. *Journal Bangladesh Agriculture University*, 2(2): 313324.
- Amadou, M. (1994). Analyse Et Modélisation De L'évaporation-Transpiration D'une Culture De Mil En Région Aride Sahélienne. Thèse De Doctorat. En Science De l'Université Paris XI ORSAI. N° D'ordre 3490 ; 106.
- Andrews, D. J. and Kumar, K. A. (2006). Pennisetum glaucum (l.) r.Br. Fiche de Protabase. Brink, m.
 & belay, g. (dir.). Prota (ressources végétales de l'afrique tropicale / ressources végétales de l'afrique tropicale). Wageningen, Pays-Bas
- Apata, T.G., Samuel, K. and Adeola, A. (2009). Analysis of Climate Change Perception and Adaptation Among Arable Food Crop Farmers in South Western Nigeria. Contributed Paper Prepared for Presentation at The International Association of Agricultural Economists, Beijing, China, August 16.
- Arrhenius, S. (1896). On the influence of carbonic acid in the air upon the temperature of the ground. *Philosophical Magazine and Journal of Science*, 41(251): 237-276.
- Arunah.L., Chiezey, U.F., Aliyu, L., Babaji, B.A., Odion, E.C. and Sani, B.M. (2006). Application of Inorganic Fertilizers and Poultry on Sorghum-crude Protein Yield and Growth Parameters. Proceeding of the 2nd National Conference on Organic Agriculture in Nigeria 114-117.
- Ati, O., Stigter, C. and Oladipo, E. (2002). A Comparison of Methods to Determine the Onset of the Growing Season in Northern Nigeria. *International Journal of Climatology*, 22(6): 731-742.
- Ayalew, M. (1997). What Are Food Security, Famine and Hunger? Accessed February 2008 Brad.Ac.Uk/Researchiijaslijasno2layalew.Html.
- Bacci, L., Cantini, C. Pierini, F. and Maracchi, G. (1998). Effects of Agronomic Practices On Growth, Development and Yield of a Short-Day Cultivar of Millet and their Agronomic Consequences in Mali. In : Le Futur Des Céréales Photopériodiques pour une Production Durable en Afrique Tropicale Semi-Aride (F.N. Reyniers And L. Bacci, Eds). Florence, CIRAD Et Ce. S.I.A.
- Balme, M., Vischel, T., Lebel, T., Peugeot, C. and Galle, S. (2006). Assessing The Water Balance in The Sahel: Impact of Small Scale Rainfall Variability On Runoff: Part 1: Rainfall Variability Analysis. *Journal of Hydrology*, 331(1–2): 336-348.
- Bandura, A. (1997). Self-Efficacy Towards a Unifying Theory of Behavioral Change. *Psychology Review*, 191-215.
- Baptista, S., Brottem, L. and De Sherbinin, A. (2013). *Background Paper On Regional Climate Change Vulnerability Assessment of West Africa*. ARCC.
- Bardege, B., Neufeldt, H., Mowo, J., Abdelkadir, A., Muriuki, J., Delle, G., Tewodros, A. Barry, S., Burton I., Richard, J.T. and Klein, W. (2013). An Atomy of Adaptation to Climate Change and Vulnerability, Springs.
- Barrios, S., Ouattara, B. and Strobl, E. (2008). The impact of climatic change on agricultural production: Is it different for Africa? *Food Policy*, 33(4): 287-298.
- Basavaraj, G., Rao, P. P., Bhagavatula, S. and Ahmed W. (2010). Availability and Utilization of Pearl Millet in India. *Journal of SAT Agricultural Research*, 8.
- Bationo, A., Christianson, C. B. and Baethgen, W. E. (1990). Plant Density and Nitrogen Fertilizer Effects on Pearl Millet Production in Niger. *Agronomy Journal*, 82: 290-295.

- Bationo, A. and Buerkert, A. (2001). Soil Organic Carbon Management for Sustainable Land Use in Sudano-Sahelian West Africa. *Nutrient Cycling in Agroecosystems*, 61(1-2): 131-142.
- Begg, J.E. (1965). The Growth and Development of a Crop of Bulrush Millet (Pennisetum Typhoides S. & H.). *Journal of Agriculture Science*, 65:341–349.
- Begg, J.E. and Burton, G.W. (1971). Comparative Study of Five Genotypes of Pearl Millet Under a Range of Photoperiods and Temperatures. *Crop Science*, 11:803–805.
- Bégué, A. (1991) Estimation De La Production Primaire En Zone Sahélienne A Partir De Données Radiométriques. Cas D'un Couvert Discontinu : Le Mil. Phd. Université De Paris VII.
- Ben Mohamed, A., Duivenbooden, V. N. and Abdoussallam, S. (2002). Impact of Climate Change On Agricultural Production in The Sahel-Part 1. Methodological Approach and Case Study for Millet in Niger. *Climatic Change*, 54: 327-348.
- Berke F., Colding, J. and Folke, C. (2000). Rediscovery of Traditional Ecological Knowledge as Adaptive Management. *Ecological Applications* 10:1251-1262.
- Biasutti, M. (2013) Forced Sahel Rainfall Trends in The CMIP5 Archive. *Journal of Geophysical Research D: Atmospheres*, 118(4), 1613-1623.
- Bley, J. (1990). Experimentelle und Model Analytische Untersuchungen Zunwasser- Und Nahrst of Fhaushalt von Perlhirse (Pennisetumamericanum[L.]) In Sudwest- Niger. Dissertation. Universitat Hohenheim. 132.
- Boko, M., Niang, I., Nyong, A., Vogel, C., Githeko, A., Medany, M., Osman-Elasha, B., Tabo, R. and Yanda, P. (2007). Africa Climate Change 2007: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. In: M. L., Parry, O. F., Canziani, J. P., Palutikof, P.J., van der Linden and Hanson, C.E., (eds.), Cambridge, UK. Cambridge University Press, pp. 433-467.

Braimoh, A. K. (2004). Modeling land-use change in the Volta Basin of Ghana. Ph.D.

- Breman, H. and Sissoko, K. (1998). *L'intensification Agricole au Sahel. Paris*, Karthala Editions Pp 996.
- Boote, K.J., Jones J.W., Mishoe, W. and Wilkerson, G.G. (1986). Modeling growth and yield of groundnut. In: Agrometeorology of Groundnut. Proceedings of an International Symposium, Niamey, Niger, 21–26 August 1985. Patencheru, ICRISAT.
- Boubacar, B. I. (1985). Suivi De La Croissance Et Du Développement D'une Gamme de Variétés De Mil (Pennisetum Typhoïdes) Dans Les Conditions Pluviométriques de l'Ouest Nigérien. Rapport de Stage, ICRISAT / Niamey Juin 1985. 46.
- Bourdet, Y. (2011). HIMO Et Lutte Contre La Pauvreté Au Mali.
- Butt T.A., Mccarl B.A., Angerer J., Dyke P.T., and Stuth J.W. (2005). The Economic and Food Security Implications of Climate Change in Mali. *Climatic Change* 68:355-378.
- Butt T.A. and Mccarl, K. (2006). Policies for Reducing Agricultural Sector Vulnerability to Climate Change in Mali, Published in *Climate Policy*, 5: 583-598.
- Caesar, J., Alexander, L. and Vose, R. (2006). Large-Scale Changes in Observed Daily Maximum and Minimum Temperatures: Creation and Analysis of a Gridded Data Set. *Journal of Geophysical Research* 111, DO5101, 10. 1029/2005 JD006280
- Callendar, G.S. (1938). The artificial production of carbon dioxide and its influence on temperature. Q. J. R. *Meteorology Social*, 64: 223–237.

thesis, University of Bonn, 166 pp.

- CEDEAO- Club Sahel/ OCDE /CILSS. (2008). *Climate and Climate Change*. The Tlason Regional Integration in West Africa. Environment Series, P. 13.
- Calzadilla, A., Zhu, T., Rehdanz, K., Tol, R. S. J. and Ringler, C. (2009). *Economy-Wide Impacts of Climate Change On Agriculture in Sub- Saharan Africa*, FNU-170.
- Cantini, C. (1995) Manuel D'exercices sur le Developpement, la Croissance et la Production des Plantes. Ecophysiologie. Manuel Technique No. 13-1995. Florence, Ce.
- Challinor, A., Wheeler, T., Garforth, C., Craufurd, P. and Kassam, A. (2007). Assessing the vulnerability of food crop systems in Africa to climate change. *Climatic Change*, 83(3): 381399.
- Chiotti, Q. P. and Johnson, T. (1995). Extending The Boundaries of Climate Change Research: A Discussion On Agriculture *Journal of Rural Studies* 11 (3): 335-350.
- Chipo, P. M. (2010). These: Farmer Strategies Towards Climate Variability and Change in Zimbabwe and Zambia 11-59 (2010-11-21).
- Chitalu. G. (2013). Genetic Characterization of Pearl Millet (Pennisetum Glaucum (L.) R. Br.) Genotypes in Zambia P.8.
- Chuku, C.A. and Okoye, C. (2009). Increasing resilience reducing vulnerability in sub-Saharan African agriculture: strategies for risk coping and management. *African Journal of Agricultural Research* 4: 1524–1535.
- Cleaver, K.M. and Schreiber, G.A. (1994). Supplement to Reversing the Spiral: The Population, Agriculture, And Environment Nexus in Sub-Saharan Africa. Washington DC : World Bank. Chopart,

J. L. and Kone, D. (1994). Fluctuation de L'alimentation Hydrique du Maïs en Région Centre. In Eds.

F. N. REYNIERS et L. NETOYO, Bilan Hydrique Agricole et Sécheresse 226 en Afrique Tropicale.

Vers une Gestion des Flux Hydriques par les Systèmes de Culture ? Actes Sem. Int. Bamako, Mali, Décembre 1991. J. LIBBEY, Paris, 39-47.

- Chmielewski, F.M., Muller, A. and Bruns, E. (2004). Climate changes and trends in phenology of fruit trees and field crops in Germany, 1961–2000. *Agricultural and Forest Meteorology*, 121: 6978.
- Clark, W., Mitchell, R., Cash, D. and Alcock, F. (2002). Information as Influence: How Institutions Mediate the Impacts of Scientific Assessments On Global Environmental Affairs. Faculty Research Working Papers Series, RWP02-044. John F. Kennedy School of Government, Harvard University.
- Clerget B. and Traore, S. (2009). *Project Millet and Sorghum Pearl millet and sorghum yield potential in West Africa.* PROMISO
- Clement J.C., Bezançon G. and Billard., G. (1993). Prospections Des Mils Cultivés Et Des Mils Sauvages de l'Afrique De l'Ouest. In Le Mil en Afrique, S. Hamon Ed., Paris, France, ORSTOM, 9-20.
- CGIAR. (2006). (Consultative Group on International Agricultural Research). *Pearl Millet: A Hardy Staple for the World's Drylands*, CGIAR: Story of The Month – June 2006. <u>Http://Www.Cgiar.Org/Monthlystory/June2006.Html</u>.
- Cooper, P.J.M., Dimes, J., Rao, K.P.C., Shapiro, B., Shiferaw, B. and Twomlow, S. (2008). Coping Better with Current Climatic Variability in The Rain-Fed Farming Systems of Sub-Saharan Africa: An Essential First Step in Adapting to Future Climate Change? Agriculture, Ecosystems and Environment, 126: 24-35.

- Core Team. (2012). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. ISBN: 3-900051-07-0.
- Coulibaly, H. (2010). Rôle des organisations paysannes dans la diffusion de semences de céréales : articulation des réseaux semenciers étatique et traditionnels paysans pour une conservation in situ des variétés. Le cas des mils et sorghos au Mali. Thèse de Doctorat en géographie humaine, économique et régionale de l'université de paris ouest Nanterre la défense école doctorale milieux, cultures et sociétés du passe et du présent composante espace et société thèse de doctorat. 295pp.
- Daberkow, S.G. and Mcbride, W.D. (2003). Farm and Operator Characteristics Affecting the Awareness and Adoption of Precision Agriculture Technologies in the U.S. *Precision Agriculture*, 4:163–177.
- Dai, A. and Trenberth, K.E. (2004). The Diurnal Cycle and its Depiction in the Community Climate System Model. *Journal of Climate* 17(5): 930-951.
- Dell, M., Jones, B.F. and Olken, B.A. (2012). Temperature Shocks and Economic Growth: Evidence from the Last Half Century. *American Economic Journal*: 4: 66–95.
- De Graaff, J., Kessler, A. and Nibbering, J.W. (2011). Agriculture and food security in selected countries in Sub-Saharan Africa: diversity in trends and opportunities. *Food Security* 3(2): 195-213.
- De Groote, H. and Coulibaly, N. (1998). Gender and Generation: An Intra-Household Analysis on Access to Resources in Southern Mali. *African Crop Science Journal* 6(1): 79–95.
- Dendy, D. A. V. (1995). In 'Sorghum and the Millets: Chemistry and Technology'. (D.A.V. Dendy, ed.), American Association of Cereal Chemists, St. Paul, USA Pp. 11-26.
- Denis, J. J. (1984). Manuel des Principales Cultures Sahéliennes ; Centre Régional De Formation et D'application en Agro Météorologie et Hydrologique Opérationnelle. (Agrhymet). Tome 1; N° 173; 133p.
- Deressa, T. T., Hassan, R. M., Ringler, C., Alemu T. and. Yesuf, M. (2009). Determinants of Farmers' Choice of Adaptation Methods to Climate Change in the Nile Basin of Ethiopia. *Global Environmental Change* 19:248-255.
- De Wit, M. (2006). *Climate Change and African Agriculture*. Accessed On 23 March. (2009). <u>Http://Www.Ceepa.Co.Za/Docs/POLICY%20NOTE%2011.Pdf.</u>
- Diarra, B., Diakite, H. and Macina, M. (2007). Elaboration de Strategie d'adaption des Ressources en Eau aux Changements Climatiques. P: Energy 2-Common Projects Ongoing 032135 NCAP2 from January 2007 NCAP Countries Activities Outputs Countries Mali 032135. 060821. MAL. Phase1. Output3. V1. Doc.
- Dilley, M. and Heyman, B. (1995). 'ENSO and Disaster: Droughts, Floods, and El Niño/Southern Oscillation Warm Events'. *Journal Disasters*, 19: 181–193.
- Dinar A., Hassan, R., Mendelsohn, R. and Benhin, J. (2008). *Climate Change and Agriculture in Africa: Impact Assessment and Adaptation Strategies*. Earth scan, London. Pp. 100-106.
- Diop, M. (1996). A Propos de da Durée de da Saison des Pluies au Sénégal. Sécheresse, 7:7-15.
- Diop, M. (1999). Caractérisation du Facteur Hydrique en Agriculture Pluviale au Sénégal : le cas de L'arachide et du Mil. Thèse Doc. De Géographie, Spécialité : Climatologie. Université De Bourgogne, UNRS UPRESA « Climatologie De L'espace Tropical. Pp 185.

- Diouf, M., Nonguierma, A., Amani, A. and Royer, A. (2000). Lutte Contre la Sécheresse au Sahel : Résultats, Acquis et Perspectives au Centre Régional AGRHYMET. Sécheresse, 11 (4), 57– 266.
- Diouf, O. (2000). Réponses Agro Physiologiques Du Mil (Pennisetum glaucum (L.) R. Br.) A la Sécheresse : Influence de la Nutrition Azotée. Thèse Doc. En Sciences ; Laboratoire de Physiologie Végétale Et D'agro Technologies Végétales, Faculté des Sciences de l'Université Libre De Bruxelles. Pp160.
- Djiré, M., Keita, A. and Diawara, A. (2013). Agricultural Investments and Land Acquisitions in MaliContext, Trends and Case Studies. IIED/GERSDA, London/ Bamako
- DNS. (Enquête Agricole De Conjoncture). (1995). 1994-1995 Résultats Définitifs, Bamako. DNA, Août 1995, 52p.
- Donald, C. M., and Hamblin, J. (1976). The Biological Yield and Harvest Index of Cereals as Agronomic and Plant Breeding Criteria. *Advance in Agronomy* 28: 361-405
- Dodd, D.E.S. and Jolliffe, I.T. (2001). Early Detection of the Start of the Wet Season in Semiarid Tropical Climates of Western Africa. *International Journal of Climatology*, 21: 1251-1262.
- Drechsel, P., Gyiele, L., Kunze, D. and Cofie, O. (2001). Population Density, Soil Nutrient Depletion, and Economic Growth in Sub-Saharan Africa. *Ecological Economics*, 38(2): 251-258.
- Dust M., Baran, N., Errera, G., Huston, J.L., Mouvet, C., Schafet, H., Vereecken, H. and Walker, A. (2000). Simulation of water and solute transport in field soils with the LEACHP model. *Agriculture Water Manage*, 44: 225-245.
- Easterling, D. R. (1997). Maximum and Minimum Temperature Trends. *Science* 277, 364–367, Http://Dx.Doi.Org/10.1126/Science.277.5324.364. (Www.Sciencemag. Org/Cgi/Doi/10.1126/Science.277.5324.364).
- Easterling, D. R., Alexander, L. V., Mokssit, A. and Detemmerman, V. (2003). CCI/ CLIVAR Workshop to Develop Priority Climate Indices. *Bulletin of The American Meteorological Society*, 8: 1403–1407.
- Ehui, S. and Pender, J. (2005). Resource Degradation, Low Agricultural Productivity, and Poverty in Sub-Saharan Africa: Pathways out of the Spiral. *Agricultural Economics*, 32(1): 225-242.
- Ekwe A., Festus O. A., Teddy A. M., Emmanuel, T. W. and Leroy W. C. (2011). Agricultural Innovations for Climate Change Adaptation and Food Security in West Africa: The case of Nigeria, Sierra Leone and Liberia. P. 15-17.
- Evans L. T. and Fischer R. A. (1999). Yield Potential: Its Definition, Measurement, and Significance. *Crop Science*, 39 (6) 1544-1551.
- Eyshi Rezaie, E., Gaiser, T., Siebert, S. and Ewert, F. (2013). Adaptation of Crop Production to Climate Change by crop Substitution. *Mitigation and Adaptation Strategies for Global Change*, 013:9528-1.
- Fana, G. and Asnake, K. (2012). *Traditional Coping Mechanisms for Climate Change of Pastoralists* in South Omo, Ethiopia, Adis Ababa. University, Ethiopia
- FAO. (2005). Impact of Climate Change, Pests and Diseases on Food Security and Poverty Reduction. Special Event Background Document for the 31st Session of the Committee on World Food Security. FAO Publication, Rome. Pp. 1-10.

- FAO. (2008). Water and The Rural Poor: Interventions for Improving Livelihoods in Sub-Saharan Africa. J. M. Faurès, G. Santini (Eds.) FAO Land and Water Division Rome, FAO.
- FAO. (2012). Twenty-Seventh FAO Regional Conference for Africa. ARC/12/REP Rev.1, Brazzaville Congo 23-27 April 2012.
- FAO and ICRISAT. (1997). L'économie Mondiale du Sorgho et du Mil : Faits, Tendances et Perspectives. ISBN 92-5-203861-2. Pp.66
- FAOSTAT. (2008). Food and Agriculture Organization of the United Nations, FAOSTAT database. available at http://faostat.fao.org/site/362/DesktopDefault.aspx?PageID=362.
- Felch D. and Traore, S. (2010). Agrometeorology of Some Selected Crops, Chapter 10 P 36-45.
- Fischer, G., Frohberg, K., Parry, M. L., and Rosenzweig, C. (1994). Climate Change and World Food Supply, Demand and Trade: Who Benefits, Who Loses? *Global Environmental Change*, 4 (1): 7-23.
- Fisher G, Shah M. and Van Velthuizen H. (2002). *Climate Change and Agricultural Vulnerability*. International Institute for Applied Systems Analysis. Laxenburg, Austria.
- Fisher, M., Chaudhury, M. and Mccusker, B. (2010). Do Forests Help Rural Households Adapt to Climate Variability? Evidence from Southern Malawi. *World Development*, 38:1241-1250.
- Fleming, J. R. (1998). Historical Perspectives on Climate Change. New York. Oxfor University Press.
- Frappart F., Hiernaux P., Guichard F., Mougin E. and Kergoat L. (2009). Rainfall Regime Across the Sahel Band in the Gourma Region, Mali. *Journal of Hydrology*, 375: 128–142.
- Funk, C, Dettinger, M.D., Michaelsen, J.C., Verdin, J.P., Brown, M.E., Barlow M. and Hoell A. (2008).
 Warming of the Indian Ocean Threatens Eastern and Southern African Food Security but Could be Mitigated by Agricultural Development. Proceedings of the *National Academy of Sciences*, 105(32): 11081-11086.
- Fussell, L.K., Pearson, C. J. and Norman M.J.T. (1980). Effect of Temperature During Various Growth Stages On Grain Development and Yield of Pennisetum Americium. *Journal of Experimental Botany*, 31:621–633.
- Garcia, R. V. and Escudero, J. C. (1981). Drought and Man: The Constant Catastrophe. Volume 2. New York. Pergamon Press.
- Gbetibouo G. A. (2009). Understanding Farmers' Perceptions and Adaptations to Climate Change and Variability the Case of the Limpopo Basin, South Africa. Environment and Production Technology Division. IFPRI Discussion Paper 00849. 52 P 19.
- Giannini, A., Biasutti, M., Held, I. M. and Sobel., A.H. (2008). A Global Perspective on African Climate. *Climatic Change*, 90: 359–383.
- Giller, K.E., Tittonell, P., Rufino, M.C., Van Wijk, M.T., Zingore, S., Mapfumo, P., Adjei-Nsiah, S., Herrero, M., Chikowo, R., Corbeels, M., Rowe, E.C., Baijukya, F., Mwijage, A., Smith, J., Yeboah, E., Van Der Burg, W.J., Sanogo, O.M., Misiko, M., De Ridder, N., Karanja, S., Kaizzi, C., K'ungu, J., Mwale, M., Nwaga, D., Pacini, C. and Vanlauwe, B. (2011). Communicating Complexity: Integrated Assessment of Trade-Offs Concerning Soil Fertility Management Within African Farming Systems to Support Innovation and Development. *Agricultural Systems*, 104(2): 191-203.

- Glantz, M. H. (1995). Assessing the Impacts of Climate: The Issue of Winners and Losers in A Global Climate Change Context. In: Zwerver, S., Van Rompaey, Kok, R.S., Berk, M.M. (Eds.) Climate Change Research: Evaluation and Policy Implications. Amsterdam. Elsevier Science.
- Haarsma, R.J., Selten F.M., Weber S.L. and Kliphuis, M. (2005). Sahel Rainfall Variability and Response to Greenhouse Warming. *Geophysical Research Letters*, 32(17).
- Hassan R. and Nkemechena C. (2008). Determinants of African farmers' strategies for adapting to climate changes: multinomial choice analysis. *AFJARE*, 2(1): 85-104.
- Hegerl, G.C., Zwiers, F.W., Braconnot, P., Gillett, N.P., Luo, Y., Marengo Orsini, J.A., Nicholls, N., Penner, J.E. and Stott, P.A. (2007). *Understanding and Attributing Climate Change*. In: Solomon, S., Qin, D., Manning, M., Chen, Z., Marquis, M., Averyt, K.B., Tignor, M., Miller, H.L. (Eds.), Climate Change: The Physical Science Basis. Contribution of Working Group I to The Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York.
- Hoogenboom, G., Jones, J.W., Wilkens, P.W., Batchelor, W.D., Bowen, W.T., Hunt, L.A., Eckering, N.B., Singh, U., Godwin, D.C., Baer, B., Boote, K.J., Ritchie, J.T. and White, J.W. (1994). *Crop models*. In: Tsuji, G.Y., Uehara, G., Balas, S. (Eds.), DSSAT Version 3, vol. 2. University of Hawaii, Honolulu, HI, 95_/244.
- Hoogenboom, G., Wilkens, P.W.and Tsuji, G. Y. (Eds.) 1999. DSSAT version 3 (4). International Consortium for Agricultural Systems application. University of Hawaii Honolulu, HI, 234– 235
- Hoogenboom, C.H., Jones, J.W., Porter, G., Boote, K.J., Batchelor, W.D, Hunt, L.A., Wilkens., P.W., Singh, U., Gijsman, A.J. and Ritchie, J.T. (2003). The DSSAT Cropping System Model. *European Journal of Agronomy*, 18: 235-265.
- Hoogenboom, G., Jones, J.W., Porter, C.H., Wilkens, P.W., Boote K.J., Hunt, L.A. and Tsuji, G.Y. (Eds). (2010). *Decision Support System for Agrotechnology Transfer Version* 4.5. Volume 1: Overview. University of Hawaii, Honolulu, Hi. WILKENS, P. W., G. HOOGENBOOM, C.H. PORTER.
- Houghton, J. T., Jenkins, G. J. and Ephraums, J. J., (Eds.) (1990). *Climate Change: The IPCC Scientific Assessments*. New York. Cambridge University Press.
- Hountondji, Y.C., De Longueville F. and Ozer P. (2011). *Trends in Extreme Rainfall Events in Benin* (*West Africa*), 1960-2000. Proceedings of The 1st International Conference On Energy, Environment and Climate Change, August 26-27, 2011, Ho Chi Minh City, Vietnam. 7 P. Http://Hdl.Handle.Net/2268/96112.
- Hulme, M. (1996). Climate Change in Southern Africa: An Exploration of Some Potential Impacts and Implications in the SADC Region. United Kingdom. Climatic Research Unit, University of East Anglia, Norwich, 96 pp.
- Hulme, M., and Sheard, N. (1999). Climate Change Scenarios for the Northern Andes Climate Research Unit. Norwich. University of East Anglia.
- Hulme, M., Doherty, R., Ngara, T., New, M. and Lister, D. (2001). African Climate Change: 19002100. *Climate Research*, 17(2): 145-168.
- IFAD. (2011). Enabling The Rural Poor to Overcome Poverty in Mali, Project Fact Sheet.
- IFPRI (International Food Policy Research Institute). (2001). Empowering Women to Achieve Food Security, Washington, DC.

IMF. (2006). Mali Country Report 06/310. IMF

- IPCC. (2001). Climate Change Impacts, Adaptation, And Vulnerability. Contribution of Working Group II to The Third Assessment Report of the Intergovernmental Panel On Climate Change, Mccarthy, J. J., Canziani, O. F, Leary, N. A., Dokken, D. J. and White, K.S., (Eds.) Cambridge. Cambridge University Press.
- IPCC. Working Group 2. (2001). Third Assessment Report, Annex B: Glossary of Terms. Intergovernmental Panel On Climate Change (IPCC) (2007), Climate Change 2007, Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, Annex I., M.L.
- IPCC. (2007). "Summary for Policy Makers", In M. Parry, Ocanziani, & P. Vander Linden (Eds) Climate Change 2007: Impacts Adaption and Vulnerability Contribution of Working Group II for The Fourth Assessment Report of the Intergovernmental Panel On Climate Change, Cambridge University Press.

IPCC. (2007a). *Climate Change 2007: Mitigation*. Contribution of Working Group III to The Fourth Assessment Report of the Intergovernmental Panel On Climate Change, Metz, B., Davidson, O., Bosch, P., Dave, R., And Meyer, L., (Eds.) Cambridge, UK. Cambridge University Press. IPCC. (2013). Summary for Policymakers. In: T.F. Stocker Et Al. (Editors), Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to The Fifth Assessment Report

- of the Intergovernmental Panel On Climate Change Cambridge University Press, Cambridge, United Kingdom And New York, NY, USA.
- Jalloh, A., Nelson, G.C., Thomas, T.S., Zoubmoré, R. and Roy-Macauley, H. (2013). West African Agriculture and Climate Change: A Comprehensive Analysis. Research Monograph. ARD Food Policy Research Institute. Washington: IFPRI.
- Jarvis, A., Lau, C., Cook, S., Wollenberg, E., Hansen, J., Bonilla, O. and Challinor, A. (2011). An integrated adaptation and mitigation framework for developing agricultural research: Synergies and Trade-offs. *Experimental Agriculture*, 47(2): 185 – 203.
- Jones, C. A. and Kiniry J. R. (1986). *CERES-Maize: A Simulation Model of Maize Growth and Development*. Texas A & M University Press, College Station, Texas.
- Jones, J.W., G. Hoogenboom, C.H. Porter, K.J. Boote, W.D. Batchelor, L.A. Hunt, P.W. Wilkens, U. Singh, Gijsman A.J. and Ritchie J.T. (2003). The DSSAT Cropping System Model. *European Journal of Agronomy*, 18: 235-265.
- Kablan R., Yost R. S., Brannan K., Doumbia M. D., Traoré K., Yoroté A., Toloba Y., Sissoko S., Samaké O., Vaksman M., Dioni L. and Sissoko M. (2008). "Aménagement en courbes de niveau." Increasing Rainfall Capture, Storage, and Drainage in Soils of Mali. Arid Land Research and Management, 22:1, 62-80.
- Kadomura, H. (1994). Climatic changes, droughts, desertification and land degradation in the Sudano-Sahelian Region—a historico-geographical perspective. In Kadomura, H. (ed.) Savannization Processes in Tropical Africa II Tokyo. Tokyo Metropolitan University, pp. 203–228.
- Karl, T.R. and Easterling D.R. (1999). Climate extremes: selected review and future research directions. *Climatic Change*, 42: 309 325.
- Karl T.R. and Trenberth, K.E. (2003). "Modern Global Climate Change". *Science* 302 (5651) : 1719–23.

- Kanno, H., Caldwell, J. S., Sasaki, K., Yorote, A., Berthe, A., Doumbia, M., Ozawa K. and T. Sakurai. (2008). Seasonal Meteorological Variation in Southern Mali, West Africa. Part1. Surface Meteorological Observation from 2001 To 2004. *Geographical Reports Tokyo Metropolitan* University, 43:1-10.
- Kassam, A. H. and Kowal, J. M. (1975). Water Use, Energy Balance and Growth of GeronMillet at Samaru, Northern Nigeria. Agricultural Meteorology, 15:333-342.
- Kates, R.W. (2000). Cautionary tales: Adaptation and the global poor. *Climatic Change*, 45: 5–17.
- Kemausuor, F., Dwamena, E., Bart-Plange, A. and Kyei-Baffour N. (2011). Farmers' Perception of Climate Change in The Ejura-Sekyedumase District of Ghana. ARPN *Journal of Agricultural* and Biological Science, 6:10-11.
- Kleijnen, J.P.C (1995). Verification and Validation of Simulation Models. *European Journal of Operational research*, 82 (1), 145-16.
- Koch, I.C, Vogel, C. and Zarina P. (2006). Institutional Dynamics and Climate Change Adaptation in South Africa. *Springer Science+Business Media*, 12 : 1323–1339.
- Konate, A. (2010). *Nouvel Horizon, Changement climatique au Mali*. Les prévisions des spécialistes en la matière. http://temoust.org/changement-climatique-au-mali-les,13320
- Kouressy, M., Niangado O., Vaksman M. and Reyniers F.N. (1998). Etude De La Variabilité Phénologique Du Mil Du Mali Et De Son Utilisation Pour L'amélioriation Variétale. In : Le Futur Des Céréales Photopériodiques Pour Une Production Durable En Afrique Tropicale Semi-Aride (F.N. Reyniers And L. Bacci, Eds). Florence, CIRAD Et Ce. S.I.A.
- Kousik, C.S., Thakur R.P. and Subbarao, K.V. (1988). Influence of Environmental Factors On Production and Dispersal of Tolyposporium Penicillariae Sporidia. Ind. Journal of Aerobiology, 1:85–91.
- Kouyaté Z., Krasova-Wade T., Yattara I.I. and Neyra M. (2014). Effects of cropping system and Cowpea variety (Vigna unguiculata L. Walp) on the diversity of native cowpea bradyrhizobia and millet yield in the Sudano Sahelian zone of Mali. Int. Res. *Journal of Agriculture Science and Soil Science*, 4 (2): 30-39
- Kristie, L., Ebi N. D. L. and Carlos F. C. (2005). Climate Variability and Change and Their Health Effects in Small Island States: Information for Adaptation Planning in The Health Sector. World Health Organization, Geneva, Switzerland, 47 Pp.
- Kruger A.C. and Shongwe, S. (2004). Temperature trends in South Africa: 1960-2003. *International Journal of Climatology*, 24 (15): 19-29.
- Kruger, A.C. (2006). Observed Trends in Daily Precipitation Indices in South Africa: 1910 2004. International Journal of Climatology, 26: 2275–2285.
- Kruger, A.C. and Sekele, S. S. (2013). Trends in Extreme Temperature Indices in South Africa: 1962–2009. *International Journal of Climatology*, 33: 661–676.
- King'uyu, S.M., Ogallo, L.A. and Anyamba, E.K. (2000). Recent trends of minimum and maximum surface temperatures over Eastern Africa. *Journal of Climate*, 13: 2876-2886.
- Knox, J.W., T.M. Hess, Daccache A. and Perez Ortola M. (2011). What are the projected impacts of climate change on crop productivity in Africa and South Asia? DFID Systematic Review Final Report. Cranfield University, 77 p.

- Kyekyeku, N.F. (2012). Climate Change Vulnerability and Coping Mechanisms Among Farming Communities in Ghana, Khansas State University, Manhattan, Kansas.
- Lamb, P. J. and Peppler R. A. (1982). «Further Case Studies of Tropical Atlantic Surface Atmospheric and Oceanic Patterns Associated with Sub-Saharan Drought». *Journal of Climate*, 5: 476-488.
- Lamptey, B. (2009). Ananalytical Framework for Estimating Theurban Effect On Climate. *International Journal of Climatology*, 30: 72–88.
- Lavinia, B. (2011). Investment Strategies for Climate Change Mitigation.
- Lebel, T. and Ali A. (2009). Recent Trends in the Central and Western Sahel Rainfall Regime (19902007). *Journal of Hydrology*, 375(1-2): 52-64.
- Le Barbé, L., Lebel, T. and Tapsoba, D. (2002). Rainfall Variability in West Africa During the Years 1950–90. *Journal of Climate*, 15(2): 187-202.
- Lecina, S., Martinez-Cob, A., Perez, P.J., Villalobos, F.J. and Baselga, J.J. (2003). Fixed versus variable bulk canopy resistance for reference ET estimation using the Penman-Monteith equation under semi-arid conditions. *Agricuture Water Manage*, 60: 181–198.
- Leisinger, K.M. and Schmitt, K. (1995). Survival in the Sahel: An Ecological and Developmental Challenge.
- L'Hote, Y., Mahé,G., Some, B. and Triboulet, J.P. (2002). Analysis of a Sahelian Annual Rainfall Index from 1896 to 2000; the drought continues. *Journal* of *Hydrological science*, 47(4), 563– 572.
- Liu, H. L, Yang J Y, Drury C F, Reynolds W D, Tan C S, Bai Y L, He P, Jin J. and Hoogenboom G. (2011). Using the DSSATCERES- Maize Model to Simulate Crop Yield and Nitrogen Cycling in Fields Under Long-Term Continuous Maize Production. *Nutrient Cycling in Agroecosystems*, 89: 313-328.
- Le Treut H., Somerville R., Cubasch U., Ding Y., Mauritzen C., Mokssit A., Peterson T. and Prather M. (2007). *Historical Overview of Climate Change Science. In: Climate Change 2007: The Physical Science Basis.* Contribution of Working Group I to The Fourth Assessment Report of the Intergovernmental Panel On Climate Change (Solomon S., Qin D., Manning M., Chen Z., Marquis M., Averyt K.B., Tignor M. And Miller H.L., Editors).
- Loague, K. and Green, R.E. (1991). Statistical and graphical methods for evaluating solute transport models: Overview and application. *Journal of Contaminant Hydrology*, 7: 51-73.
- Lobell, D.B. and Burke, M.B. (2008). Why are agricultural impacts of climate change so uncertain? The importance of temperature relative to precipitation. *Environment Research Letters*, 3: 8.
- Lobel, D.B., Graeme, L.H., Greg, M., Messina, C., Michael, J.R. and Wolfram, S. (2013). *The Critical Role of Extreme Heat for Maize Production in The United States, New York*
- Lu, J., Vechhi, Gabriel A., and Reichler, Thomas. (2007). "Expansion of the Hadley Cell under Global Warming". *Geophysical Research Letters* 34 (6): L06805. Bibcode: 2007georl. 3406805L.
- Ly, M., Traore, S. B., Agali, A. and Benoit Sarr. (2013). Evolution of Some Observed Climate Extremes in The West African Sahel. *Journal Weather and Climate Extremes* 19–25.
- Mahalakshmi, V., Bidinger F.R. and Rao G.D.P. (1988). Timing and Intensity of Water Deficits During Flowering and Grain Filling in Pearl Millet. *Journal of Agronomic*, 80:130–135
- Mahalakshmi, V. and Bidinger F.R. (1985). Water Stress and Time of Floral Initiation in Pearl Millet. *Journal of Agricuture Science*, 105: 437–445.

- Maiti, R. K. and Bidinger, F. R. (1981). *Growth and Development of the Pearl Millet Plant*. ICRISAT Research Bulletin No 6.
- Malhi, Y. and Wright, J. (2004). Spatial patterns and recent trends in the climate of tropical rainforest regions. *Philosophy Transactions Royal. Social*, 359: 311-329.
- Makuvaro, V. (2014). Impact of climate change on smallholder farming in zimbabwe, using a modelling approach. PhD Degree in Agricultural Meteorology Department of Soil, Crop and Climate Sciences Faculty of Natural and Agricultural Sciences University of the Free State Bloemfontein South Africa. 317p.
- Matlon, P.J. (1990). Improving productivity in sorghum and pearl millet in semi-arid Africa. *Food Resource Institut Study*, 22: 1-43.
- Mangat, B.K., Maiti, R.K. and Kharval I.S. (1999) Pearl Millet Biology. In: Pearl Millet Breeding (I.S. Khairwal, K.N. Rai, D.J. Andrews and G. Harinarayana, Eds). New Delhi, Oxford & IBH Publishing Co.
- Marland, G., Boden, T. and Andres, R. (2000). *Global, regional, and national CO2 emissions trends.* A compendium of data on global change, Carbon Dioxide Analysis Center, Oak Ridge National Laboratory, USA. U.S. Department of Energy. Oak Ridge, Tenn.
- Mashingaidze, K. and Muchena, S.C. (1982). The Induction of Floret Sterility by Low Temperatures in Pearl Millet (Pennisetum Typhoides (Burm.) S. & H.). Zimbabwe J. *Agriculture Resource*, 20: 29–37.
- McCarthy J, Canziani O.F, Leary, N.A., Dokken, D.J. and White C. (2001). *Climate Change 2001: Impacts, Adaptation, and Vulnerability.* Contribution of Working Group II to The Third Assessment Report of the Intergovernmental Panel On Climate Change. Cambridge University Press, Cambridge.
- McCarthy, Canzian, O.F., Leary, N., Dokken, D. and White, K.S. (2008). Climate Change 2001: Impacts, Adaptation and Vulnerability, Contribution of Working Group II to, The Forth Assessment Report of the Inter-Governmental Panel of Climate Change, Cambridge University Press, UK And New York.
- McCarthy D.S., Vleka P.L.G., Bationo, A., Tabo, R. and Fosud M. (2010). Modeling Nutrient and Water Productivity of Sorghum in Smallholder Farming Systems in a Semi-Arid Region of Ghana. *Field Crops Research*, 118: 251–258.
- McCarthy, D.S. and Vlek, P.LG. (2012). Impact of Climate Change on Sorghum Production under Different Nutrient and Crop Residue Management in Semi-Arid Region of Ghana: a Modeling Perspective. *Journal of African Crop Science*, 20: 275-291
- McClean CJ, Lovett J.C, Küper W, Hannah L, Sommer JH, Barthlott W, Termansen M, Smith GF, Tokumine S. and Taplin J. (2005). African Plant Diversity and Climate Change. Annals of The Missouri Botanical Garden, 92(2): 139-152.
- McGuigan, C., Reynolds, R. and Wiedmer, D. (2002). *Poverty and climate change: Assessingimpacts in developing countries and the initiatives of the international community.* London School of Economics Consultancy Project for the Overseas Development Institute.
- McIntyre, B.D., Flower, D.J. and Riba, S.J. (1993). Temperature and Soil Water Status Effects On Radiation Use and Growth of Pearl Millet in A Semi-Arid Environment. *Agricultural and Forest Meteorology*, 66:211–227.

- McMichael T., Montgomery, H. and Costello A. (2012). Health Risk, Present and Future, From Global Climate Change. BMJ, 344: E1359.
- Meinke, K. (1996). Topological Methods for Algebraic Specification. *Theoretical Computer Science*, 166: 263- 290.
- Mendelsohn, R., Dinar, A. and Dalfelt A. (2000). *Climate Change Impacts On African Agriculture*. Preliminary Analysis Prepared for the World Bank, Washington, District of Columbia, 25 Pp.
- Mendelsohn, R., and Dinah, A. (2005). *Exploring Adaptation to Climate Change in Agriculture*. The Potential of Cross-Sectoral Analysis. ARD, World Bank. Issu 1.
- Meinzen, D.R., Quisumbing, B.J., Biermay, J.P., Wilde, V., Noordeloos, M., Ragasa, C. and Beintema, N. (2010). Engendering Agricultural Research, Global Forum on Agricultural Research, Rome, Italy.
- Michels, K., Sivakumar, M.V.K. and B.E. Allison. (1993). Wind Erosion in the Southern Sahelian Zone and Induced Constraints to Pearl Millet Production. *Agricultural and Forest Meteorology*, 67:65–77.
- Min, S.-K., Zhang, X., Zwiers, F.W. and Hegerl, G.C. (2011). Human Contribution to More- Intense Precipitation Extremes. *Nature*, 470:378–381.
- Ministere De L'environnement Et De L'assainissement. (2006). *Rapport National Sur l'Etat De l'Environnement. (2005)*. Ministère De l'Equipment Et Des Transports, 2007, Programme D'action National D'adaptation Aux Changements Climatiques.
- Mishra, A., Hansen, J.W., Dingkuhn, M., Baron, C., Traoré, S.B., Ndiaye, O. and Ward, M.N. (2008). Sorghum yield prediction from seasonal rainfall forecasts in Burkina Faso. *Agricultural and Forest Meteorology*, 148(11): 1798-1814.
- Mohamed, A.B. (2011). Climate Change Risks in Sahelian Africa. *Regional Environmental Change*, 11(1): 109-117.
- Mohino, E., Janicot, S., and Bader J. (2011). Sahel Rainfall and Decadal to Multi-Decadal Sea Surface Temperature Variability. *Climate Dynamics*, 37(3): 419-440.
- Mortimore, M.J. and Adams, W.M. (2001). Farmer Adaptation, Change and 'Crisis' In The Sahel. *Global Environmental Change*, 11(1): 49-57.
- Morton, J. F. (2007). The Impact of Climate Change on Smallholder and Subsistence Agriculture. *Proceedings of The National Academy of Sciences* 104:19680-19685.
- Moyo, M., Mvumi, B.M., Kunzekweguta, M., Mazyimavi, K., Craufurd, P. and Dorward P. (2012).
 Farmer Perceptions On Climate Change and Variability in Semi-Arid Zimbabwe in Relation to Climatology Evidence. *African Crop Science Journal*, 20: 317 335.
- Mubaya, C. P. Njukib, J., Mutsvangwa, E. P., Mugabe, F. T. and Nanja D. (2012). Climate Variability and Change or Multiple Stressors? Farmer Perceptions Regarding Threats to Livelihoods in Zimbabwe and Zambia. *Journal of Environmental Management*, 102:9-17.
- Mukhtar, A. and Fayyaz, U.H. (2011). *APSIM and DSSAT Models as Decision Support Tools*. 19th International Congress on Modelling and Simulation, Perth, Australia, 12–16 December 2011 Adepartment of Agronomy, PMAS-Arid Agriculture University, Rawalpindi, Pakistan.
- Muller, C., Bondeau, A., Popp, A., Waha, K. and Fader, M. (2010). *Climate change impacts on agricultural yields*. Background note to the World Development Report 2010. Potsdam Institute for Climate Impact Research, Potsdam.

- Myers, R. L. (2002). Alternative crop guide: pearl millet. Jefferson Institute. Washington, D.C. http://www.jeffersoninst.org/ (accessed 01 Aug. 2014).
- Naab, J., Bationo A., Benson, M. W., Traore, P.S, Zougmore, R., Ouattara M, Tabo, R. and Vlek P. (2012). African Perspectives on Climate Change and Agriculture: Impacts, Adaptation and Mitigation Potential.
- Nabikolo, D., Bashaasha, B., Mangheni, M.N. and Majaliwa, J. G.M. (2012). Determinants of Climate Change Adaptation Among Male and Female Headed Farm Households in Eastern Uganda. *Journal of African Crop Science*, 20 (2): 204-212.
- National Research Council: *Panel on Advancing the Science of Climate Change*. National Research Council. (2010). Advancing the Science of Climate Change.
- Nelson, V. and Stathers, T. (2009). Resilience, power, culture, and climate: a case study from semiarid Tanzania, and new research directions Gender and Development. *National Ressource Institut*, 17 (1), pp.81—94.
- New, M., Hewitson, B., Stephenson, D.A., Tsiga, A., Kruger, A., Manhique, A., Gomez, B., Coelho, C.A.S., Masisi, D.N., Kululanga, E., Mbambalala, E., Adesina, F., Saleh, H., Kanyanga, J., Adosi, J., Bulane, L., Fortunata, L., Mdoka, M.L. and Lajoie, R. (2006). Evidence of Trends in Daily Climate Extremes Over Southern and West Africa. *Journal Geophysical Resource*, 111: D14102.
- Nhemachena, C. and Hassan, R. (2007). Micro-Level Analysis of Farmers' Adaptation to Climate Change in Southern Africa, IFPRI Discussion Paper No. 714 (Washington, DC: International Food Policy Research Institute, 2007).
- Nicholson, S.E. (1980). The Nature of Rainfall Fluctuations in Subtropical West Africa. Monthly Weather Review, 108(4): 473-487.
- Nicholson, S.E., Kim, J. and Hoopingamer, J. (1988). Atlas of African Rainfall and its Interannual Variability. Department of Meteorology, Florida State University, Tallahassee.
- Nicholson, S.E., Some B. and Kone, B. (2000). An Analysis of Recent Rainfall Conditions in West Africa, Including The Rainy Seasons of The 1997 El Nino and The 1998 La Nina Years. *Journal of Climate*, 13 (14), 2628-2640.
- NSIDC. (National Snow and Ice Data Centre) (2010). Arctic Climatology and Meteorology Accessed on 21 June 2010 http://nsidc.org/arcticmet/basics/weather_vs_climate.html.
- Nwanze, K.F. and Sivakumar, M.V.K. (1990). Insect Pests of Pearl Millet in Sahelian West Africa: II. Raghuva Albipunctella De Joannis (Noctuidae, Lepidoptera): Distribution, Population Dynamics and Assessment of Crop Damage. *Journal of Pest Management*, 36: 59-65.
- Nwanze, K.F. (1989). Insect Pests of Pearl Millet in Sahelian West Africa: I. Acigona Ignefusalis (Pyralidae, Lepidoptra): Distribution, Population Dynamics and Assessment of Crop Damage. *Journal of Pest Management*, 35:137–142.
- Nyanga, P.H., Johnsen, F.H., Aune, J.B. and Kalinda, T.H. 2011. Smallholder Farmers' Perceptions of Climate Change and Conservation Agriculture: Evidence from Zambia. *Journal of Sustainable Development*, 4(4): 73-85.
- O'Brien, K. L. and Leichenko, R. M. (2000). Double Exposure: Assessing The Impacts of Climate Change Within the Context of Economic Globalization. *Global Environmental Change*, 10: 221-232.

- Ohshita, S. B. (2007). The Scientific and International Context for Climate Change Initiatives, 42 U.S.F.L. REV. 1- 37.
- Olof, D. (2008). Climate Change and Poor Management of Natural Resources as Binding Constraint to Growth in Mali. An Annex to *the Integrated Economic Analysis* 3(13-20).
 - Olof, D. and César E. (2013). Mali Environmental and Climate Change Policy Brief. The Centre for Environment and Sustainability, GMV. Sida's Helpdesk for Environment and Climate Change. *Göteborg, Sweden* 405 30.
- Omotosho, J.B., Balogun, A.A. and Ogunjobi, K. (2000). Predicting Monthly and Seasonal Rainfall, Onset and Cessation of the Rainy Season in West Africa Using Only Surface Data. *International Journal of Climatology*, 20: 865-880.
- Ong, C.K. (1983a) Response to Temperature in A Stand of Pearl Millet (Pennisetium Typhoides S. & H.): I. Vegetative Development. *Journal of Experimental Botany*, 34 (140) 322–336.
- Ong. (1983b). Response to Temperature in A Stand of Pearl Millet (Pennisetium Typhoides S. & H.). Reproductive Development. *Journal of Experimental Botany*, 34 (140) 337–348.
- Ong. (1983c). Response to temperature in a stand of pearl millet (Pennisetium typhoides S&H): 4. Extension of individual leaves. *Journal of Experimental Botany*, 34 (149) 1731-1739
- Onyewotu, L.O.Z., Stigter, C.J., Oladipo, E.I. and Owonubi J.J. (1998). Yields of Millet Between Shelterbelts in Semi-Arid Northern Nigeria, with a Traditional and a Scientific Metho d of Determining Sowing Date, And at Two Levels of Organic Manuring. Neth. *Journal of Agricultural Science*, 46:53–64.
- Onyewotu, L., Stigter, K., Abdullahi, Y. and Ariyo J. (2003). Shelterbelts and Farmers' Needs. LEISA. *Magazine on Low External Input and Sustainable Agriculture*, 19(4):28–29.
- Orindi, V. and Eriksen, S. (2005). Main Stream Adaption to Climate Change in Development Process in Uganda: Nairobi. *African Centre for Technology Studies (ACTS) Ecopolicy* 9966 (41): 130-5.
- Osbahr, H., Twyman C., Adger W. N. and Thomas D. S. G. (2008). Effective livelihood adaptation to climate change disturbance: scale dimensions of practice in Mozambique. *Geoforum*, 39:1951-1964.
- Osbahr, H., Dorward, P., Stern, R. and Cooper, S. (2011). Supporting Agricultural Innovation in Uganda to Climate Risk: Linking Climate Change and Variability with Farmer Perceptions. *Experimental Agriculture*, 47 (2): 293-316.
- Oteng-Darko, Kyei-Baffour N. and Ofori E. (2012). Simulating Rice Yields Under Climate Change Scenarios Using the Ceres-Rice Model. *African Crop Science Journal*, 2: 401 408.
- Oumar, I., Mariac, C., Pham J. L. and Vigouroux, Y. (2008). Phylogeny and Origin of Pearl Millet (Pennisetum Glsucum [L.] R. Br. As Revealed by Microsatellite Loci. *Theoretical and Applied Genetics*, 117:489-497.
- Oushy, H. (2010). Pearl Millet Fact Sheet. NMSU-AWATT, Afghanstan.
- Oxfam. (2007). Understanding gender differential impacts of tsunami and gender mainstreaming strategies in tsunami response in Tamil Nadu, India. no city
- Ozer, P., Hountondji Y.C., Laminou Manzo O. (2009). Evolution Des Caractéristiques Pluviométriques Dans L'est Du Niger De 1940 A 2007. *Geo-Eco-Trop*, 33: 11-30.
- Panwar, M. and Wilson J. (2001). Reaction of Pearl Millet Varieties during Rust Epidemics in Haryana, India. *International Sorghum and Millets Newsletter*, 42:79–81.

- Parry, M., Rosenzweig, C., Iglesias, A., Fisher, G. and Livermore. M. (1999). Climate Change and World Food Security: A New Assessment. *Global Environmental Change*, 9 51– S67.
- Parry, M.L, Canziani, O.F, Palutikof, J.P, Van Der Linden, P.J. and Hanson, C.E. (2007). *Climate Change 2007: Impacts, Adaptation and Vulnerability*. Contribution of Working Group II to The Third Assessment Report of the Intergovernmental Panel On Climate Change Cambridge University Press, Cambridge, United Kingdom.
- Paturel, J.E., Servat, E., Kouamé, B., Lubès, H., Ouedraogo, M. and Masson, J.M. (1997). Climatic variability in humid Africa along the Gulf of Guinea. Part II: an integrated regional approach. *Journal of Hydrology*, 191:16–36.
- Pearce, D, Cline, W., Achanta, A., Fankhauser S, Pachauri, R, Tol, R. and Vellinga P. (1996). *The Social Costs of Climate Change: Greenhouse Damage and Benefits of Control*. In: Bruce, J, Lee, H. and Haites, E. (1995). Climate Change: Economic and Social Dimensions of Climate Change. Cambridge University Press, Cambridge.
- Pearson, C.J. (1975): Thermal Adaptation of Pennisetum: Seedling Development. Aust. *Journal of Plant Physiology*, 21:431–424.
- Peterson, T.C., Taylor, M.A., Demeritte, R., Duncombe, D.L., Burton, S., Thompson, F., Porter, A., Mercedes, M., Villegas, E., Semexant-Fils, R., Klein-Tank, A., Martis, A., Warner, R., Joyette, A., Mills, W., Alexander, L. and Gleason, B. (2002). Recent Changes in Climate Extremes in The Caribbean Region. *Journal of Geophysical Research*, 107 (21): 4601.
- Pielke, R. A. Jr. (1998). Rethinking the Role of Adaptation in Climate Policy. *Global Environmental Change*, 8 : 159–170.
- Piéri, C. (1989). *Fertilité Des Terres de Savanes*. Bilan de Trente Ans de Recherche et de Développement Agricoles au Sud du Sahara. (Ministère de la Coopération et du Développement, CIRAD) : 443 Pp.
- PNUD. (2006). Gestion de l'Environnement pour un Développement Humain Durable.
- Quayum, M.A. and Ali, A.M. (2012). Adoption and Diffusion of Power Tiller in Bangladesh. Bangladesh Journal of Agriculture Ressource, 37: 307–325.
- Quay, W. (2008). Coping Mechanisms to Climate Variability in Ghana. *African Journal of Agricultural Research*, 3 (5): 334-341.
- Quisumbing, A., Haddad, L. and Peña, C. (1995). Gender and Poverty: New Evidence from 10 Developing Countries. FCND Discussion Paper No. 9, International Food Policy Research Institute, Washington, D.C.
- Rachie, K. O. and Majmudar J. V. (1980). *Pearl Millet*. The Pennsylvania State University Press, London, Pp. 307.
- Rai, K. N. G., R. Velu, V. N. Bhattacharjee, V. Kulkharni, T. Muralidharan, T. Longvan and S. Rveendran. (2007). *Gene Action for Grain Zinc Content in Pearl Millet*. International Conference on Zinc. Crops, Istanbul, Turkey.
- Rai, K. N., AppaRao S. and K. N. Reddy. (1997). Pearl millet. In: p. 243-258. D. Fuccillo et al. Biodiversity in trust. Cambridge University Press, Cambridge, UK. Ramstorf, S. (2007). A Semi-Empirical Approach to Projecting Future Sea-level Rise. *Science*, 315: 368-70.
- Reilly, J., Hohmann, N. and Kane, S. (1994). Climate Change and Agricultural Trade: Who Benefits, Who Loses? *Global Environmental Change*, 4 (1): 24-36.

- Reilly, J. and Schimmelpfennig, D. (1999). Agricultural impact assessment, vulnerability, and the scope for adaptation. *Climatic Change*, 43 : 745–788.
- République Du Mali. (2009). Evaluation Economique de la Gestion Environnementale au Mali Couts et Bénéfices, UNEP, UNDP, SBA.
- Rezaei E, E., Gaiser, T., Siebert, S., Sultan, B. and Ewert, F. (2014). Combined impacts of climate and nutrient fertilization on yields of pearl millet in Niger. *European Journal of Agronomy*, 55: 77-88.
- Ritchie, J.T. and Alagarswamy, G. (1989). Simulation of Sorghum and Pearl Millet Phenology. in Modeling the Growth and Development of Sorghum and Pearl Millet (Eds S. M. Virmani, H. L. S. Tandon & G. Alagarswamy), Pp. 24–26. Research Bulletin No. 12. Patancheru, Andhra Pradesh, India: ICRISAT.
- ROCAFREMI. (2002). Sélection et Mise à La Disposition des Paysans de Variétés et Semences Appropriées ; Résultats du Projet P1 1991-1996 .42 P.
- Roncoli, C., Ingram, K. and Kirshen, P. (2001). The Costs and Risks of Coping with Drought: Livelihood Impacts and Farmers¹ Responses in Burkina Faso. *Climate Research*, 19 (2): 119132.
- Rosenzweig, C. and Hillel, H. (1995). Potential Impacts of Climate Change on Agriculture and Food Supply. *Consequences*, 1 (2) 22-30.
- Rosenzweig, C. and Parry, M.L. (1994). Potential Impact of Climate-Change On World Food Supply. *Nature*, 367: 133-138.
- Rosenzweig, C. and Parry, M. L. (1994). Potential impact of climate change on world food supply. *Nature*, 367:133–138.
- Roudier, P., Sultan, B., Quirion, P. and Berg, A. (2011). The impact of future climate change on West African crop yields: What does the recent literature say? *Global Environmental Change*, 21(3), 1073–1083.
- Sanfo, S., Lamber, J. P.A., Mueller, M. and Fonta, W. (2014). Farmers' Perceptions of Climate Change and Climate Variability Versus Climatic Evidence in Burkina Faso, West Africa. Conference: Climate Change in Africa. Negotiations, Translations, And Socio-Political Implications. 10th 12th September 2014, Center for Development Research (ZEF), Bonn, Germany, At Bonn, Germany.
- Sarr, B. (2011). *Return of Heavy Downpours and Floods in A Context of Changing Climate. Climate Change in the Sahel.* A Challenge for Sustainable Development. AGRHYMET Monthly Bulletin (Special Issue), Pp.9–11.
- Sauerborn, J. (1991). The Economic Importance of the Phytoparasites Orobanche and Striga. In: Proceedings of the 5th International Symposium on Parasitic Weeds (J.K. Ranson, L.J. Musselman, D. Worsham and C. Parker, Eds). Nairobi, CIMMYT.
- Schlenker, W. and Lobell, D.B. (2010). Robust negative impacts of climate change on African agriculture. *Environmental Research Letters*, 5(1): 014010.
- Schneider, S. H. (1989). *Global Warming: Are We Entering the Greenhouse Century?* Sanfrancisco. Sierra Club Books.
- Schwartz, P. and Randall, D. (2003). An Abrupt Climate Change Scenario and Its Implications forUnitedStatesNationalSecurity.AccessedOn20July2009.

<u>Http://Www.Greenpeace.Org/Raw/Content/International/Press/Reports/An-Abrupt-ClimateChangescena</u>. Pdf.

- Scholes, R.J. and Biggs, R. (2004). *Ecosystem services in Southern Africa: A Regional Assessment*. CSIR, Pretoria, South Africa.
- Scoones, I. (1996). Hazards and Opportunities: *Farming livelihoods in Dryland Africa: Lessons from Zimbabwe*. London. Zed Books.
- Seo, S. N. and Mendelsohn, R. (2006). Climate Change Impacts On Animal Husbandry in Africa: A Ricardian Analysis. Centre for Environmental Economics and Policy in Africa (CEEPA) Discussion Paper No. 9, Pretoria. University of Pretoria, 42 Pp.
- Servat, E., Paturel, J.E., Kouame, B., Travaglio, M., Ouedraogo, M., Boyer, J.F., Lubes-Niel, H., Fristch, J.M., Masson, J.M. and Marieu, B. (1998). *Identification, caractérisation et conséquences d'une variabilité hydrologique en Afrique de l'Ouest et Centrale*. In: E. Servat, D. Hughes, J.M. Fritsch and M. Hulme (Editors), Water Resources Variability in Africa during the XXth Century, Abidjan, pp. 323-337.
- Shackleton, S. E. and C. M. Shackleton. (2012). Linking Poverty, HIV/AIDS and Climate Change to Human and Ecosystem Vulnerability in Southern Africa: Consequences for Livelihoods and Sustainable Ecosystem Management. International. *Journal of Sustainable Development & World Ecology*, 19:275-286.
- Singh, S.D. and King S.B. (1991). Pearl Millet Rust Present Status and Future Research Needs. Journal of Tropical Plant Diseases, 9:35–52.
- Sirohi, S. and Michaelowa A. (2007). Sufferer and Cause: Indian Livestock and Climate Change. *Climatic Change*, 85: 285–298.
- Sivakumar, M.V., Konaté, M. and Virmani, S.M. (1984). Agroclimatologie De l'Afrique De l'Ouest : Le Mali. Bulletin d'Information. Patancheru, ICRISAT.
- Sivakumar, M. and Hatfield, J. (1990). Spatial variability of rainfall at an experimental station in Niger, West Africa. *Theoretical and applied climatology*, 42(1): 33-39.
- Sivakumar, M. V. K. and Salaam, S. A. (1994). A Wet Excavation Method for Root/Shoot Studies of Pearl Millet On the Sandy Soil of the Sahel. ICRISAT Sahelian Center. *Exploitation Agricole*, 30: 329-336.
- Sivakumar, M.V.K., Das, H.P. and Brunini, O. (2005). Impacts of Present and Future Climate Variability and Change On Agriculture and Forestry in The Arid and Semi-Arid Tropics. *Climatic Change*, 70: 31-72.
- Solomon, S., Qin, D., Manning, M., Chen, Z., Marquis, M., Avery, K. B., Tignor, M. and Miller, H.L. (2007). *Climate Change: The Physical Science Basis*. Cambridge. Cambridge University
- Solomon, S., Plattner, G.K., Knutti, R. and Friedlingstein, P. (2009). Irreversible Climate Change Due to Carbon Dioxide Emissions. *Proceedings of The National Academy of Sciences*, 106: 1704– 1709.
- Solomon, S., Daniel, J. S., Sanford, T. J., Murphy, D. M., Plattner, G. K., Knutti, R. and Friedlingstein, P. (2010). Persistence of Climate Changes Due to Arrange of Greenhouse Gases. *Proceedings* of the National Academy of Sciences, 107: 18354–18359.
- Soman, P., Stomph T.J., Bidinger F.R. and Fussell L.K. (1987): Improvement in Stand Establishment in Pearl Millet. In: Food Grain Production in Semi-Arid Africa (J.M. Menyonga, J.M. Bezuneh and A. Ypodeowei, Eds). Ouagadougou, OAU/ STRC-SAFRAD.
- Staatz, J., Kelly, V., Boughton, D., Dembélé, N.N., Sohlberg, M., Berthé, A., Skidmore, M., Diarrah, C.O., Murekezi, A., Richardson, R., Simpson, B., Perakis, S., Diallo, A.S., Adjao, R., Sako, M., Me- Nsopé, N. and Coulibaly, J. (2011). *Malian agricultural sector assessment 2011*. USAID-Mali and Michigan State University.
- Stewart, J.I. (1991). Principles and performance of response farming. In: R.C. Muchow and J.A. Bellamy (Editors), Proceedings of the international symposium on Climatic risk in crop production: Models and management for the semiarid tropics and subtropics, Brisbane, Australia, pp. 361-382.
- Stigter, K. (2005). Wind Related Problems in Africa Researched by The TTMI-Project. Newsletter of the Europe/Africa Initiative of the International Association for Wind Engineering. Birmingham, IAWE.
- Stott, P.A., Jones, G.S., Christidis, N., Zwiers, F.W., Hegerl,G. and Shiogama, H. (2011). Single-Step Attribution of Increasing Frequencies of Very Warm Region Al Temperatures to Human Influence. *Atmospheric Science Letters* 12, 220–227.
- Sultan, B. and Janicot, S. (2003). The West African Monsoon Dynamics. Part II: The "Preonset" and "Onset" of the Summer Monsoon. *Journal of Climate* 16: 3407-3427.
- Sultan, B., Baron, C., Dingkuhn, M., Sarr, B. and Janicot, S. (2005). Agricultural impacts of largescale variability of the West African monsoon. *Agricultural and Forest Meteorology*, 128 (1–2): 93-110.
- Sultan, B., Roudier, P., Quirion, P., Alhassane, A., Muller, B., Dingkuhn, M., Ciais, P., Guimberteau, M., Traore, S. and Baron, C. (2013). Assessing climate change impacts on sorghum and millet yields in the Sudanian and Sahelian savannas of West Africa. *Environmental Research Letters*, 8(1): 014-040.
- Sultan B, Guan K., Kouressy M., Biasutti M., Pian C., Hammer G L, McLean. G and Lobell D B. (2014). Robust Features of Future Climate Change Impacts on Sorghum Yields in West Africa. *Environmental Research Letters* 9:104006.
- Sutton, D. (2007). Accessed on 20 June 2009. http://www1.agric.gov.ab.ca/\$department/deptdocs.nsf/all/cl9706
- Syngenta Foundation for Sustainable Agriculture. (2005). Pearl Millet Production Methods. <u>Http://Syngentafoundation.Org/Millet_Production_Methods_Pearl_Millet.Htm.</u>
- Syngenta Foundation for Sustainable Agriculture. (2006). Harnessing Modern Science in Africa to Sustain Sorghum and Pearl Millet Production for Resource Poor Farmers. Www.Syngentafoundation. Com/Millet. Html.
- Tenge, J.D.G. and Hella, J.P. (2004). Social and Economic Factors Affecting the Adoption of Soil and Water Conservation in West Usambara Highlands, Tanzania. Land Degradation and Development, 15:99-114.
- Teklewold, H., Dadi L., Yami A. and Dana N. (2006). Determinants of Adoption of Poultry Technology: A Double Hurdle Approach. Livestock Research for Rural Development 18 (3). Http://Www.Lrrd.Org/Lrrd18/3/Tekl18040.Htm.
- Thakur, R.P. and King S.B. (1988). Smut Diseases of Pearl Millet, Information Bulletin No. 25. Patancheru, ICRISAT.
- Thakur, R.P. (1990). *Basic Research On Management of Pearl Millet Diseases*. In: Basic Research for Crop Disease Management (P. Vidyasekaren, Ed.). New Delhi, Daya Publishing House.

- Theodorides, T.N. and Pearson C.J. (1981). Effect of Temperature on Total Nitrogen Distribution in Pennisetum Americanum. Aust. *Journal of Plant Physiology.*, 8:201–210.
- Thomas, D.G., Twyman, C., Osbahr, H. and Hewitson, B. (2007). Adaptation to Climate Change and Variability: Farmer Responses to Intra-Seasonal Precipitation Trends in South Africa. *Climatic Change*, 83 (3): 301-322.
- Thornton, P.K., Van De Steeg J., Notenbaert A. and Herrero M. (2009). The Impacts of Climate Change on Livestock and Livestock Systems in Developing Countries: A Review of What We Know and What We Need to Know. *Agricultural Systems*, 101: 113–127.
- Tingem, M., Rivington, M. and Colls, J. (2008). Climate variability and maize production in Cameroon: Simulating the effects of extreme dry and wet years. Singapore, *Journal Tropical Geography*. 29:357–370.
- Tingem, M. and Rivington, M. (2009). Adaptation for crop agriculture to climate change in Cameroon: Turning on the heat. *Mitigation and Adaptation Strategies for Global Change*, 14(2): 153168.
- Tizle, C.Y. (2007). *The Dynamics of Soil Degradation and Incentives for Optimal Management in The Central Highlands of Ethiopia*. PhD Thesis. Faculty of Natural and Agricultural Sciences, University of Pretoria; Pretoria, South Africa.
- Tol, R. S. J, Downing, T. E., Kuik, O. J. and Smith, J. B. (2004). Distributional Aspects of Climate Change Impacts. *Global Environmental Change*, 14 (3): 259-272.
- Traore, B., Corbeels, M., Van Wijk, M.T., Rufino, M.C. and Giller, K.E. (2013). Effects of Climate Variability and Climate Change on Crop Production in Southern Mali. *European Journal of Agronomy*, 49: 115- 125.
- Traore, B., Vanwijk, M.T., Descheemaeker, K., Corbeels, M., Rufino, M.C. and Giller, K.E., (2014). Farmer's Perceptions On Climate Change and On Agricultural Adaptation Strategies in Southern Mali. *Experimental Agriculture*, 51(4): 615-634.
- Traoré, P.C.S., Kouressy, M., Vaksmann, M., Tabo, R., Maikano, I., Traoré, S.B. and Cooper, P. (2007). *Climate Prediction and Agriculture: What Is Different About Sudano-Sahelian West Africa?* Chapter 19. In: M.V.K. Sivakumar and J. Hansen (Editors), Climate Prediction and Agriculture: Advances and Challenges. Springer, USA, Pp. 189-203.
- Traore, B. (2014). *Thesis on Climate change, climate variability and adaptation options in smallholder cropping systems of the Sudano Sahel region in West Africa*. C.T. de Wit Graduate School of Production Ecology and Resource Conservation. Pp173
- Traoré, S.B., Reyniers, F.N., Vaksmaann, M., Koné, B., Sidibé, A., Yoroté, A., Yattara, K. and M., K. (2000). Adaptation a la Sécheresse des Ecotypes Locaux de Sécheresse. *Sécheresse*, 11 : 22737.
- Traoré S. B., Alhassane, A., Muller, B., Kouressy M., Somé, L., Sultan, B., Oettli, P., Ambroise C. S., Sangaré, S., Vaksmann, M., Diop, M., Dingkhun, M. and Baron, C. (2010). Characterizing and Modelling the Diversity of Cropping Situations Under Climatic Constraints in West Africa. *Atmospheric Science Letters*, 12 (1): 89–95, Trenberth, K. E., Jones, P. D., Ambenje, P., Bojariu, R., Easterling, D., Tank, A. K., Parker, D., Rahimzadeh, F., Renwick, J. A. and Rusticucci, M., (2007a). In: Solomon, S., Qin, D., Manning, M., Marquis, M., Averyt, K. B., Tignor, M., Miller, H.L., Chen, Z. (Eds.), *Observations: Surface and Atmospheric Climate Change*. Chapter3 of Climate Change 2007a: The Physical Science Basis. Contribution of

Working Group I to The Fourth Assessment Report of the Intergovernmental Panel On Climate Change. Cambridge University Press, Cambridge, UK and New York, Pp. 235–336.

- Trenberth, K.E., Fasullo, J.T. and Kiehl, J. (2007b). *Earth's Global Energy Budget*. Bulletin of The American Meteorological Society.
- Tsuji, G. Y., Uehara, G. and Balas, S. (1998). Dssat Version 3, Vol. 4. Honolulu, Hi, Usa: University of Hawaii. Hoogenboom, G., Jones, J. W., Wilkens, P.W., Porter, C. H., Batchelor, W. D., Hunt, L. A., Boote, K. J., Singh, U., Uryaseva, O., Bowen, W. T., Gijsman, A. J., Du Toit, A. S., White, J.W. and Tsuji, G. Y. (2004). Decision Support Systems for Agrotechnology Transfer Version 4.0. University of Hawaii.
- UN. (United Nations). (2007). *The Millennium Development Goals* 2007. Online: Http://Www.Un.Org/Millenniu/Pdg/Mdg2007.Pdf.
- UNDP. (2004). *Meeting The Climate Challenge Sustaining Livelihoods: Lessons for The Future.* UNDP and Global Environmental Facility.
- UNEP. (1993). *The Issue of Winners and Losers, Climate Change Fact Sheets*. Accessed On 22 July 2009. Http: Www.Unep. Ch/Iuc/Submenu/Infokit/ Oldkit/FS-INDEX.HTM.
- UNEP. (United Nations Environment Programme). (2006). African regional implementation.
- UNFCCC. (2002). *Poverty and Climate Variability*. Reducing The Vulnerability of the Poor: A Contribution to The Eighth Conference of the Parties to The United Nations Framework. Convention On Climate Variability. Consultation Draft, UNFCCC.
- UNFCCC. (2007). *Climate Change: Impacts, Vulnerabilities and Adaptation in Developing Countries.* Bonn. Climate Change Secretariat UNFCCC.
- Vaksmann, M. and S. Traore. (1994). Adéquation Entre Risque Climatique Et Choix Variétal Du Mil. In : Bilan Hydrique Agricole Et Sécheresse En Afrique Tropicale (F.N. Reyniers And L. Netoyo, Eds). Paris, John Libbey Euro text.
- Vaksmann, M., Traoré, S. B. and Niangado, O. (1996). Le Photopériodisme Des Sorghos Africains. Agriculture Et Développement, 9: 13-18.
- Van Der Pol and Traore, S. (1993). Soil Nutrient Depletion by Agricultural Production in Southern Mali, Published in *Fertilizer Research* 36: 79-90.
- Vandenbeldt, R.J. and J.H. Williams. (1992). The Effect of Soil Surface Temperatures on the Growth of Millet in Relation to The Effect of Faidherbia Albida Trees. Agricultural and Forest *meteorology*, 60: 93-100.
- Vanlauwe, B., Bationo, A., Chianu, J., Giller, K.E., Merckx, R., Mokwunye, U., Ohiokpehai, O., Pypers, P., Tabo, R., Shepherd, K.D., Smaling, E.M.A., Woomer, P.L. and Sanginga, N. (2010). Integrated Soil Fertility Management. ISSN 0030-7270 – p.17-24
- Vedwan, N. (2006). Culture, Climate and The Environment: Local Knowledge and Perception of Climate Change Among Apple Growers in Northwestern India. *Journal of Ecological Anthropology*, 10(1): 4-18.
- Vermuelen, S., Dossou, C., Macqueen, D., Walubengo, D. and Nangoma, E. (2008). Springing back: climate resilience at Africa's grass roots. London. IIED.
- Vincent, L.A., Peterson, T.C., Barros, V.R., Marino, M.B., Rusticucci, M., Carrasco, G., Ramirez, E., Alves, L.M., Ambrizzi, T. and Berlato, M.A. (2005). Observed trends in Indices of Daily Temperature Extremes in South America 1960–2000. *Journal of Climate*, 18: 5011–5023.

- Virmani, S.M., Sivakumar, M.V.K. and Reddy, S.J. (1982). Rainfall Probability Estimates for Selected Locations of Semi-Arid India. Research Bulletin No. 1. Second Edition. Patancheru, ICRISAT.
- Vlek, P.L.G., Le, Q.B. and Tamene, L. (2010). Assessment of Land Degradation, Its Possible Causes and Threat to Food Security in Sub-Saharan Africa. In: Rattan Lal and B.A. Stewards, (Eds.), Advances in Soil Science - Food Security and Soil Quality. CRC Press, Boca Raton, Pp. 57 – 86.
- Waha, K., Müller, C., Bondeau, A., Dietrich, J.P., Kurukulasuriya, P., Heinke, J. and Lotze-Campen, H. (2013). Adaptation to Climate Change Through the Choice of Cropping System and Sowing Date in Sub-Saharan Africa. *Global Environmental Change*, 23 (1): 130-143.
- Waiswa, M. (2003). Strategic Choices for Enhancing Capacity of Rural Communities Adapt to Climate Variability: A Case for Uganda. Insights and Tools for Adaptation Learning from Climate Variability 18-19 November United States of America, Washington DC.
- Wallace, J.L., Lloyd C.R. and Sivakumar, M.V.K. (1993). Measurement of Soil, Plant and Total Evaporation from Millet in Niger. *Agricultural Forest Meteorology*, 63:149–169.
- Warrick, R. A. and Bowden, M. J. (1981). *The Changing Impacts of Drought in The Great Plains*. In Lawson, M. P., And Baker, J. C., (Eds.) The Great Plains: Perspectives and Prospects, Pp. 111-137, Centre for Great Plains Studies. Lincoln, Nebraska. University of Nebraska.
- Washington, R. and Harrison, M. (2004). African Climate Report. A Report Commissioned by The UK Government to Review African Climate Science, Policy and Options for Action, Climatology Research Group, Oxford University Centre for The Environment, UK.
- Watson, R. T. (2001). Climate Change: Synthesis Report: Contribution of Working Groups I, II, And III to The Third Assessment Report of the Intergovernmental Panel On Climate Change. Ambridge. Published for The IPCC by Cambridge University Press.
- Weart, S. (2003). The Discovery of Global Warming. Cambridge, Harvard University Press. 240 pp.
- WISARD. (1999). (Web-Based Information Services for Agricultural Research for Development), Marker-Assisted Improvement of Pearl Millet Downy Mildew Resistance in Elite Hybrid Parental Lines for Africa and Asia.

Http://Www.Wisard.Org/Wisard/Shared/Axp/Projectsummary.Asp?Kennummer=8186.

- Wheaton, E. E. (1994). Impacts of a Variable and Changing Climate on the Canadian Prairie Provinces: A Preliminary Integration and Annotated Bibliography. *American Journal of Climate Change*, 3: 366-377
- Winkel, T. and Do F. (1992). Caractères Morphologiques Et Physiologiques De Résistance Du Mil (Pennisetumglaucum (L.) R. Br.) *Sécheresse. Agriculture Tropical*, 46 (4): 339-351.
- Winkel, T., Payne, W. and Renno J.-F. (2001) Ontogeny Modifies the Effects of Water Stress On Stomatal Control, Leaf Area Duration and Biomass Partitioning of Pennisetum Glaucum. New Phytologist, 149 (1):71–82.
- WMO. (World Meteorological Organization). (1996). Agrometeorology of the Pearl Millet (V. Mahalakshmi, Rapporteur). Cagm Report No. 66 (/TD-No. 746), Geneva.
- World Bank. (2013a) Turn Down the Heat- Climate Extremes, Regional Impacts and The Case for Resilience.
- World Bank. (2013b). Mali Economic Update, January 2013.

- World Bank. (2006). Mali- Des Diagnostics Sectoriels Vers Une Stratégie Intégrée De Croissance : Mémorandum Economique.
- World Bank. (2007). Project Paper on a Proposed Additional Financing for an Agricultural Services and Producer Organizations Project Report No: 39630-ML.
- Wozniak, G.D. (1984). The Adoption of Interrelated Innovations: A Human Capital Approach. *The Review of Economics and Statistics*, 66 (1): 70-79.
- WRI. (1998). 1998–99 World Resources Database Diskette: A Guide to the Global Environment. Washington, DC, USA. World Resources Institute, diskettes.
- Yadav, O. P. (2011). Project Coordinator's Review On Pearl Millet Research. 2010-2011.
- Yahaya, I. (2009). Evaluation De L'effet De L'application Périodique De L'urine Hygienisée Sur La Culture Du Mil (Pennisetum Glaucum). Mémoire De Fin D'étude Pour L'obtention Du Diplôme d'Ingénieurs Des Techniques Agricoles (ITA), Option : Productions Végétales ;
 Faculté d'Agronomie De L'université Abdou Moumouni De Niamey ;39p.
- Zhang, X. and Yang, F. (2004). *RClimDex (1.0) User Guide*. Climate Research Branch Environment Canada. Downs view Ontario, Canada.
- Zegada-Lizarazu, W. and Iijima, M. (2004) Differences in Drought Avoidance Root Characteristics among Several Millet Species. In: New Directions for A Diverse Planet: Proceedings for The 4th International Crop Science Congress, Brisbane, Australia, 26 September–1 October 2004 (T. Fischer, Ed.), Gosford, Australia, The Regional Institute.

Zeng, N. (2003). Drought in The Sahel. Science, 302: 999-1000.

- Zoellick, S. and Robert, B.A. (2009). *Climate Smart Future*. The Nation Newspapers. Vintage Press Limited, Lagos, Nigeria, Pp 18.
- Zongo, J. D., Sedego, M. C., Seremé, P. and Zangré, G. R. (1988). Synthèse Des Prospections des Mils (Pennisetum Typhoides, Stapf Et Hubbard) au Burkina Faso. Proceedings of Regional Pearl Millet Improvement Workshop. 15-16 Aug.

